

DETAILED OPTIONS ASSESSMENT REPORT

RAIL DELIVERY GROUP

NOVEMBER 2015

FINAL

Prepared by:

Cambridge Economic Policy Associates Ltd



CONTENTS

1.	Intro	duction and background	.4
1.1	L. S	Scope of this chapter	.4
1.2	<u>?</u> . F	Purpose	4
1.3	3. I	Introduction	5
1.4	ł. (Overview of assessment methodology	6
1.5	5. 7	The detailed assessment options	7
1.6	5. I	Input from RDG representatives	9
1.7	7. F	Report structure	9
2.	Asses	sment methodology	10
2.1	L. S	Scope of this chapter	10
2.2	2. [Detailed assessment methodology	10
2.3	3. /	Assessment criteria	12
2.4	4. 9	States of the World	13
2.5	5. 7	The five-level traffic light grading system	14
2.6	5. (Overall grades	15
3.	Selec	tion of options for detailed assessment	16
3.1	L. S	Scope of this chapter	16
3.2	2. (Option selection	16
3.3	3. 5	Selection of detailed assessment options	18
4. 3	Sumn	nary of detailed assessments	21
4.1	L. S	Scope of this chapter	21
4.2	<u>2</u> . ľ	Network charge options	21
4.3	3. F	Performance incentive options	30
4.4	1. F	Possessions incentive options	36
5.	Analy	/sis	41
5.1	L. S	Scope of this chapter	41
5.2	2. [Detailed assessment options as part of a package	41
5.3	3. \	Variations by State of the World	44
5.4	4. 5	Summary	46
ANNI	EX A	Simulation modelling	49
ANNI	EX B	Fixed charge based on avoidable cost detailed assessment	71
ANNI	EX C	Administered scarcity charge detailed assessment	91

ANNEX D	Path reservation charge detailed assessment 107
ANNEX E detailed ass	Recalculate benchmarks for traffic changes and remove capacity charge sessment
ANNEX F detailed ass	Recover end-user compensation through the performance regime (Schedule 8) sessment
ANNEX G	More frequent ACS calculation detailed assessment152
ANNEX H assessment	Reform Schedule 4 discount structure for notice period of possession detailed 163
ANNEX I	Assessment criteria 179
ANNEX J	Key features of alternative SoWs 182

Disclaimer:

This report has been commissioned by RDG. CEPA accepts no liability for use of this note or any information contained therein by any third party. © All rights reserved by Cambridge Economic Policy Associates Ltd.

1. INTRODUCTION AND BACKGROUND

The Rail Delivery Group (RDG), supported by Cambridge Economic Policy Associates (CEPA), has undertaken Phase 3 of its review of the charges and incentives regime for use of Network Rail's infrastructure. This Detailed Assessment Report is the third report in Phase 3. Two earlier standalone reports have been published in this phase:

- **'Factors' Report**: An initial contextual document, setting out the factors impacting the form and/or the effectiveness of charges and incentives. This has informed the analysis in the subsequent stages of the review.
- Initial Assessment Report: An initial high-level assessment of a list of 22 options for the charges and incentives regime, which looked at each option in the context of several key criteria in line with RDG's Vision. It identified the seven options to be taken forward into the more detailed assessment.

This report describes the detailed assessment of the seven options identified in the previous stage of this phase. The rest of this report outlines the approach and results of this detailed assessment.

1.1. Scope of this chapter

This introductory section sets out the:

- context for the report;
- the seven options that this report considers;
- approach to this stage of analysis; and
- the structure of the remainder of the report.

1.2. Purpose

The purpose of this report is to discuss our detailed assessment of seven options for changes to the charges and incentives regime for use of Network Rail's infrastructure. We set out our approach to the work, describe our modelling, and summarise the detailed assessments which are appended in Annexes B to H.

The seven options considered in this report should not be interpreted as those where there is any industry consensus as to the desirability of those options, rather that they are options that stakeholders thought merited more detailed analysis to inform the debate for ORR's next periodic review.

This report can be read on a standalone basis, but accompanies the two earlier reports produced in this Phase of RDG's Review of Charges.

1.3. Introduction

RDG's Contractual and Regulatory Reform workstream is carrying out a review of the charges and incentives regime. This project began in Spring 2014 and is expected to be completed by the end of 2015.

Once completed, RDG's review should allow the industry constructively to inform the Office of Rail and Road's (ORR's) next periodic review process (the 2018 Periodic Review (PR18)), and future reviews, by presenting the industry's own views on the regime.

By setting out the industry's views before the start of PR18, RDG can provide ORR with information that can help inform ORR's decisions, and potentially allow it to prioritise work in certain areas.

RDG has commissioned Cambridge Economic Policy Associates (CEPA) to assist in Phase 3 of its review. This follows RDG's previous work in Phases 1 and 2 of the review,¹ which produced:

- RDG's vision for the charges and incentives regime in the long run (the RDG Vision);²
- an assessment of the current charges and incentives regime;³ and
- a description of current and potential alternative States of the World (SoWs).⁴

We, CEPA, are working with RDG in Phase 3 to develop and assess options for a new and/or updated charges and incentives regime.

The detailed assessments developed in Phase 3 of our work reflect CEPA's independent assessment of a number of potential options for change to the current charges and incentives regime. These assessments have had the benefit of significant input from RDG representatives and the wider rail industry e.g. in order to scope options which mitigate industry concerns or reservations about a particular form of charge. This input has allowed us to ensure that the development of the options and our assessments of them are grounded in the reality of the range of business models currently in operation within the rail industry.

In addition, RDG members wanted to be able to demonstrate to ORR the range of views that exist within the industry, including where there is no collective view. We have sought to capture the range of views within this report. Stakeholder views on the implications or uncertainties of each option are specifically documented as part of each option assessed. This is intended to provide ORR access to the range of industry views of each option, and

¹ The publications released to date in RDG's Review of Charges are accessible via:

http://www.raildeliverygroup.com/what-we-do/our-work-programme/contractual-regulatory-reform/review-of-charges.html.

² RDG (Dec 2014) *"RDG vision for the charges and incentives regime in the long run"* available <u>here</u>

³ RDG (May 2015) "Assessment of the current charges and incentives regime" available here

⁴ RDG (May 2015) "Current and potential alternative states of the world" available <u>here</u>

should help inform ORR as it takes the work forward from the final phase of the RDG Review of Charges into the PR18 programme.

The objective of Phase 3 of RDG's Review of Charges is to develop options for changes to the charges and incentives regime. As shown in Figure 1.1 below, work has already been completed to perform an initial assessment of a list of 22 options, to filter them to seven options for detailed assessment, and to review the factors that affect the form and/ or effectiveness of the charging and incentives regime.⁵



Figure 1.1 – How this report fits into Phase 3 of the RDG Review of Charges

Source: CEPA amendment of RDG diagram

1.4. Overview of assessment methodology

The RDG Vision, established in Phase 1, provided the assessment criteria used to assess each option. The descriptions of current and potential alternative SoWs, established in Phase 2a, provided a set of scenarios under which the list of 22 options could be assessed. The work to assess the current charges and incentives regime, completed in Phase 2b, informed the RDG work at the start of Phase 3 to establish an initial list of 22 options. The RDG's charges and incentives user guide,⁶ developed as part of RDG's Review of Charges, helped to identify the counterfactual elements of the regime against which options could be assessed. The work on factors impacting the form and/or the effectiveness of charges and incentives, also completed in Phase 3, informed the analysis of what could be achieved by each option in

⁵ CEPA (Nov 2015) "Review of factors impacting the Form and/or the Effectiveness of Charges and Incentives"

⁶ RDG (Jul 2014) "Charges and incentives user guide" available <u>here</u>

different SoWs. In the second report for Phase 3, seven options were selected as meriting more detailed assessment, which is the subject of this report.

While building on the work completed earlier in RDG's review, this report also serves as a standalone resource with its set of detailed option assessments provided in Annexes B to H. The individual assessments capture industry views on each option under each SoW. They provide a useful body of evidence should the ORR wish to pursue one of these options or if the sector were to move towards an alternative SoW.

1.5. The detailed assessment options

The options shown in green in Figure 1.2 below are those for which further detailed assessment was undertaken. The selection of these options does not represent an industry consensus that any options represent an improvement on the present charging system; they are not recommended options. Rather, these are options that RDG representatives considered would merit further assessment. For example, an option may have been selected because it sought to address a weakness or gap in the current arrangements and / or because ORR might pursue such an option and the industry wishes to provide views as part of the debate and to inform ORR's consultation in more detail from the initial assessment.

Figure 1.2 – The options selected for detailed assessment (in blue) in the context of all options assessed



When developing the options for assessment, we have, as far as possible, attempted to envisage implementation of those options that maximise their practical feasibility in operation, although clearly more detailed work may be required. Further explanation of how we chose the seven options to assess in more detail is set out in the initial assessment report.⁷

In this context, it is important to note that the selection of options for further assessment does not rule out other options. Initial assessments remain valid and some could be pursued; only a minority were entirely disregarded for reasons such as legality or practicality. We also note that given the large number of potential combinations of reforms that could be made at the same time,⁸ the analysis presented in this report has focused primarily on the merits of each option in isolation.

The options considered in further detail include three network charging options, two options for reforming the Schedule 4 possessions regime, and two options for reforming the Schedule 8 performance regime, in one case removing the capacity charge. No stations

⁷ CEPA (Nov 2015) "Initial options assessment report"

⁸ There are 128 combinations of the seven options considered in this report.

charges options were taken forward for more detailed assessment. The group considered that the most significant issues in relation to stations were structural and contractual. Therefore, whilst charges are an important consideration for stations, the group thought that the other issues should be addressed before seeking to reform stations charges.

1.6. Input from RDG representatives

Our findings have been discussed with RDG and other industry representatives through a series of workshops and one-to-one meetings to gather the information we required to develop our findings.

The project was guided by a working group (Review of Charges Executive Group) which included representatives from passenger operators, freight operators, Network Rail and governments (Department for Transport (DfT), Transport Scotland and Welsh Government), with ORR attending as an observer.

1.7. Report structure

The remainder of this report is set out as follows:

- Section 2 sets out the assessment methodology applied and the criteria used. It also summarises the simulation modelling work for the detailed assessments, which is covered in more detail in Annex A. Information on the evaluation criteria and the SoWs in which options were considered is provided in Annexes I and J.
- Section 3 summarises the seven options considered in the detailed assessment.
- Section 4 summarises the detailed assessments, which are provided in full in Annexes B to H.
- Section 5 provides a summary of the findings of the detailed assessments, including how the options might fit together and the degree of scope for change in the current SoW.

2. Assessment methodology

This section sets out the methodology used for the detailed analysis of charges and incentives regime options. It builds on earlier stages of the RDG Review of Charges.

2.1. Scope of this chapter

This section describes:

- the detailed assessment methodology;
- the SoWs used;
- the "traffic light" grading system used; and
- how overall grades were awarded to each option.

2.2. Detailed assessment methodology

Figure 2.1 provides an overview of the high-level assessment process.

Figure 2.1 – Detailed assessment process



The detailed impact assessments used standardised templates, designed with input from RDG representatives. The template required an assessment against 19 agreed criteria and also included additional sections to capture relevant wider information and to inform the process. These additional sections comprised:

- a detailed description of the option, including other issues aiding understanding of the option;
- the counterfactual charging arrangements, which identify the benchmark the proposal was assessed against;
- factors impacting the form and/or the effectiveness of the proposal as identified earlier in Phase 3;
- implementation issues, including information requirements, drivers of differentials in the charge, how the charge would be calculated, practical issues, and an assessment of resources required for implementation; and
- implications of the proposal for different stakeholders, informed by the simulation modelling work carried out.

We carried out simulation modelling to consider the potential impact of implementing the options considered in the detailed assessment, to the extent that it was practical to do so. In most cases we have undertaken simulation modelling on the basis of realistic assumptions, confirmed with industry and/or using existing data and analysis. The modelling is 'indicative' and is designed to give an 'order of magnitude' impact of the option rather than precise estimate of the impact. This simulation modelling is discussed further in Annex A.

The simulation modelling informed our understanding of the impact on stakeholders and also assisted us to refine, and in some cases adjust, our assessment of the options against the criteria. However, any option considered for implementation would require more substantial and detailed modelling to be undertaken.

We assessed each option against the 19 agreed criteria, which are set out in Section 2.3. The main assessment assumes that each option would operate in the current SoW and is therefore affected by the constraints that are discussed in the 'Factors' report e.g. the modest impact of charges on franchised passenger operators. However, we also made subsidiary assessments, on the same basis, of the options in the seven alternative SoWs that are set out in Section 2.4.

Each option was assessed based on its performance against the relevant counterfactual, i.e. an assessment of whether or not the option being assessed would be an improvement on the relevant part of the existing regime.

We graded each option using a directional "traffic light" system, described in Section 2.5. The traffic light system is slightly more granular compared to the initial assessment – we have added two additional grades of '- -' and '++'. The assessment criteria do not provide for a meaningful quantitative metric and therefore, we have retained the qualitative summary assessment for the detailed assessment.

As with the initial assessments, each detailed assessment includes an overall traffic light grading for each option in each SoW. This grading reflects an 'in the round' judgement; it is not a simple sum of the subsidiary grades. The overall grade is supported by a qualitative summary of the reasons for it.

Throughout the process of developing the detailed assessments we sought feedback from industry representatives, primarily from RDG's Review of Charges Executive Group, through workshops, sharing of drafts, one-to-one meetings and discussions with individuals. This engagement process allowed us to capture some nuances/detail that are relevant for each option and to set out industry views on each option under each SoW. In a separate section of the assessment template, we present the viewpoints of different industry representatives regarding the different options.

2.3. Assessment criteria

As shown in Figure 2.2, the assessment criteria we used for the initial assessment of options were drawn directly from the RDG Vision. Using the RDG Vision in this manner ensured that the assessments captured the most important considerations for the rail industry. These are the same criteria used in the initial assessment.

Axioms	Objectives							
 System safety Consistency with law Funding of Network Rail efficient costs Allowance for market conditions A single approach to the network as a whole 	 Service costs recovery Efficient whole-system whole-life industry net costs Efficient long run investment decisions Efficient performance management Efficient use of network capacity 							
udgement criteria	Outputs							
Predictability	Network Rail accountability							
Simplicity	Non-arbitrary allocation of costs							
Transparency	Optimal traffic growth							
Low transaction costs	Aligning industry incentives							

• Value for money for funders, taxpayers and users

Full descriptions of the criteria drawn from the RDG Vision are provided in Annex I. A small number of minor clarifications, capturing industry feedback on how the RDG Vision should be used for the initial assessments, were made to the descriptions of four of the nineteen criteria:

- consistency with law clarified to note particular regulations and laws that would be captured, as well confirming that this would be the criterion to capture specific impact tests considered by ORR such as those regarding the environment;
- allowance for market conditions clarified to note that legal considerations related to market conditions would be captured here rather than under consistency with law;
- simplicity clarified to capture an extended description developed in Phase 2b⁹ of RDG's work and to note that it would capture practicality i.e. is it possible to calculate and apply the charges at the required level of granularity; and

⁹ L.E.K. (May 2015) "Assessment of the current charges and incentives regime" available here p7

 value for money for funders, taxpayers and users – a description was provided by RDG for this criterion as it was not defined in the RDG Vision. This reflects the use of the "Output" category of the RDG Vision as criteria rather than the outcomes that would occur when the other criteria were fulfilled.

The criteria used here differ from the "proposed criteria" that the ORR presented at the "discussion on the structure of charges" workshop on 21st July 2015 but the differences are primarily in presentation and emphasis rather than substantive content.

2.4. States of the World

A number of the external factors that interact with the charges and incentive regime may vary over the medium to long term. These have the potential to influence the effectiveness of any given charging or incentive option. A SoW, therefore, describes the broader industry arrangements in which the charges and incentives regime might operate.

Seven alternative SoWs were developed during Phase 2a of RDG's Review of Charges¹⁰ and have been adopted here to allow the impact assessments to capture a broader range of potential future industry landscapes. These are in addition to the current SoW, which describes the industry as it is today, i.e. considering Network Rail (as the infrastructure provider), passenger and freight services, funders, governments and regulation. Alternative SoWs reflect potential changes along one, or several dimensions that affect these market participants. These are listed in Table 2.1. Further detail is provided in Annex J.

No.	State of the World	(Short name)
1.	A more dynamic railway	(Dynamic railway)
2.	On-rail competition via flexible franchising	(On-rail comp)
3.	More highly specified franchises	(Specified franchises)
4.	Freight protection / subsidy	(Protect freight)
5.	Beneficiary pays for capability	(Beneficiary pays)
6.	Change in approach to capacity allocation	(Capacity allocation)
7.	More regional decision making	(Regional powers)

Table 2.1:	Alternative	States	of the	World

SoWs reflect high-level changes to the structure of the industry. Changes are not described in detail because the purpose of the SoWs is to help test potential changes to charges and incentives. Greater detail of alternative SoWs would require additional assumptions and these are unlikely to accurately reflect any actual changes to the current SoW. Therefore, the impact of charging options considered in the detailed assessments under alternative

¹⁰ RDG (May 2015) *"Current and potential alternative states of the world"* available <u>here</u>

SoWs are also made at the level of principle. A review of detailed implementation issues in and additional quantitative assessments, for alternative SoWs were not considered to be practicable and have not been attempted. Impacts in the current SoW were often common across alternative SoWs, so comments were only made by exception for alternative SoWs.

2.5. The five-level traffic light grading system

We do not think that it is helpful to provide an assessment of each option based on a mechanistic assessment, e.g. by taking a weighted average of the scoring against each of the 19 criteria. Whilst our analysis is relatively detailed, the criteria do not provide for meaningfully precise metrics or weightings. However, in view of the more detailed information against which we have assessed the options, we have extended the "traffic light" system we used in the initial assessment from three levels to five levels, to indicate how well an option was expected to perform against each criterion. The following grades were given with reference to the current charges and incentives regime:

- **Dark Red (--):** indicates that the option is expected to have a clearly or strongly negative impact on a given criterion (compared to the current regime);
- **Red (-):** indicates that the option is expected to have a somewhat or probably negative impact on a given criterion (compared to the current regime);
- Amber (=): indicates that the expected impact on the criterion was equivalent to the current regime, or the impact is unclear; and
- **Green (+):** indicates that the option is expected to have a somewhat or probably positive impact on a given criterion (compared to the current regime); and
- **Dark Green (++):** indicates that the option is expected to have a clearly or strongly positive impact on a given criterion (compared to the current regime).

These grades were considered under each SoW for each criterion. It is important to reiterate that each option has been considered in isolation. Therefore, grades reflect the impact of the introduction of that particular option only and not any other charges that could be introduced alongside it to mitigate any anticipated negative impacts or enhance the positive. We discuss this further in Chapter 5 where we consider the options as part of an overall charging package.

An example of scoring for one assessment criterion is shown in Figure 2.3 below.

_								
	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
Funding of NR efficient costs	=		+	-	-	=	++	=

Figure 2.3 – Example of traffic light system

In the assessment templates, each traffic light score is also accompanied by a brief description justifying the grade.

It is important to stress that the traffic lights indicate directional impacts only and are not directly related to potential magnitudes of impacts. The individual grades were not designed to capture relative magnitudes of impacts. Any such observations were made in the accompanying commentary or reserved for the overall grading.

2.6. Overall grades

Each charging option received an overall grade based on a balanced consideration of the individual grades for each assessment criterion. The overall assessment grade is not a simple sum of the individual grades, because the individual grades are qualitative and there is no clear metric which can be used for weightings. A particular option may receive a mixture of Red, Amber and Green grades based on individual criteria while receiving, say, a Green grade overall. In any case, the aim of this review is not necessarily to provide a clear recommendation, but rather to contribute to ORR's periodic reviews and indicate the views and impacts on various stakeholders of the different options. The overall grade is illuminated by a commentary.

The evaluation has been informed by workshops with RDG's Review of Charges Executive Group, comments on drafts received from the Executive group and other stakeholders, and meetings with individual stakeholders.

However, whilst the traffic-light grading reflects our, CEPA's, assessment of an option, the commentary highlights the main areas where CEPA's assessment differs from the views of RDG representatives. Therefore, it is important that readers consider both the traffic light grading and the accompanying commentary to understand the findings of RDG's work.

3. SELECTION OF OPTIONS FOR DETAILED ASSESSMENT

This chapter introduces the seven options which have been selected for more detailed assessment. It includes three options for network charges, two options for the Schedule 4 possessions regime and two options for the Schedule 8 performance regime.

3.1. Scope of this chapter

This section:

- discusses the selection process to identify the options for detailed assessment; and
- describes, at a high level, the options selected for detailed assessment.

3.2. Option selection

An initial assessment of 22 options was completed earlier in Phase 3 of RDG's Review of Charges.¹¹ An overview of the initial assessment is provided in Table 3.1 below but with the revised grades provided for the detailed assessment options highlighted within the table.

As the overview indicates, only capacity auctions rated poorly throughout the initial assessment. Although some other options were given red grades overall e.g. for reasons of practicality or legality, the majority of options assessed were found to have some positive attributes in the current or alternative SoWs.

¹¹ CEPA (Nov 2015) "Initial options assessment report"

Table 3.1: Overview of CEPA initial and detailed option assessments in the current SoW

Option group				_	Netw	vork ch	arges	_				Stat	ion cha	rges	Performance regime				Possessions regime			
Option	Avoidable cost	Ability to pay mark-ups	Scarcity charge (LRMC)	Scarcity charge (administered)	Scarcity charge (auctions)	Environmental charge	Reservation charge	Track occupancy charge	Geog. VUC	Avg. cost charge	Revenue sharing	Regulate station QX	Station-by-station LTC	Station revenue sharing	Reset benchmarks more frequently	More granular rebranded capacity charge	Payments < 100% compensation	Recover end-user compensation	More frequent ACS calc.	Benchmarked possessions regime	Possessions compensation < 100%	Reform discounts
Option No.	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22
Axiom																						
System safety	=	=	=	=	-	=	=	=	=	=	=	=	=	=	=	=	=*	=	=	=	=*	=
Consistency with law	=	=	=	=	-	=	=	=	=	-	=	=	=	=	=	=	=	=	=	=	=	=
Funding of Network Rail efficient costs	=	=	=	=	-	=	=	=	=	-	=	=	=	=	=	+	=	=	++	=	=	=
Allowance for market conditions	=	+	=	-	-	-	=	=	-	-	+	=	=	+	=	=	-*	=	=	=	-*	+
A single approach for the network as a whole	+	=	+	=	-	=	=	+	=	+	-	+	+	-	+	=	=	=	=	=	=	=
Objective																						
Service costs recovery	=	=	=	=	-	=	=	=	=	=	=	=	+	=	-	+	=	=	++	=	=	=
Efficient whole-system whole-life industry net costs	=	=	=	=	-	+	+	=	+	=	+	=	+	+	=	+	+	=	=	=	=	-
Efficient long run investment decisions	=	=	=	=	-	+	=	=	+	=	=	=	=	=	=	=	-*	+	+	=	=	=
Efficient performance management	=	=	=	=	-	=	=	=	=	-	=	=	=	=	=	=	+	+	=	=	+*	+
Efficient use of network capacity	=	+	=	+	-	-	+	=	+	-	=	=	=	=	=	+	=	=	=	=	-	+
Judgement criteria																						
Predictability	=	=	-	-	-	=	-	=	=	+	=	=	-	=	+	=	=	=	-	=	=	=
Simplicity	=	=	-	-	-	-	=	=	-	+	=	+	+	-	+	=	=	-	-	+	=	+
Transparency	+	=	+	+	-	=	+	=	=	+	=	+	+	=	+	+	=	=	=	+	=	=
Low transaction costs	-	=	=	-	-	-	-	=	-	+	=	-	-	=	-	-	=	-	-	=	=	=
Outcome																						
Network Rail accountability	=	=	=	=	-	=	=	=	+	=	+	=	+	+	+	=	_*	++	=	=	-*	=
Non-arbitrary allocation of costs	++	=	+	=	-	+	=	=	+	-	=	=	+	=	-	=	=*	=	++	=	-	-
Optimal traffic growth	=	+	=	+	-	-	=	=	+	-	=	=	-	=	+	+	=	=	=	=	=	=
Aligning industry incentives	+	=	+	+	-	=	+	-	=	-	+	=	+	+	=	=	+	+	=		+*	+
Value for money for funders, taxpayers and users	=	=	=	=	-	-	+	=	=	-	=	=	+	=	=	+	-	+	+	=	_*	-
Overall																						
Option assessed in isolation	=	=	=	+	-	-	+	-	=	-	+	=	-	-	+	+	=	+	+	=	=	=

Key: * Grading would change if direction of options 17 and 21 flip; Options selected for assessment in Stage 3 are indicated by a black border 🗌

3.3. Selection of detailed assessment options

Seven of the 22 options were selected for further analysis. These seven options were chosen based upon discussions with RDG's Review of Charges Executive Group, informed by CEPA's analysis in the initial assessments. Table 3.2 summarises the reasons for selection of the seven options chosen.

The selection of the seven options should not be interpreted as representing an industry consensus that **any** of these options represent an improvement on the present charging system. Similarly, the fact that an option was not selected for further assessment should not be seen as a rejection of that option, but rather that it did not merit more detailed investigation at this stage.

The seven options are those where RDG considered that there was merit in undertaking more detailed analysis because:

- the option scored well in the initial assessment, and RDG wanted to explore the opportunities of the option further; or
- RDG thought that the option was likely to be considered in the next periodic review and wanted to set out industry views, supported by further evidence, to inform the debate in PR18.

The options considered for detailed assessment include three network charging options. These were selected because they address known issues related to the allocation of fixed costs and scarce capacity. There are also two performance regime options and two which relate to the possessions regime. These were selected, in the main because they investigate the relationship between the infrastructure manager and the operators.

No stations charges options were taken forward for more detailed assessment. The group considered that the most significant issues in relation to stations were structural and contractual. Therefore, whilst charges are an important consideration for stations, the group thought that the other issues should be addressed before seeking significant reform of stations charges. The three stations charges options considered in the initial assessment were discussed with a working group reviewing stations charging, which agreed with these conclusions.¹²

¹² RDG (Oct 2015) "Review of Charges: Stations Charges"

Area	Option	Summary								
Network charges	1. Avoidable cost (Option 1)	• Avoidable cost provides an alternative means by which to allocate fixed cost. It fits well with ongoing work being undertaken by Brockley Consulting for Network Rail that is assessing the ability to allocate costs on this basis. Although it was graded amber by the initial assessment in the current SoW, it has more positive attributes in other SoWs.								
		 It was considered that this option would be developed as a mark-up which is why the initial assessment option of marks-ups was not selected for further assessment. 								
	2. Administered scarcity charge (Option 4) noting	• The scarcity-based options are relevant to the ongoing debate about how to allocate capacity. They consider alternative approaches to this in part to prompt debate about which might be most effective.								
	linkages to Geographically Disaggregated VUC	• Both options have the potential to add value in the current SoW, although benefits may be greater in SoWs which encourage greater competition.								
	(Option 9) and differences to LRMC (Option 3)	• As is the case for avoidable cost, these options also consider elements of other initial assessments that were not progressed to this stage of the analysis e.g. scarcity charge options could fit well with further								
	 Reservation charge (Option 7) 	geographic disaggregation of the VUC.								
Performance regime	4. Reset benchmarks more frequently (Option 15)	• Selected for detailed assessment given its potential to address the current issues with the capacity charge: the capacity charge's link to the Schedule 8 performance regime was not considered to be sufficiently clear in name or application.								
		• This option was initially rated amber, whilst a more granular and rebranded version of the capacity charge was rated green.								
		 However RDG and industry participants considered that the issues with the current charge are significant and require a wholesale re-evaluation of the approach rather than a degree of 'tweaking'. 								
		• Other 9stakeholders were concerned about the cost redistribution effect of removing the capacity charge. We considered that these can be addressed straightforwardly and that this would be considered further in the detailed assessment of this option.								
	5. Recover end-user compensation (Option 18)	• This option was selected for detailed assessment given that the performance regime is not considered to give adequate attention to the short-term impact of delays which require passenger operators to								

Table 3.2: Summary of options selected for detailed assessment

Area	Option	Summary								
		provide passengers with compensation.								
		• Passenger compensation is currently undergoing a series of changes, with the move to Delay Repay and C2C introducing automatic refunds in 2016, industry therefore considers that it is an appropriate time to consider how passenger compensation requirements could be incorporated into Schedule 8.								
		• CEPA initially gave preference to the revenue sharing option but industry representatives were of the view that this was considered in PR13. There was a strong preference to undertake further analysis of end user compensation given the current profile of this issue so this item was added to the list of detailed assessments and revenue sharing was removed.								
Possessions regime	6. More frequent ACS calculation (Option 19)	• This option was selected for detailed assessment as it addresses concerns in the industry about over- recovery of Schedule 4 costs by Network Rail when the workplan used to set the ACS at the periodic review subsequently changes resulting in a lower number of actual possessions being taken.								
	7. Reform Schedule 4 discounts (Option 22)	• This option involves reforming the notification discount factors applied to Schedule 4 compensation rates when possessions are booked more than a given amount of time in advance of taking place. The option was selected for detailed assessment as a result of industry concerns that the current discount structure may incentivise early notification of possessions but not the efficient planning of work.								
		• It is also considered an area where the approach may not have kept up to date with the fact that passengers now have better information and more immediate access to information about timetables and impact of engineering works.								
		• The option was rated amber by CEPA because the impacts are more difficult to estimate and or may depend on how the option is implemented; this option would need to be part of and consistent with the wider possessions planning regime if it is to be effective. Despite reservations about the scale of benefit it was considered, by industry to be an area worthy of further analysis.								

4. SUMMARY OF DETAILED ASSESSMENTS

Full detailed assessments of the seven options considered in this report are provided in Annexes B to H. This chapter summarises the key assumptions and findings from the detailed analysis.

4.1. Scope of this chapter

This chapter provides summaries for each of the seven detailed assessments. The summaries are organised into sections as shown in Table 4.1.

Table 4.1: Detailed assessment options

Network charges	Performance	Possessions
 Avoidable cost Administered scarcity charge Reservation charge 	 4. Reset benchmarks more frequently 5. Recover end-user compensation 	 6. More frequent ACS calculation 7. Reform Schedule 4 discounts

Each summary draws out key elements from the full detailed assessments provided in Annexes B to H. In each case, we provide information on:

- the underlying issue in the regime that the option seeks to address;
- what we envisage the option would change or add to the regime;
- key impacts identified in the detailed assessment;
- a summary of our findings from the detailed assessment; and
- the option's overall grading against the RDG Vision across the States of the World considered.

At the end of each summary, we also highlight the main observations provided by industry stakeholders during this process. Further details on this and on the other points highlighted above can be found in the full detailed assessments, provided in Annexes B to H.

4.2. Network charge options

This section contains summaries for the following options, noting the annex to this report in which full details can be found:

- Fixed charge based on avoidable cost (Annex B);
- Administered scarcity charge (Annex C); and
- Path reservation charge (Annex D).

4.2.1. Fixed charge based on avoidable cost

Avoidable cost charging is an option to set charges used to recover Network Rail's net revenue requirement in a way that is more reflective of underlying costs than the existing Fixed Track Access Charges (FTACs), and with greater clarity of purpose.

Industry participants supported this option being selected for further investigation given its potential to send more informative price signals to train operators and their funders in the context of anticipated changes to money flows, i.e. a significant reduction in the Network Grant. We estimate that changes in money flows could result in an increase in charges to recover the net revenue requirement, putting them at almost ten times their current level¹³ and increasing the influence of the approach used to calculate them. It was also selected for detailed assessment to investigate freight sector concerns about the implications of using avoidable cost information in charges.

For this option, we envisage that the current FTACs would be replaced with a new charge based on long-run incremental cost (LRIC) principles. This would produce a set of tailored charges based on causal links between train services and infrastructure costs.¹⁴ They would be highly disaggregated by geography and by the specific demands train services place on the network. We assume in this detailed assessment that this option could be implemented as a "mark-up" consistent with the terminology of Commission Implementing Regulation (EU) 2015/909. Further legal analysis is required in this area but given its potential role in protecting the financial viability of non-franchised operators, we note that this is potentially a crucial assumption in the analysis.

The current FTAC methodology, which allocates costs based on traffic metrics (e.g. train km), results in charges that recover costs in areas where traffic is greatest. As shown in the table below, an avoidable cost methodology has the potential to be much richer.

	FTAC	Avoidable cost charge
Increase with own traffic	Yes	Yes
Capture utilisation of assets	No	Yes
Capture costliness of assets used	No	Yes
Capture cost of user-specific demands	No	Yes

Table 4.2: Comparison of FTAC and avoidable cost charging

Indicative modelling completed in support of this assessment uncovered potential for large shifts in charges between operators. In particular, it might be possible to see shifts between train services operating on busy lines where economies of scale are realised to those where

¹³ £4.4bn per year (2012/13 prices).

¹⁴ We envisage that this option could be implemented as a "mark-up," potentially affording "ability to bear" protections to more vulnerable operators but further legal analysis is required in this area.

they are not. We also saw potential to attribute avoidable costs to freight operators, which could be substantial, but found they would require significant protection from them, with estimated charge increases (before considering any potential "mark-up" protections afforded by EU legislation) easily outstripping current profits.

Overall, we found that while there could be informational benefits from investigating avoidable costs, this should not be conflated with those benefits that might arise through charging. In the current SoW, we anticipate minimal benefit from putting avoidable cost information into charges, given the nature of current franchising arrangements and decision making.

It is not clear that there would be sufficient benefit from this option to outweigh the nontrivial burden of the supporting calculations, and to calibrate the level of the mark-up to ensure the viability of open access and freight operators. Despite the weak overall performance of this option against the RDG Vision in the current SoW, we did identify potential for this option to be beneficial in four alternative SoWs (indicated above), making this option far more attractive if the sector were to move in those directions.

Limiting the scope of the charge to franchised passenger operators, it is possible to envisage some elements being introduced at PR18. For wider application, we anticipate far more time being needed to get the supporting framework right, at which point we might be in quite a different SoW.

Table 4.3: Avoidable cost

Overall CEPA grading of performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
=	++	+	=	=	+	+	=	

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was some consensus on the investigation of avoidable cost information as a route to improving understanding of Network Rail's costs but there was only limited support for using such information in charges, particularly in the current SoW. Despite this, some passenger operators saw this as means to make the FTAC more rational, particularly if combined with the removal of the capacity charge, which was seen to allocate fixed charges in a similarly arbitrary manner.
- Passenger operators expressed the view that it is important to distinguish between informational benefits and those from using avoidable cost information in charges. They saw minimal benefit from avoidable cost charging in the current SoW particularly given the burden of calculating the charges and ensuring that operators without public service contract income would be able to pay the charge and remain financially viable.
- The issue of ensuring the viability of freight operators was shared by all industry participants, with some passenger operators expressing the view that this issue must be explicitly addressed as it is an inevitable requirement. They also expressed the view that it is essential for further work to be done to close down the legal uncertainties of this option either at this stage of policy development or at least before implementation. A particular legal issue for many parties was the status of the charge, a "mark-up," "directly incurred cost" or something else.
- Passenger operators considered it important to address the impact on service sponsors and funders, particularly those associated with regional public service obligation (PSO) contract.
- Open access operators expressed the view that there is a strong link between the existing charges structure and processes for capacity allocation and Network access processes. Existing Open Access operators do not pay FTAC with one rationale being that this reflects the different markets served and the fact that they access to the network in a different way. In particular, Franchised Operators are able to access the whole of the relevant passenger market while the Open Access operator access is controlled by the Not Primarily Abstractive (NPA) test amongst other mechanisms. If Open Access operators were to face an avoidable cost charge, there would need to be a parallel adjustment to the Capacity Allocation process.
- Transport Scotland explained that this option is only significant to Scotland if there is a move away from the current model where all Scottish fixed charges are allocated to the ScotRail franchise.
- Network Rail considers that the current approach that is used to allocate Network Rail's 'fixed' costs is too simplistic; it could be improved to better reflect the actual underlying railway economics. It considers that the informational benefits of this could be powerful, even if it were concluded not to fully reflect this in charges.

4.2.2. Administered scarcity charge

The purpose of a scarcity charge is to encourage the release of lower value uses from scarce capacity and enable higher value uses to take over that capacity.¹⁵ Many important parts of the railway network are used to full capacity, and in some cases there are valuable additional uses that are prevented, or required to use inferior timings or routings. An additional variable charge, applied only to 'scarce' parts of the network, could encourage better use of that scarce capacity, e.g. by discouraging low value uses of the network or by encouraging Network Rail to find a way of scheduling more trains.

We have carried out a detailed assessment of an administered, value-based, scarcity charge because it appears to be the most practical form of this type of charge. The alternative to administered is market-determined, but that appeared impractical given the difficulty of devising workable mechanisms consistent with the complexity of railway scheduling. The alternative to value-based is cost-based, but that appeared too complex and less relevant.

A reservation charge, which has also assessed in more detail, is a different type of charge related to efficient use of network capacity. A reservation charge and administered scarcity charge could both exist in the same charges regime.

The scarcity charge would be an additional variable charge, i.e. levied in addition to the variable usage charge. Our interpretation of European legislation suggests that it permits a scarcity charge, which is not a mark-up, in addition to the cost directly incurred within the charge for the minimum access package. It would therefore not be legally limited by ability to pay. It does not have to be restricted to locations formally declared as congested infrastructure.

The definition of value we are using is the commercial opportunity cost, the net income lost by putting a scarce resource to one use rather than another use that has been excluded. We initially suggested the opportunity cost could reflect full economic value, which would include non-financial socio-economic values. However, our further work suggests that it is more practical and appropriate, at least in the first instance, and particularly in the present state of the world, to use the 'commercial' opportunity cost. There is precedent for such an approach in other industries, and was the basis of earlier studies for ORR on its application, and presents opportunities to keep it simple and practical, through being narrow in application.

To focus on the key issues and minimise potential issues of feasibility and complication, we have envisaged a simple scarcity charge that would apply as a path charge to only the busiest and most commercial corridors, with a high peak charge, a much lower off-peak charge, and no charge at times of day when there is spare capacity. We assume that the path charge would apply to all services passing through the key bottleneck of the path. In

 $^{^{\}rm 15}$ What is meant by "value" in this context is discussed later in this section.

this simple vision of the charge, many, or even most, locations in the network, where the infrastructure is used to capacity, would not attract the charge, because such is the level of non-commercial services operating there that capacity would not be said to be the prime constraint on expanding commercial operations. It would not correspond to the technical EU definition of "congested infrastructure."

Clearly a more detailed scarcity charge could be implemented, charging many more nodes according to their relative scarcities. For example, the definition of opportunity cost could be extended to include social value but would complicate the calculation and application of the charge. Nevertheless, we can note that there are options for a more sophisticated charge if the simple charge works well and is deemed worthy of extension.

We have assessed that in the current SoW such an administered scarcity charge could have positive benefits in incentivising more remunerative use of existing capacity among commercial users of that capacity. Whilst discouraging uses of low commercial value, it would not entirely resolve policy issues in relation to competition between open access and franchised operators for commercially profitable services. This is because it does not address issues relating to abstraction and cross-subsidy of subsidised services.

This charge would also have important effects at the boundary of commercial and subsidised services, where they share the commercially valuable capacity, making funders decide whether the social benefit of their funded service justified paying the premium. Some might consider this an inappropriate way of making trade-offs between commercially beneficial and socially beneficial uses of infrastructure. For services that provide social benefits but are not procured by funders e.g. freight services, they may not be able to accessing funding to offset the effect of this charge.

In alternative SoWs which introduce greater on-rail competition, or which allow train operators to play a greater role in capacity allocation, this option could have greater positive impact. In SoWs where train operators are more protected from charges, or where funders play a greater role in capacity allocation, it would be less relevant.

Any 'premium' charge has the effect of increasing the price of access and reducing demand. Therefore, if the current capacity charge remained, we would not expect the scarcity charge to be added to the capacity charge, and it would be inappropriate to do so. In locations where both a capacity charge and a scarcity charge applied, the actual charge would be the larger of the two.

A scarcity charge potentially works well with a geographically distinguished variable usage charge (VUC). Previous work by ORR on a geographically disaggregated VUC suggested it would be low in intensively used areas, and thus a countervailing charge representing that intense use would be useful to avoid further concentrating use in those areas. In practice the simple and focused scarcity charge we have envisaged as most practical to implement would only have this countervailing effect in a few locations of particularly high scarcity. Ultimately a more sophisticated scarcity charge would work best with a more granular VUC.

Table 4.4: Administered scarcity charge

Overall CEPA grading of performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
+	++	++	-	+	+	++	+	

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Several industry representatives suggested that socio-economic as well as commercial value should be included in the measure of scarcity value. However, freight and passenger operators raised a some potential risks with the introduction of a scarcity charge:
 - Some operators argued that there was some uncertainty regarding the assumption made in this report that a scarcity charge could be a "mark-up" such that ability to pay would be taken into account in its application. This uncertainty in part come from scarcity not being included as an admissible cost in the implementing regulation 2015/909, which defines costs directly incurred. Some also expressed a degree of uncertainty regarding the assumption made in this report that under Directive 2012/34, a scarcity charge could apply beyond locations formally declared as congested infrastructure. Both these points of legal uncertainty would need to be clarified ahead of implementing such a charge.
 - Using price to encourage reallocation of capacity between funded and commercial services would be politically controversial.
 - The binary nature of the proposed scarcity charge could lead to a high degree of instability as the route flipped between being "full" and having a spare path.
 - It could lead to gaming by a dominant operator filing a route to make it scarce and impose charges on competitors.
 - If the scarcity charge applies to freight and Open Access operators, there would need to be a parallel adjustment to the capacity allocation process to make them freer to obtain paths they are willing to pay for.
 - There would be boundary problems if the charge were to change by a large amount at a sharp time boundary, and difficulties if a path were re-timed to the expensive side of the boundary.
- Network Rail considers that there could be merit in carrying out more work to better understand the full economic value (i.e. societal and commercial) of each train path. They would be concerned about pricing off traffic of high social value' if a purely commercial value approach were adopted.
- Transport Scotland observed that the application of a scarcity charge on the West Coast Main Line (WCML) and the East Coast Main Line (ECML) could result in disproportionately increased service funding costs for Transport Scotland, given the funding treatment of cross-border services and allocation of the FTAC. Transport Scotland has a tightly defined public service specification and does not anticipate changing that policy.

4.2.3. Path reservation charge

A reservation charge is non-refundable charge for booking capacity. It is an alternative kind of scarcity charge whose purpose is to discourage users from booking capacity that they do not in practice require. Such a charge might encourage operators to manage their network usage more efficiently and discourage booking capacity beyond what they predict will be required, except to the extent a firm option of being able to run is sufficiently valuable to them to pay the charge.

The reservation charge we considered for the purpose of this assessment was a **deposit** (or obligation) based scheme payable only for **firm access rights**. It can be defined equivalently either as a non-returnable deposit which is then set off against charges for paths actually operated, or else as a charge for rights not used. We assume the reservation charge being assessed as a fee **per train km**.

While it is not what EU law would define as a "cost directly incurred," basing the reservation charge on the expected VUC would be a simple way to calculate a per km charge, which is explained in the *Implementation* section of the detailed assessment template. We have therefore used the VUC as the basis for our modelling and have assumed a separate rate would apply to passenger operators and freight.

This charge most strongly impacts those train operators that have low utilisation levels for their booked capacity, and evidence suggests that this is most significant for train operators running bulk freight services. Other train operators tend to have high levels of utilisation of allocated capacity. The chart below shows the modelled relative impact of the reservation



charge on total charges paid by stylised operators (compared to bulk freight).

In the current SoW, reservation charges could have some positive benefits in incentivising better use of existing capacity but in some forms could significantly adversely impact freight. Impacts on franchised passenger operators under the current regime are likely to be small given that the current arrangements provide protection from change and limit the scope to change

service levels. This form of scarcity management has some precedent given that it is in use on HS1, in France and Germany and is being considered for Crossrail.

Assuming adverse impacts could be managed, such that freight (and potentially other smaller operators) retains the flexibility required, the overall benefit could be positive but may be small. More detailed analysis would be required to assess whether likely benefits

outweigh issues such the transaction costs involved in introducing and managing a reservations system.

In alternative SoWs that introduce greater on-rail competition or which place greater emphasis on the value of capacity this option could have greater positive impact (although it would need to be weighed against other options for value based charging). In SoW where operators have increased protection from change, the option would have less impact.

Table 4.5: Path	reservation	charge
-----------------	-------------	--------

Overall CEPA grading of performance against the RDG Vision in each SoW									
Current Sow	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
+	+	+	+	=	+	+	+		

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was consensus in the industry and Government that a path reservation charge would almost exclusively 'bite' upon freight operators, raising a wide range of concerns that it could be impractical given the nature of the market within which they operate.
- Freight operators require some flexibility in their access due to variability in demand. However, freight operators noted that passenger operators have paths in expectation of demand, whereas freight operators react to demand, which is why freight operators have a lower utilisation rate.
- *Reservation charges in other countries (e.g. France) are thought by freight operators to have led to a significant decrease in rail freight volumes and reduction in quality of service.*
- Work on the holding and management of freight capacity has been ongoing since 2007. This has contributed to a significant release of access rights by the rail freight sector (information from Network Rail and freight indicate that approximately 20% of access rights have been released since 2014).
- A charge could encourage short-term booking of paths, which would add to operational costs. Short term planning may also pose a significant risk to performance
- The charge would have to be designed to ensure that
 - o customers and operators are not punished for unmanageable variation in demand;
 - o there is not incentive for operators to run shorter trains; and
 - o small operators and new entrants are not penalised.
- Freight operators noted that with more freight operators in the market there are likely to be more paths required but not used.
- Passenger operators did not expect that this option would have any material impact, as the real issue is capacity allocation and timetable development. Unless freight operators were charged for the paths "reserved" for them in the timetabling process, nothing would actually be changed by the introduction of the charge.
- Freight operators suggested that transaction costs are likely to be far too high for the level of benefit that could be realised.
- Network Rail commented that as the network becomes more 'full' it will become increasingly important that all mechanisms that could improve utilisation are considered. However, there has already been a significant amount of work undertaken by Network Rail and freight operators to determine paths that could be 'given back' for other uses. Network Rail would however be

concerned if the charge led to freight operators making greater use of short term planning resources to gain access to the network, in seeking to avoid a reservation charge.

4.3. Performance incentive options

This section contains summaries for the following options, noting the annex to this report in which full details can be found:

- Reset Schedule 8 performance benchmarks more frequently for changes in traffic volumes (Annex E); and
- Recover end-user compensation through Schedule 8 (Annex F).

4.3.1. Recalculate benchmarks for traffic changes and remove capacity charge

Schedule 8, the performance regime, is in place to provide train operators with compensation for disruptions to their services and therefore to their business. This option is envisaged to update Network Rail's Schedule 8 performance benchmark annually to take account of traffic growth (or decline), as an alternative to the current separate 'capacity charge' which aims to reimburse Network Rail for its traffic-related portion of Schedule 8 compensation payments. The capacity charge's current link to Schedule 8 is not obvious in its name: it is sometimes wrongly assumed to be similar to a 'scarcity charge'. The capacity charge is not understood well by the industry or considered to be cost reflective, and therefore rather than explore opportunities to make the charge more granular or adjust its calculation in any other way, this option envisages removing the charge entirely.

Network Rail's Schedule 8 benchmark is set at the Periodic Review preceding the control period. The Periodic Review also outlines updates to the benchmark each year of the control period to reflect the changes in Network Rail's performance targets. This option proposes that Network Rail's benchmark also includes an annual update to account for the actual increases or decreases in traffic (using the latest year of available data each time), which could allow for the removal of the capacity charge. The capacity charge is currently intended to recover the amount of additional compensation paid out by Network Rail through Schedule 8 due to increases in network traffic making it more difficult for Network Rail to recover services after a delay, and is levied per actual train mile at an ex ante tariff rate set at the Price Review.

There are a range of options for change to address the current issues with the capacity charge, and this option is envisioned with the following key characteristics:

• Annual updates under this option are intended to 'smooth' the effect of traffic growth, as the benchmark would be updated a little each year to take account of the level of traffic.

- The industry's understanding of the charge and its intentions could be improved by better integrating the recovery of Schedule 8 Congestion-Related Reactionary Delay (CRRD) costs into Schedule 8 itself.
- The method of updating the benchmark would ideally be simple and mechanistic, to ease understanding and calculation, therefore also minimising transaction costs.

There are some benefits to this option, mainly ensuring that the Schedule 8 benchmark is more reflective of the appropriate achievable level of performance and the likely improvement in industry understanding that could result. However, there will also be costs involved in the initial calculation of the appropriate adjustment to the benchmark, which would likely require a level of resources comparable to that required to recalibrate the capacity charge at CP5. Subsequent annual updates could be devised as simple mechanical calculations, reducing the complexity and cost of implementation.

This change would reduce the variable charges that operators pay, and hence Network Rail's income from track charges when new capacity is put into operation. This also reduces the marginal cost of running an additional train. The capacity charge reflects, when levied on additional trains (introduced after the start of a Control Period), the marginal performance costs of running additional services. The charge does not reflect this perfectly as it does not adjust for time periods such as peak and off-peak, only for weekend vs weekday services, so the rate is averaged within those two. It also has a single flat rate across the network for freight and charter operators so does not reflect geographic differences in the marginal performance cost of running additional services. Nonetheless, the capacity charge at present, to some extent, discourages additional services running where it would worsen congestion, even though this is not its primary purpose. Removing the capacity charge might result in some additional trains that might not otherwise have run, which could be considered a less efficient outcome in terms of use of network capacity (the cost to other running trains of the congestion that additional train running will cause may be higher than the benefits the additional train itself will bring).

Without other changes to the regime, we expect that revenue previously recovered through the capacity charge, which is paid by all operators, would instead be recovered through the FTAC, paid only by franchised passenger operators. This would lead to a redistribution of charging income from open access and freight operators to franchised passenger operators, and also some redistribution between franchised operators. This is because the capacity charge is offset against the FTAC at present. This redistribution could also affect the allocation of cost across administrative boundaries, due to the current arrangement of cross-border services between England and Scotland.¹⁶ However, the redistribution across boundaries is likely to be relatively small.

¹⁶ This arrangement means that English franchises pay the capacity charge but not the FTAC for use of the Scottish areas of the network, and vice versa. However Transport Scotland state that this is unbalanced against Scotland.

We note that there are options, such as a scarcity charge, which could be implemented alongside this change to the regime to reverse some of the redistribution effects, and also to provide more explicit and targeted incentives around use of capacity than the capacity charge does at present.

Overall, we have rated the impact of this option as slightly positive in many SoWs, as we don't expect impacts to differ materially between SoWs.

Table 4.6: Recalculate	benchmarks fo	or traffic changes	and remove c	apacity charge
------------------------	---------------	--------------------	--------------	----------------

Overall CEPA grading of performance against the RDG Vision in each SoW									
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
+	+	+	+	+	+	+	+		

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Funders and passenger train operators raised concerns that annual adjustments would take time and resources. This option has therefore been envisaged as a simple and mechanistic adjustment which should reduce the transaction costs involved while having the benefit of improving the transparency of the relationship between performance and variations in traffic levels. Network Rail noted that a similar mechanistic approach is already in place for freight operators' benchmarks, and that seems to reflect the impacts of changes in traffic well without being overly complex.
- The current capacity charge level depends in part on the use of the network by other operators, but the concern raised by the freight sector is that this option would increase that effect since the current capacity charge is levied per train mile actually travelled by the operator. In contrast, this option would adjust Network Rail's benchmark for all operators in that service group. A further concern raised by passenger operators is that changing Network Rail's benchmarks may have knock-on effects on TOC benchmarks given that it will adjust the level of delay that is considered 'efficient.'
- DfT were concerned that by creating more frequent opportunities to change the benchmarks and therefore the terms of the franchised operators' contracts, there would be more frequent requests by franchised operators for renegotiation of the contracts. Assuming the impact is sufficient to be a Qualifying Change this would place additional resource requirements on both DfT and the franchised operators. It may therefore be useful to consider whether the mechanistic process employed for the charge could be extended to dealing with impacts on the franchise agreements.
- DfT and Transport Scotland expressed concerns that this option could cause a 'shift' in costs to Transport Scotland due to the arrangements that cross-border services currently have in place with regards to the capacity charge and FTAC. Currently, cross-border services pay the FTAC in the operator's country only, and pay the capacity charge relevant to the route regardless of country. Therefore, removing the capacity charge would remove the element of charges that cross-border services pay to use the infrastructure outside of its own country.

4.3.2. Recover end-user compensation through the performance regime (Schedule 8)

Schedule 8 reimburses train operators for estimated revenue losses as a result of disruptions to its services, given that passengers may choose not to travel in future due to the reputational impact of disruptions to service. This option was selected for detailed assessment given that the performance regime does not provide compensation to train operators that are required to provide delay compensation to their end-users. This option particularly looks at passenger operators as these are required by contract to provide end-user compensation, and also recognises that it might be appropriate to include an element for freight operators.

A key objective of this option is to make the intention of the performance regime clearer to the public, as there is currently a high level of misunderstanding in assuming that the Schedule 8 compensation for passenger operators should be passed onto passengers as delay compensation, rather than its actual intention to compensate train operators for the longer term revenue impact of delay (and some short-term impacts such as replacement buses). Given that passenger compensation is currently undergoing a series of changes, it is an appropriate time to consider how passenger compensation requirements could be incorporated into the performance regime. It is important, at the same time, to consider whether an appropriate end-user compensation mechanism for freight could be implemented given that freight operators often compensate their end-users for delays.

Schedule 8 currently compensates train operators for the longer-term financial impact of disruptions to services. There is no component to reimburse train operators for end-user compensation (i.e. passenger compensation paid through Delay Repay, or freight operators' case-by-case contract); this option proposes introducing such a component in addition to the current Schedule 8 compensation.

This option aims to redistribute the risk associated with end-user delay compensation. Currently in passenger services all of the risk of delay compensation is on the passenger operators, despite each type of passenger operator causing only between 10 and 30 percent of their own delays. While franchised passenger operators do include a 'risk premium' for this into their franchise bid value, they are then at risk for any deviations from their estimate.

This option has several clear benefits. It would be likely to reduce incentives on passenger operators to limit compensation paid to passengers. It also aligns the cost of delays more closely to those who caused them by 'passing through' an element of passenger compensation to the at-fault party. The main concerns relate to its practicality and transaction costs. In particular, the effectiveness of this option relies on passenger and freight operators passing through compensation to their end-users. This might be particularly difficult to measure in freight where explicit compensation payments might be foregone to instead charge customers lower prices to account for the risk of potential delay. There is typically a low claim rate among eligible delayed passengers (no comparable freight

data is available), often attributed to a lack of awareness driven by train companies not actively providing passengers with the relevant information. Use of delay claims processes, such as cash refunds and online forms, make it likely that there will be an increase in the rate of claims by those eligible. Therefore, an automatic refund might provide an effective delivery mechanism to ensure that there is not a large disparity between the amount claimed by passengers and the amount passed through Schedule 8 (at a lower level of transaction costs than some manual methods would bring).

There are difficulties associated with applying this option to freight, since there are not standard compensation terms for freight operation, and it could be discriminatory to make use of the varying and confidential terms in contracts. In principle, standard rates could be applied according to some categorisation. But it would also impact upon the economic viability of freight if this resulted in increased payment rates to cover the cost of the scheme – freight operators may prefer to take the risk themselves and pay lower payment rates. Nevertheless, implementing this option will need to ensure that Schedule 8 is not unduly discriminatory against freight: freight should be included in the two-way payments if it is deemed appropriate. This option assumes that it is possible to determine an appropriate mechanistic calculation for freight end-user compensation, however it is not certain that freight would desire the additional complexity that even a simple mechanistic approach would bring when freight operators are currently able to negotiate the issue through their contracts individually with each customer.

While our analysis has showed an overall positive grading of this option, the magnitude is unclear, and further work would be required on reasonable practicality and cost. Train operators might also oppose the added complexity that this might bring to Schedule 8 (see RDG Phase 2b Feature 8.12). This option will, in several areas, have a stronger impact (negative or positive as appropriate) in the SoW where there will be greater flexibility to train operators (Dynamic railway and On-rail competition). However, these extremes have balanced each other out to mean that these SoW are given a small positive grading as with the other SoWs.

 Table 4.7: Recover end-user compensation through the performance regime (Schedule 8)

Overall CEPA grading of performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
+	++	++	+	+	+	+	+	

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was some consensus on considering this option, as something which might improve the passenger perception of the industry. The intention of the performance regime, passenger operators noted, could be clearer to both industry and passengers.
- It was noted by the freight sector, but applies to both freight and passenger operators, that increasing the costs of delaying a service increases the expected cost of running an additional service if it is assumed that an additional service might at some point delay other operators. However, this is part of the incentive impact that is intended with the redistribution of risk through this option: this option envisages that operators would be at risk for the delays that they cause to other services.
- It would be more difficult to implement this option for freight operators. Freight contracts with
 their customers are confidential, one-to-one, and likely to vary widely depending on the type of
 customer (some types of commodity may be more affected by delays to transportation than
 others). Furthermore, freight operators are able to choose to forgo a compensation clause in a
 contract to instead charge their customers lower prices to account for the risk of potential delay.
- Franchised passenger operators will have included in their franchise bids a risk premium to reflect the uncertain level of passenger compensation they will need to provide. This option aims to reduce that risk premium in new franchises but given the staggered franchising process there would be a time lag until it was accounted for in all contracts. There could be some transaction costs and transitional issues if ORR tried to implement this option without double-reimbursing franchised operators.
- Passenger operators asked whether this option fits within Schedule 8, or whether it might be better placed as part of a wider review of passenger compensation. They also questioned the extent to which this option would impact Network Rail's incentives to prevent delay, as it is small in magnitude compared to overall business and Schedule 8.

4.4. Possessions incentive options

This section contains summaries for the following options, noting the annex to this report in which full details can be found:

- More frequent Access Charge Supplement calculation (Annex G); and
- Reform Schedule 4 discount structure for notice period of possession (Annex H).

4.4.1. More frequent Access Charge Supplement calculation

The Access Charge Supplement (ACS) allows Network Rail to recover the amount it is expected to pay out in passenger Schedule 4 compensation over the control period, should it undertake the estimated level of works efficiently. The ACS is set at the start of a price control period based on the estimated volume of engineering works (maintenance and renewals) that will be carried out during that control period and a Schedule 4 unit cost for each type of activity estimated from historical data.

This option was selected for detailed assessment as it addresses some concerns in the industry that Network Rail has tended to delay work relative to the anticipated programme, thus taking fewer possessions than scheduled, and resulting in over-recovery of costs relative to the ACS set at the periodic review.

We envisage that the most likely possibility for implementing this option involves recalculating the ACS annually based on changes in the volume of engineering works planned, as more accurate predictions become available. Our study of this option, as we have envisaged it, suggests that it brings benefits primarily in terms of better reflecting Network Rail's efficient Schedule 4 costs based on the works carried out, and avoiding the risk that Network Rail will benefit from deferring or cancelling maintenance and renewals works. The main downside of the option would appear to be the potential difficulty with concluding whether variations in activity and expenditure volumes in any particular year are efficient, relative to ORR's Final Determination (the ACS is calculated to reflect the efficient level of work that Network Rail must undertake to maintain and renew the network in the control period). However Network Rail has already started reporting in its Regulatory Accounts the amount of financial under/over performance related to variation in the volume of works. This suggests the information necessary for recalculating the ACS on a more frequent basis should be available.

ACS reform potentially reduces the problem of Network Rail's over-recovery of costs in an environment where it under-delivers on the volume of work. This option has a direct impact only on franchised passenger operators as they are the only train operators to pay the ACS at the moment. But in the present SoW any potentially useful effect in reducing operator costs by reducing their risk exposure is limited, because of protection against changes in charges in existing franchise arrangements.

The impact on specific (types of) operator will depend on the extent to which the planned
activities set out at the periodic review subsequently change on different routes. Train operators using a route where the planned activity volumes are delivered according to the initial plan would not see their ACS change (at least in relation to operations on that particular route) while operators on a route where a significant portion of activities are being deferred would benefit from a reduction in their ACS during the control period.

To some extent, the benefit that the option brings in terms of better reflecting efficient Schedule 4 costs based on volume of works carried out by Network Rail is counterbalanced by the costs associated with added complexity and volatility of charges within the price control period in the current SoW. However, even if franchised operators are less impacted by this option in this SoW, the option would still result in a better allocation of costs between funders and Network Rail. The proposed option would bring additional benefits in the "Dynamic railway" SoW where franchised passenger operators are exposed to changes in access charges. In that case, a more frequent recalculation of the ACS reflecting volume of work actually carried out would reduce the financial risk exposure of passenger operators and potentially have a positive impact on the value for money of franchises and passenger fares.

Table 4.8: More f	frequent Access	Charge Supplement	calculation
-------------------	-----------------	-------------------	-------------

Overall CEPA grading of performance against the RDG Vision in each SoW							
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
+	++	+	+	+	+	+	+

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Stakeholders have generally commented that the benefits of the option for train operators are unlikely to be significant in the current SoW but that under-delivery of planned activity volumes leading to over-recovery of Schedule 4 costs has been an issue in recent years. In particular:
 - Some stakeholders remarked that the difference between the Schedule 4 ACS paid and the compensation received was material.
 - Open access operators commented that they do not currently see the benefit of paying the ACS in return for full Schedule 4 compensation.
- Train operators also raised general concerns regarding the benefits of having the ACS and stated the need to establish greater clarity as to its purpose before moving on to assess any reform options.
- Passenger operators commented that a more general objectives-driven review of the possessions regime is needed; and the benefits of this option would be counterbalanced by increased complexity and volatility of charges.
- Network Rail considers that it is important that the ACS reflects the efficient level of compensation Network Rail expects to pay to train operators, as a result of undertaking engineering work to maintain and renew the network.
- Funders considered that whilst this option addresses a recognised issue and could bring some benefits, a yearly change in the ACS would result in yearly adjustments to franchise payments increasing the negotiating burden between funders and franchised operators.

4.4.2. Reform Schedule 4 discount structure for notice period of possession

This option involves reforming the discount structure applied to Schedule 4 compensation rates paid by Network Rail to passenger train operators for revenue loss due to disruption arising from planned track possessions. The current discounts vary according to how much notice of the possession is given and reflect the existing evidence that there is a lower impact on long-term passenger demand from disruption announced well in advance.

Reforming the structure of discounts aims to address some concerns in the industry that while discounts incentivise early booking of possessions, they can be counterproductive for efficient planning of works and possessions. The discount structure provides incentives to Network Rail to plan and book possessions early, in particular more than 26 weeks in advance, when the largest discount is given. The concern is that the workplan for some of those possessions may not be fully developed at that stage, potentially resulting in inefficient use and late cancellations of possessions. Currently more than 90% of possessions are booked in this timeframe.

The discount structure also makes it costly for Network Rail to re-plan possessions, reducing the incentive for Network Rail to make changes to possession plans after the initial notification, even in cases when it would otherwise be beneficial. This option was also considered as an area where the approach may not have kept up to date with the fact that passengers now have better and more immediate access to information about timetables and impact on services of engineering works.

There are numerous ways in which such an option could be implemented, including increases, as well as reductions, in the discounts, and changes in the notice dates at which the discounts apply. We have examined several options that we envisage cover a range of possibilities for implementing this option. The scenarios considered involve:

- Reducing or removing the discounts current notification timeframes are kept but early notification discounts are reduced or removed altogether; and
- Reforming notification thresholds and discount rates notification thresholds are adjusted as well as reducing the discounts applied for early notification.

The viability of the option critically rests on how much difference early notifications make in reducing the disruptive impact of possessions and whether reducing discounts will alter Network Rail's possessions planning processes. An analysis of the impact of planned disruption on passenger demand is currently being undertaken by the Passenger Demand Forecast Council (PDFC) and is expected to conclude in early 2016. A decision on the level of discount factors should incorporate this latest evidence.

Given the current timetabling process, and particularly the requirement to have the timetable, and therefore most possessions agreed by T-12¹⁷ (the Informed Traveller timetable), reducing discounts alone is likely to have only a marginal impact on Network Rail's possessions planning process. Any reduction in discounts, particularly for the earliest timeframe, will increase total Schedule 4 compensation (and also be reflected in a higher ACS). In the current SoW, higher Schedule 4 payments would provide additional revenue to passenger train operators with the higher ACS costs being largely passed on to funders, at least for the duration of existing franchise agreements. In the "Dynamic railway" SoW, with less franchise protection, the higher ACS would be borne by train operators with the overall impact being neutral if Schedule 4 costs are in line with the baseline set at the periodic review. The measure which could potentially provide most benefits involves reconsidering the notification thresholds, for example, by setting a threshold more closely linked to passenger milestones (such as T-12 when advance tickets become available).¹⁸ These benefits are harder to quantify without a clear view of how Network Rail's possessions planning would change as a result; however introducing more flexibility by allowing Network Rail to benefit from a discount when possessions are notified in time for T-12 could bring benefits in terms of more efficient planning of possessions without increasing Schedule 4 costs significantly, if discount factors are not significantly reduced.

At the moment there are several ongoing reviews that could have an impact on the Schedule 4 discount structure including: RDG's Asset, Programme and Supply Chain Management (APSCM) sub-group's work on how possessions are planned and delivered, Network Rail's IAP¹⁹ and the PDFC's work on the impact of planned disruption on passenger demand. Any changes to possessions planning and timetabling or new evidence on passenger behaviour arising from these initiatives should be taken into consideration.

¹⁷ T-12 refers to 12 weeks before the timetable is operational.

¹⁸ It should be noted that some intercity operators may allow passengers to book tickets up to six months in advance although this is mainly for mid-week services which are less likely to be affected by possessions.

¹⁹ The **Industry Access Programme** (IAP) is a Network Rail and industry project looking at how to optimise track access including possessions for maintenance, renewals and enhancement works

					• ·
Tahle 1 9. Retarm	Schedule A	discount structure	tor notice	nerind n	tnncceccinn
Tubic 4.5. Acjoint	Juneaure +	discourt structure	joi notice	рспой ој	possession

Overall CEPA grading of performance against the RDG Vision in each SoW							
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
=	=	=	=	=	=	=	=

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Stakeholders have commented that the most important milestone to be borne in mind when setting discounts is to have possessions reflected in the Informed Traveller timetable (T-12) when tickets are first put on sale. In practice this would mean having possessions agreed prior to that deadline in order to allow possessions to be incorporated into the timetable uploaded at T-12.
- Passenger operators noted that it is important to bear in mind that the main reason for Schedule 4 discounts is related to the impact on passengers' willingness to travel and not to incentivise better planning (both possessions and work planning).
- Passenger operators also stressed the need to have clarity about the purpose of the current regime in order to assess whether a reform option is sensible or not. A more general objectives driven review of the whole performance incentive/compensation arrangements is needed.
- Network Rail noted that the Notification Discount Factors (both the level of discount and the timing thresholds) must be based on robust evidence of genuine lost future operator revenue at different notice periods. Therefore, further work to review this as part of PR18 appears appropriate. However, the inherent uncertainty for Network Rail in planning a long way ahead of a possession must also be recognised.
- Transport Scotland noted that the issue of possessions being booked early has been raised multiple times before. Any perverse incentives in the regime should be removed and aligned with the Informed traveller T-12 deadline.

5. ANALYSIS

The detailed analysis of options in this report primarily focuses on each reform being made independent of others. In this chapter, we consider how the options might fit together. It draws insights from our work on factors that affect the form and/or effectiveness of the regime.

5.1. Scope of this chapter

This section:

- considers the operation of detailed options that have been assessed both as a charging package and also as part of the current charging and incentives regime;
- reflects on the 'Factors' report²⁰ in particular on the degree of scope for change in the current SoW and how this might change in other states; and
- sets out a summary of findings from the analysis presented in this report.

5.2. Detailed assessment options as part of a package

The detailed analysis of options in this report primarily focuses on each reform being considered, in the current SoW, i.e. within the range of wider factors that affect the current regime and which are the subject of our 'Factors' report, and within the context of the current structure of charges and incentives as shown in Figure 5.1.





²⁰ CEPA (Nov 2015) "Review of factors impacting the Form and/or the Effectiveness of Charges and Incentives"

The large number of potential combinations of reforms that could be made at the same time²¹ has led the analysis presented in this report to focus primarily on the merits of each option in isolation.

As discussed further below in Section 5.2, the effectiveness of an option may depend on the SoW in question. Its effectiveness will also depend on what other charges and incentives are in place and therefore, it is relevant to use insights from the analysis of individual options to consider what packages of multiple options might perform well together. However, these packages should not be taken to represent a proposed set of changes. The packages are used to illustrate how well each option could perform with other options.

It is relatively early in the process to be considering such charging packages, not least because at the time of writing, ORR had yet to consult on its options but also because there is significant work to do to consider the precise form and mechanics of the options themselves. But to demonstrate some of the key potential interactions between the different options, in this section we present analysis of a potential package combining avoidable cost charging (option 1), an administered scarcity charge (option 4) and resetting Schedule 8 performance benchmarks more frequently for changes in traffic volumes (option 15).

These potential modifications to the charging and incentives regimes are highlighted in Figure 5.2 below. Orange boxes and text note additions to the regime, grey boxes represent charges that are potentially removed.



Figure 5.2 – Potential structure of a new charging and incentives package

²¹ There are 128 combinations of the seven options considered in this report.

In principle, the current FTAC could be replaced by a mark-up based on avoidable cost, although we note the limited impact of this option in the current state of the world and its potentially long lead time. Use of avoidable costs would however bring greater transparency and cost reflectivity to what is currently often considered as a 'balancing item' in the existing regime. It could also be used to consolidate existing freight charges, which seek to recover 'fixed costs', into a single mark-up. However, one of the potential issues of an avoidable costs based approach to fixed charges is that it would likely increase costs on more lightly used areas of the network and, as a result, potentially influence decisions as to the future usage of the network – for example leading to consideration of reduced operation of lightly used parts of the network and increased usage of already congested parts of the network. For this reason an administered scarcity charge might be complementary to this option.

Imposition of a scarcity charge could also fill the gap left by a removal of the current capacity charge achieved through updating Schedule 8 benchmarks more frequently for increases in volume, and could be an improvement on it in terms of scale, applicability and transparency. The option that we have considered would, in our view, not need to be a mark-up, since EU legislation appears to CEPA to provide for a capacity charge that is additional to the costs directly incurred, but not a mark-up, and therefore applicable to all users of the network. However, as we noted in the 'Factors' report, the legal position on this is not entirely clear and some parts of the industry consider that such a charge could only legally be a mark-up.

Introduction of a scarcity charge would also perhaps create an opportunity to revisit further geographical disaggregation of the VUC (an option considered earlier in Phase 3); although we note that the industry view is that the current VUC functions reasonably well and is well understood.

The analysis above assesses only how the options we have considered at a detailed level might fit with and / or amend the current structure of charges. There is however scope for a number of packages of options to be considered. We understand that ORR is taking this approach to its review of the charges and incentive regime and that its packages will form the basis of its consultation in December.

This report predominantly considers options in isolation, when considering what exact options might be introduced in practice, failing to consider multiple reforms creates risk. Failing to consider the wider impacts of changes to a particular charge might result in overlooking opportunities to further improve the regime and, more importantly, could fail to miss the bigger picture, disrupting the efficacy of the charges and incentives regime as a whole. The indicative package of options considered here identifies cases where options might complement each other but this may not always the case. The case for making multiple reforms must also be balanced against stability of the regime, the costs and benefits of making change and the level of complexity entailed. There is significant further work to do in this respect. To this end, we consider that this project provides a substantial body of material and analysis to facilitate the process of considering packages of charges but its scope does not extend to the detailed analysis required to consider the interactions between them and the overall financial effect. It has however been produced sufficiently early to allow ORR and the rail industry to consider these issues fully in anticipation of and during PR18.

5.3. Variations by State of the World

The current SoW introduces a range of limiting factors to the impact on the industry of many of the options that have been considered e.g. the limited impact of charges and incentives on franchised passenger operators within the current franchising regime. However, as the 'Factors' report indicates there is some scope for change. The SoWs developed by the RDG in Phase 3 have therefore been used throughout the assessment of options, allowing the assessment not to be constrained by current industry arrangements and to help identify in which SoWs they might be more or less effective.

Table 5.1 below synthesises the findings from the individual assessment of options to consider how each SoW might affect judgements of how each option performs against the RDG Vision.

SoW	Performance of options
"A more dynamic railway" (Dynamic railway)	Among network charges, both the avoidable cost and administered scarcity charge rate better in this SoW than in the current. That is because in a more dynamic railway, the parties are assumed to have more ability to respond to incentives, are less insulated from change, and properly attributing cost to them through charges is more important. We have to assume that the industry has been reshaped in this SoW to give the parties the financial strength to respond to the dynamism created by it. Specific changes to give freight resilience are specifically addressed in another SoW.
	If industry participants are more able to make decisions on what services to operate, it becomes more important to reflect scarcity. We assume that this SoW is more likely to apply for intercity services rather than local PSO-type services as funders may still wish to specify socially desirable services. The experience of Heathrow Airport, where it can take decades for low value runway uses to be reassigned to higher value uses (for example, the persistence of British Midland), indicates how administrative allocation of capacity without realistic scarcity charging can result in capacity being provided for lower value usages for extended periods. Equally, it becomes more important for genuinely avoidable network costs to be signalled to users and avoidable through charging. The reservation charge mainly affects freight and so is not rated differently in this SoW.
	The reform of ACS is rated much better in this SoW, because franchisees are less protected against changes in charges. Therefore, passenger train operators will be more concerned about the impact of accurate costing information on its own charges and incentive mechanisms.
	The changes in this SoW do not materially affect the assessment of Schedule

Table 5.1: Performance of options against the RDG Vision in alternative SoWs

SoW	Performance of options
	8 recovery of end-user compensation, and reforming possession discount regime.
"On-rail competition via flexible franchising"	Among network charges, both the avoidable cost and administered scarcity charge rate better in this SoW than in the current SoW, for the same reasons as noted in the Dynamic railway SoW.
(On-rail comp)	The reform of the ACS is not rated much better in this SoW, because franchisees are still protected against changes in charges.
	The changes in this SoW do not materially affect the assessment of Schedule 8 recovery of end-user compensation, and reforming possession discount regime.
"More highly specified franchises" (Specified franchises)	The scarcity charge option rates badly in this SoW, because there is even less ability of operators to respond to the incentives it may provide. The changes in this SoW do not materially affect the assessment of the other options.
"Freight protection/ subsidy" (Protect freight)	We have identified in the analysis of options above, particularly in the network charges options, that greater alignment of costs to their causes would tend to increase the charges for the freight operators. There is a strong perception that the rail freight sector provides substantial social benefits through removing a large quantity of heavy freight from the roads. Unless road freight is charged for the social costs it causes, it may be appropriate to provide rail freight funding for the social benefits it provides. Without a sustainable approach to freight charges, there is a risk of driving freight off the railway. Therefore, in this SoW, the options that seek to better align costs with those services that cause them receive better grades than in the current SoW.
	This SoW would help to facilitate changes in the charges and incentives regime that might be desirable. However, on its own, the overall impact of the options is not affected very much by this SoW for large parts of the railway industry. Thus the grading of options are not materially changed from the present SoW.
"Beneficiary funding of infrastructure" (Beneficiary pays)	With increased beneficiary funding of infrastructure, the proper allocation of network costs between parties becomes more important, as it comes back to the beneficiaries who will be concerned about potentially large movements in their costs and the impact this has on their budgets. Thus, this SoW is most relevant for the Avoidable Cost option, which becomes more desirable in this SoW. Other options are not materially affected by this SoW. compared to the
	current SoW.
"Change in approach to capacity allocation" (Capacity allocation)	In this SoW, both Avoidable Cost charging and Scarcity Charging are rated more highly, because of the potential that capacity may be allocated more closely to patterns of demand, and thus signals on the best use of capacity, and for cost allocation, become more important to facilitating capacity decisions, especially in situation (b) in the SoW definition, i.e. where there is a significant increase in network capacity. This SoW does not affect the appraisal of other options
"More regional	This SoW, implies a more planning-driven rather than market-driven

SoW	Performance of options
decision-making"	approach to the usage of capacity, since Passenger Transport Executives
(Regional powers)	(PTEs) tend to be more focused on this than DfT. Therefore in this SoW, scarcity charging is less relevant than it is in the current SoW.
	This SoW does not affect the appraisal of other options.

Overall the analysis that we have undertaken of both the initial and detailed options demonstrates that the current SoW considerably limits the scope for effective change of the regime. This is supported in detail by the analysis presented in the 'Factors' report. It does not mean that there is no scope for change in the current SoW. Significant informational benefits may accrue from options such as avoidable cost based charging in the current SoW and there is some scope for scarcity options to have an impact on currently congested parts of the network. The performance and possessions regime options are not as sensitive to SoW and this presents greater scope for change in those areas.

However, the "Dynamic railway", "On-rail comp" and "Capacity allocation" SoWs would appear to provide greater scope for changes to the charging regime to have significant impact on use of the network and operator behaviour. This would potentially create greater risk to smaller operators such as freight, although each of these SoWs could exist together with the SoW which provides protections to freight.

Therefore, where significant change to the charges and incentives regime are considered, it is likely that to have the greatest effect, reform would need to extend beyond the charges and incentives themselves and into the structure of the industry. Such reform would require collaboration between ORR and wider stakeholders e.g. funders, and an analysis of the costs and benefits of change that takes into account these wider impacts.

5.4. Summary

Earlier phases of RDG's Review of Charges have considered gaps in the current charges and incentives regime, and potential options for reform of the regime with the aim of addressing some of those gaps.

When developing the options for assessment, we have, as far as possible, attempted to envisage implementation of those options that maximise their practical feasibility in operation, although clearly more detailed work may be required.

The 'Review of Factors Impacting the Form and/or the Effectiveness of Charges and Incentives' report set out the current environment that the charges and incentives regime operates within. It also attempted to make some broad forecasts of how the 'factors' might change in different SoWs. This drew attention to how the current charges and incentives regime fits into the current SoW and highlighted features that should be considered when assessing options for change:

- Many potential changes to charges and incentives would have limited impact in the current SoW, because the franchising and funding environment tends to negate the effects of many changes in charges and incentives. The current industry structure tends to limit the ability of franchised operators to take financial risk, and funders place emphasis on limiting likelihood of financial distress of franchisees.
- The complexity of the industry structure creates many interconnections between different parties and different contractual arrangements. Therefore, in the current SoW, feasible adjustments to the charges and incentives regime are limited in scope because of the many connections.

Nevertheless, in the current SoW, there is still potential for some of the options considered in this report to be feasible and have some effectiveness in addressing some of the gaps in the current regime. Many of the 22 options considered in the initial assessment were appraised as being potentially of some benefit. However, if the SoW changes, this could widen the range of feasible changes to the regime.

The detailed assessment stage has focused on options that RDG representatives considered to merit more detailed consideration. The fact that an option was not assessed in more detail does not necessarily mean that it has been rejected by RDG. In most cases it reflects that further investigation of an option was not considered to be productive at this stage.

A summary of the CEPA grading from the detailed option assessments against the RDG Vision in the current SoW is provided in Table 5.2 below. Table 5.3 presents CEPA's overall grading in each SoW considered. Further detail on the individual options is provided in Annexes B to H.

Table 5.2: Overview of CEPA detailed option assessments in the current SoW

Option group	Network Charges		Perfor	mance	Possessions		
Option	Avoidable cost	Scarcity charge (administered)	Reservation charge	Reset benchmarks more frequently	Recover end- user compensation	More frequent ACS calc.	Reform discounts
Option number	1	4	7	15	18	19	22
Axiom							
System safety	=	=	=	=	=	=	=
Consistency with law	=	=	=	=	=	=	=
Funding of Network Rail efficient costs	=	=	=	=	=	++	=
Allowance for market conditions	=	-	=	=	=	=	+
A single approach for the network as a whole	+	=	=	+	=	=	=
Objective							
Service costs recovery	=	=	=	-	=	++	=
Efficient whole-system whole-life industry net costs	=	=	+	=	=	=	-
Efficient long run investment decisions	=	=	=	=	+	+	=
Efficient performance management	=	=	=	=	+	=	+
Efficient use of network capacity	=	+	+	=	=	=	+
Judgement criteria							
Predictability	=		-	+	=	-	=
Simplicity	=	-	=	+	-	-	+
Transparency	+	+	+	+	=	=	=
Low transaction costs	-	-	-	-	-	-	=
Outcome							
Network Rail accountability	=	=	=	+	++	=	=
Non-arbitrary allocation of costs	++	=	=	-	=	++	-
Optimal traffic growth	=	+	=	+	=	=	=
Aligning industry incentives	+	+	+	=	+	=	+
Value for money for funders, taxpayers and users	=	=	+	=	+	+	-
Overall							
Option assessed in isolation	=	+	+	+	+	+	=

Table 5.3: CEPA detailed option assessments overall grading in each SoW

Option group	Network charges			Perfor	mance	Possessions	
Option	Avoidable cost	Scarcity charge (administered)	Reservation charge	Reset benchmarks more frequently	Recover end- user compensation	More frequent ACS calc.	Reform discounts
Option number	1	4	7	15	18	19	22
Current SoW	=	+	+	+	+	+	=
A more dynamic railway	++	++	+	+	++	++	=
On-rail competition via flexible franchising	+	++	+	+	++	+	=
More highly specified franchises	=	-	+	+	+	+	=
Freight protection / subsidy	=	+	=	+	+	+	=
Beneficiary pays for capability	+	+	+	+	+	+	=
Change in approach to capacity allocation	+	++	+	+	+	+	=
More regional decision making	=	=	+	+	+	+	=

ANNEX A SIMULATION MODELLING

This annex introduces the simulation modelling we have carried out of the seven detailed options, in order better to understand the impact on stakeholders and the effect they might have. The model comprises a central module, which provides common inputs and a uniform method of analysis for each simulation, and customised simulation of each option.

The simulation model is designed to provide a consistent framework for the analysis of each option. The objective of the model is to show the financial impact of charging options on different types of passenger and freight operators. There will of course be impacts on other stakeholders (notably Network Rail and funders) but these are dealt with separately in our qualitative analysis described in the assessment templates found in Annexes B to H of this report.

We have therefore segmented the market by modelling six different types of stylised operator – franchised commuter, franchised inter-city, franchised regional, open-access, multi-customer freight and bulk freight. It is important to note that these stylised operators are simplifications of reality²² and that any given company may, in reality, contain a mix of these six different types of service. Therefore it may be more intuitive to think of impacts as applying to financial flows from a particular type of service (e.g. within a company).

Broadly speaking, the model is separated into three distinct components – inputs, calculations, and outputs. The model flow is summarised in Figure A.1 and each component is described in more detail below.

Model inputs are subdivided into two types, and are described for each option in the later sections of this annex:

- **Global inputs**: these inputs set the baseline for operators and Network Rail. They include the baseline level of charges, revenues and operating costs as well as non-financial information such as train miles.
- **Option specific inputs:** these are additional inputs relevant to the individual options. For example, a key input for the reservation charge option is the utilisation rates of capacity allocated to different operators.

Each option then has its own unique set of calculations in constant 2012/13 prices that determines the change in charges or incentive payments for each type of operator. The changes in charges/incentive payments are then fed into the common (or global) calculations, ensuring the financial impacts are treated consistently for each option. There are two key assumptions sitting behind the common calculations:

²² For example, some but not all bulk freight services may compete with road.

- FTAC is used to make up the difference between Network Rail's revenue requirement and total charges, after the charges in each option have been calculated. It is distributed based on current shares of FTAC between operators.
- We assume no network grant is in place. This reflects the recent Budget announcement in relation to rail industry money flows. This results in a significant increase in the net revenue requirement to be recovered through the FTAC, increasing from £0.6bn to £4.4bn each year (2012/13 prices). Assuming that this would purely be a change in money flows, we assumed that operator profits would not be affected. Therefore, baseline revenues were adjusted for each franchised passenger operator so as to maintain their current level of profits.

Finally, changes to charges, share of total charges, profits and profit margins are calculated for each stylised operator. The final outputs of the model are interpreted as ceteris paribus effects (i.e. all other factors held constant). In practice, there may be some behavioural change resulting from the implementation of a given option (e.g. freight may release access rights under a reservation charge). The model is not intended to be a behavioural model and therefore does not capture these effects. Instead, these are discussed, where appropriate, in the respective assessment templates.

In addition to the core calculations of the model, each option has been subjected to sensitivity analysis that flexes the values of key parameters.

The model structure is described in Figure A.1 below.



Figure A.1: Simulation model flow

A.1. Fixed charge based on avoidable cost

We examined the potential change in fixed charges that could result from moving from the simple allocation of costs based on traffic metrics, to a model based on the attribution of avoidable costs. We drew on previous studies of geographical distribution of costs and freight avoidable costs to examine scenarios based on incorporating increasing amounts of such information into charges.

Scenarios:

- Lower Avoidable cost information used to set charges for franchised services. Freight only line charge and freight specific charge rolled into the charge.
- Mid As "lower" but open access charged in line with modelled ratio of fixed to variable charges for intercity services. Freight charges set at lower bound of freight avoidable costs identified for PR13.
- Upper As "mid" but using upper bound of freight avoidable costs identified from PR13.

From the investigation of these scenarios we found that moving to avoidable cost charging has the potential for significant shifts in charges between different types of operators or at least between the different types of train services they run. Based on the assumptions we have used, it is conceivable that there could be a redistribution from inter-city services (using specialised but well utilised assets) to regional services. There also likely exists potential to attribute significant costs to freight operators. This would remain a small portion of total industry costs but we envisage that mechanisms would need to be put in place to protect certain operators' financial viability. The indicative static modelling found that a substantial portion of these costs would need to be passed on to consumers or to Government given the magnitude of the increase in charges relative to their modelled profit margins.

Our modelling of replacing the FTAC with a fixed charge based on avoidable cost drew primarily on previous studies of differences in variable usage charges on different track types,²³ and on freight avoidable costs.²⁴

Fixed costs are the residual revenue requirement that Network Rail is allowed to recover from operators once other forecast revenue from charges and grants is taken into account. Currently, the FTAC recovers these costs from franchised passenger operators. Using industry financial data it was possible to identify indicative shares of FTAC paid by each of the different stylised operators considered. We assumed that Network Rail would no longer receive the Network Grant from Government, resulting in a significant increase in the net revenue requirement to be recovered through the FTAC or avoidable cost charge, increasing from £0.6bn to £4.4bn each year (2012/13 prices). Assuming that this would purely be a change in money flows we assumed that baseline franchised operator profits would not be affected.

²³ Halcrow (2008) "Independent Reporter A: Reporter Mandate – Variable Usage Costs Final Report" available on the ORR website <u>here</u>

²⁴ L.E.K. (2013) "Estimating Freight Avoidable Costs Final Report" available on the Network Rail website here

A set of scenarios were developed to examine how the allocation of fixed costs might look if the charge were no longer based simply on traffic metrics and applied only to franchised passenger operators. Table A.1 below sets out the modelled shares of fixed costs allocated to each stylised modelled sector in each case.

Assumption	Franchised Commuter	Franchised Regional	Franchised Intercity	Open Access	Multi- customer Freight	Bulk Freight
Counterfactual	27.4%	49.0%	23.6%	0.0%	0.0%	0.0%
Lower scenario	28.1%	54.6%	17.1%	0.0%	0.0%	0.2%
Mid scenario	27.4%	53.2%	16.7%	0.5%	0.7%	1.4%
Upper scenario	26.6%	51.6%	16.2%	0.4%	1.7%	3.5%

Table A.1: Stylised allocation of residual net revenue requirement modelled

Source: CEPA indicative modelling

The change in the stylised franchised commuter allocation seen in the "lower" scenario compared to the counterfactual was driven by the percentage difference between the Halcrow (2008) estimate of an all-curvature London & South East (L&SE) variable usage charge and their "system-wide" estimate.²⁵ For inter-city, the all-curvature "Primary" variable usage charge was used. The remaining balance is allocated to the stylised franchised regional operators. In the "mid" and "upper" scenarios, open access were charged in line with the modelled ratio of fixed to variable charges for franchised intercity services.

In the "lower" scenario, the freight-specific and freight-only line charges were brought into the avoidable cost charge, with no net financial impact on freight operators (i.e. the reduction in charges was offset). In the "mid" and "upper" scenarios however, they were set based on L.E.K. (2013) estimates of £130m to £311m (2011/12 prices) of freight avoidable costs net of costs associated with existing charges.²⁶ The L.E.K. estimates used here were based on 35-year average forecast freight volumes, with one third of costs being allocated to multi-customer and the rest to bulk freight.

Assumption	Franchised Commuter	Franchised Regional	Franchised Intercity	Open Access	Multi- customer Freight	Bulk Freight
Lower scenario	2%	10%	-21%	0%	0%	0%
Mid scenario	0%	8%	-22%	240%	113%	99%
Upper scenario	-2%	5%	-24%	233%	271%	252%

Table A.2: Percentage change in operator total charges (%	6 of counterfactual Operator total charges)
---	---

Source: CEPA indicative modelling

²⁵ Halcrow (2008) Figure 6.2.1 "Table of Route & Curvature charges (£/kgtkm)"

²⁶ L.E.K. (2013) p38

The outputs of the modelling exercise are discussed in the avoidable cost detailed assessment in Annex B but it is worth underlining here that the impacts considered in this analysis were "static" in as far as we did not consider how operators or users might react to the reforms as there is significant uncertainty about that and ways in which the impacts could be managed. We envisage that in practice the charges would be set based on the aggregate of the train services expected for a given operator, rather than simple splits by operator as used in this exercise. Therefore, the impacts discussed for this option in Annex B might over-estimate the differences that would be seen between operators (which would be an average of the different train services they run) but under-estimate the difference that could be seen between a particular operator's services running on different types of track at different times of day.

There are a number of ways in which avoidable cost information could be calculated and how it could be used in charges. Therefore, at this point it is not possible to estimate the impact on particular operators with any degree of accuracy. The purpose of this exercise is to tease out some of the effects that could potentially emerge if this option were pursued in further detail.

A.2. Administered scarcity charge

Scenarios:

- Lower Peak path charge £12,000, on 50,000 chargeable paths. Off-peak path charge £600 on 210,000 chargeable paths.
- Mid Peak path charge £14,000, on 70,000 chargeable paths. Off-peak path charge £700 on 290,000 chargeable paths.
- Upper Peak path charge £16,500, on 110,000 chargeable paths. Off-peak path charge £800 on 400,000 chargeable paths.

In all scenarios, intercity services take the greatest part of the charge. However, as this additional income is assumed to reduce FTAC, regional services see a substantial charge reduction in compensation. Open Access and Freight experience small charge increases in absolute terms, but these can be large in proportionate terms especially in the upper scenario.

The effect on commuter services is mixed. Our analysis suggests that this option would reduce overall charges for this group in the low and mid scenarios and increase overall charges in the high scenario.

The scarcity charge value is based on a 2005 ITS study, uplifted to present day value. Some of the assumptions relating to charge applicability were based on an SKM study of capacity usage. Some operational assumptions are CEPA's own assumptions.

Our modelling of a scarcity charge is based upon an estimate by ITS (2005).²⁷ The ITS estimates were based on investigations using its PRAISE model which simulates railway route revenues. They found that the commercial opportunity value of Britain's key intercity routes to London, in 2005, was approximately £20/km peak and £1/km off peak. They suggested that charges should be for a full path, not the section actually driven, and that a standard path length for these purposes would be typically 300km. We find this appropriate to our envisaging of a simple scarcity charge, since we intend a simple path charge on commercially valuable corridors only. This suggests a typical path charge of £6000 per path for peak, and £300 per path off for peak, in 2005 prices.

The first step in the model is to update these values to present day conditions. It seems likely that the value of a path will grow faster than the rate of growth of ticket receipts, since many costs are fixed and economic value is related to net income. We have taken the real growth in intercity ticket income over 10 years from 2004/5 to 2014/15, and applied, as scenarios, three elasticities to this growth, of 1.0, 1.5 and 2.0, to uprate the path value.²⁸

²⁷ Scoping study for scarcity charges, (2005, revised 2006), C Nash, D Johnson, Institute for Transport Studies, University of Leeds, and J Tyler, Passenger Transport Networks, York, report commissioned by ORR.

²⁸ Since these are applied to large changes, and sensitivities are likely linear rather than constant elasticities, these are used to represent simple elasticities, not arc elasticities. But these scenarios can also represent other uncertainties about the value of the path charge.

Thus we obtain, in constant 2012/13 prices, a range of peak path values from approximately $\pm 12,000$ to $\pm 16,500$, and off-peak values from approximately ± 600 to ± 800 .²⁹

The next step in the model is to assess the likely applicability of these charges to train operators, which in our modelling framework are represented by six stylised operators. The key assumptions are shown in the following table. We explain how these assumptions are used further below.

Assumption	Franchised Commuter	Franchised Regional	Franchised Intercity	Open Access	Multi- customer Freight	Bulk Freight
Typical Path length (km)	46	67	194	378	163	142
Proportion of peak paths %	18%	20%	16%	8%	2%	2%
Proportion of paths outside chargeable time %	22%	20%	22%	5%	40%	40%
% Chargeable – lower	1.0%	0.5%	40.0%	90.0%	2.0%	1.0%
% Chargeable – mid	2.0%	1.0%	50.0%	95.0%	5.0%	2.0%
% Chargeable – upper	4.0%	2.0%	60.0%	99.0%	10.0%	5.0%

Table A.3: Selection of Modelling Assumptions for Scarcity Charging (Administered)

Source: CEPA indicative modelling

Since the charge is a charge per path, we need first to estimate how many paths are used by each operator type. We have been able to estimate average path lengths based on ORR data for different types of operator. For intercity services, the average path length is lower than the assumed Open Access average path length because some train operators in the Intercity category, in practice, provide a number of services which are more 'regional' in character.

We have based our estimates, for those train services attracting the charge, on the East Coast corridor as a case study. SKM (2012)³⁰ has shown in this case that the key bottleneck, at the time of that study, is the Stevenage-Peterborough section which admits 10 trains per hour. SKM additionally provide data as to what category of train is occupying the paths at different times of day. Open access operators have suggested that there is commercial value in operating between the morning and afternoon peak and they are unable to secure additional paths, so we assume that paths are scarce at this inter-peak plateau level of demand and an off-peak scarcity charge will apply. We assume that the scarcity charge does not apply at times when demand for paths is below this plateau level, and presumably requests for capacity can be met.

We can, from the SKM data, together with suitable data and assumptions related to Saturday and Sunday demand, estimate peak and non-chargeable proportions of paths for

²⁹ The model uses non-rounded outputs of the elasticity calculation.

³⁰ Assessment of capacity allocation and utilisation on capacity constrained parts of the GB rail network, SKM, 2012, report commissioned by ORR.

the intercity and open access services. Similar data from SKM on other busy areas enables a similar estimate for commuter and regional services. Estimates for freight services are CEPA's own assumptions partly based on this data and partly on comments received from elsewhere, such as on the proportion of freight trains that travel at night and at weekends.

The final, and most difficult, assumption is to assess what proportion of train paths are chargeable, overall for each stylised operator. Due to data limitations in this area, the values assumed here are conventional consultant assumptions rather than data driven. Establishing these values would require separate studies to establish realistic numbers. The main factor here is the number of corridors that are determined to be subject to the scarcity charge. We can reasonably assume the ECML, WCML and Great Western intercity corridors would be chargeable but it is possible that others might also be.

Our analysis suggests that the scarcity charge, modelled here, raises substantial additional revenue. This additional revenue is assumed to be deducted from FTAC. FTAC is then reallocated among franchise operators according to the metrics currently used, thus giving a final impact. The modelling results are in the following table.

Net Track Access Charge impact by scenario	Franchised Commuter	Franchised Regional	Franchised Intercity	Open Access	Multi- customer Freight	Bulk Freight
£m Lower	-10	-37	70	7	0	0
£m Mid	-9	-58	99	8	1	1
£m Upper	1	-89	124	10	2	2
% Lower	-6%	-14%	31%	156%	3%	2%
% Mid	-6%	-22%	43%	196%	10%	5%
% Upper	1%	-33%	55%	236%	23%	14%

Table A.4: Impact of Scarcity charge by scenario for stylised operators (2012/13 prices)

Source: CEPA indicative modelling

The impact on total charges is largest, in percentage terms, for open access operators, since they use scarce paths almost exclusively (although they have a relatively low use of peak paths). Even in the lower scenario, the increase in charges would completely exhaust open access operator profits. This may initially seem curious since the charges are set at a level which ought to be profitable for the marginal train service, and thus ought to be affordable by trains which are actually running.³¹ One reason could be the types of paths that open access operators are granted access to, which is based on the "not primarily abstractive"

³¹ The charge also exhausts, several times over, the profits of franchised intercity operators, but this is because they have already exhausted their profits by bidding a premium to operate the franchise, and hence this observation is uninformative, i.e. franchise bids would reflect the expected higher charges.

test³² and may limit the ability for open access operators to compete in the most profitable markets.

The charges are material to the freight operators, at least in the mid and upper scenarios. In the upper scenario, it is sufficient to exhaust most of the profit of the operators. However these charges result from the use of a fairly small proportion of their paths.

³² In deciding whether to grant access to open access operators, ORR assess whether new services could generate their own revenue, not just take it away from the current operator(s). This assessment is called the 'not primarily abstractive' test.

A.3. Reservation charge

We examined the financial impact of imposing a reservation charge on six stylised operators. The reservation charge modelled was based on a per-km charge calculated as a proportion of the VUC. Desk-based research suggested that capacity utilisation was lowest among bulk freight operators (and to a lesser extent intermodal services), while passenger services generally had a high level of path utilisation. We estimated km of unused access rights based on current utilisation rates of timetabled schedules (as utilisation rates for access rights were unavailable). We conducted sensitivities around the central scenario of access right utilisation to help capture the degree of uncertainty in point estimates of access right utilisation. These scenarios, in order of increasing impact (i.e. the lower scenario is one which results in lower impacts) were:

Scenarios:

- Lower Bulk freight use 90% of rights, no overbooking for other segments. Low level of redistribution of rights.
- Mid Bulk freight uses 70% of rights, some over-booking for all. 50% of unused rights redistributed.
- Upper Bulk freight uses 60%, others use 90% access rights. A high proportion of unused rights are redistribution.

The modelling results were driven primarily by the assumed level of over-booking, and consequently the greatest impact was found to be on bulk freight operators. The absolute level of impact on bulk freight was calculated to be modest, both in terms of profits and the level of charges paid. However, even small increases in charges have the potential to have a material impact on freight operators (in particular intermodal) due to competition from other modes of transport (primarily road) and the resultant price elasticity of demand of rail freight services.

The reservation charge envisioned in this assessment was modelled as a per-km charge applicable to all operators based on non-utilisation of firm access rights. We have modelled the charge based on a proportion of the VUC. The unit rate is calculated by taking the 25% of the baseline VUC and dividing by the number of train km for each operator. A separate unit charge is defined for passenger and freight operators by taking the average unit rate calculated in the previous step (averaged across passenger operators and averaged across freight operators).

Unit rates are then multiplied by train-km for each operator type, giving us the quantum of reservation deposits that are eventually returned to operators.^{33,34} We then scale this by capacity utilisation rates to arrive at gross reservation charges (i.e. reservation charges that are paid plus deposits that are refunded).

³³ This is because the train-km measure only captures services that actually run.

³⁴ The measure of train km used is based on services run and therefore will include some spot bids and services with contingent rights. For the purposes of modelling we ignore these.

Our estimation of access right utilisation rates is based on utilisation rates of paths in the timetable. We recognise this will be inaccurate to an extent as the timetable includes spot bids and contingent rights (which would lead to overestimation of impact), and it does not capture any rights that are held but not exercised (which would lead to underestimation of impacts). However, at the time of modelling information on the utilisation rates based on scheduled services as a proxy.

We were informed by Network Rail that approximately 22% of freight paths go unused, which is much lower than in the past due to recent releases of access rights by the industry. ³⁵ A split of this value for intermodal/bulk freight was not available. We therefore assumed that intermodal freight continued to use a high proportion of their paths (95%), as was found in the 2007 Freight RUS. We assumed the rest of the unused paths were attributable to bulk freight (i.e. the remainder of the 22%). This gave us an overall bulk freight utilisation rate of 30% in the mid scenario. While desk based research did not give us a specific path utilisation rate for passenger services it did indicate that passenger services used a high proportion of booked paths. We therefore assumed a 95% utilisation rate across all passenger services in the mid scenario. The utilisation rates assumed in the model are summarised in the table below for the low, mid and upper scenarios.

	Franchised commuter	Franchised regional	Franchised inter-city	Open access	Multi- customer freight	Bulk freight
Lower	100%	100%	100%	100%	100%	90%
Mid	95%	95%	95%	95%	95%	70%
Upper	90%	90%	90%	90%	90%	60%

Table A.5: Access right utilisation rates assumed in modelling

Source: CEPA assumptions based on Network Rail and desk based research

From the gross reservation charges we then subtract the quantum of deposits returned to operators to arrive at reservation charges for each operator type. We then assume a level of redistribution of access rights. These redistributed rights receive a rebate of 75% on the reservation charge.

The rate of redistribution is an assumption, as there is no available data on the proportion of rights that operators would exchange with others if a reservation charge were in place. The alternative to including this assumption would be to remove the rebate mechanism from our modelling (equivalent to a 0% redistribution). The effect would be that financial impacts would be scaled upwards; it would not cause a redistribution of impacts as we have assumed a uniform redistribution rate. Given the estimated financial impacts are small in

³⁵ The 2007 Freight RUS also contained information on timetabled path utilisation rates, but we understand that much work has been done in recent years in reviewing the allocation of freight paths.

magnitude, this makes no difference to the overall conclusion. Our assumptions are presented in the table below.

	Franchised commuter	Franchised regional	Franchised inter-city	Open access	customer freight	Bulk freight
Lower	33%	33%	33%	33%	33%	33%
Mid	50%	50%	50%	50%	50%	50%
Upper	67%	67%	67%	67%	67%	67%

Table A.6: Redistribution of unused access rights assumed in modelling

Source: CEPA assumptions

A.4. Reset Schedule 8 performance benchmarks more frequently for changes in traffic volumes

This option involves removing the capacity charge from the 'charging' side of the regime and instead adjusting the Schedule 8 benchmark to account for the anticipated traffic-related increases in Schedule 8 payments.

We have looked at the knock-on effects of this option on other areas of the charging regime: if Network Rail's revenue requirement does not change, the removal of the capacity charge component of revenue would increase the FTAC:

- Freight and open access operators would benefit as they don't pay FTAC;
- Franchised regional passenger services would see a net increase in their charges as they would face a large FTAC increase; and
- Franchised commuter and inter-city passenger services would also see an increase in their FTAC but not enough to counter the removal of the capacity charge.

This option involves removing the capacity charge from the 'charging' side of the regime and instead adjusting the Schedule 8 benchmark to account for the anticipated traffic-related increases in Schedule 8 payments.

The modelling of this option used relatively few assumptions and used data based on forecasts of CP5 charges. One of the key assumptions was that implementing this option alone would not affect Network Rail's net revenue requirement. This is an important assumption as the revenue formerly collected through the capacity charge is then assumed to be reallocated to the FTAC.

We then used forecasts of the total quantum of capacity charge paid by operators over CP5, split by stylised operator. This was taken from the PR13 final determination. The table below shows the counterfactual split of capacity charge by stylised operator as well as relative shares of the FTAC.

Charge		Franchised commuter	Franchised regional	Franchised inter-city	Open access	Multi- customer freight	Bulk freight
		10 operators	9 operators	6 operators	2 operators	4 operators	4 operators
Capacity charge	Share per operator	4%	2%	6%	0.5%	0.1%	0.1%
	Share for operator type	44%	21%	33%	0.9%	0.3%	0.6%
FTAC	Share per operator	3%	5%	4%	-	-	-
	Share for operator type	27%	49%	24%	-	-	-

Table A.7: Indicative Operators' shares of the capacity charge and FTAC in CP5, Source: ORR Final Determination

We then simulated the impact of removing the capacity charge altogether and redistributing it to the FTAC. The figure below demonstrates the absolute change in charges that might be experienced by each operator, if the capacity charge revenue is distributed across the FTAC in the same proportions as the 'current' split of FTAC (in CP5) described in the table above.



Figure A.2: Total charges by indicative operator, current charging (plain) and with this option (striped)

A.5. Recover end-user compensation through Schedule 8

The modelling for this option focused on passenger compensation only, as there was no available data on compensation paid to freight customers. We developed three scenarios based on the percentage of delays caused by each party; Scenario 1 covers 85 percent of delays, Scenario 2 covers 95 percent, and Scenario 3 covers all delays.

- 1. Require Network Rail to reimburse passenger operators' end-user compensation for delays caused by them. Network Rail cause 59 percent of delays and passenger operators cause 26 percent of their own delays, therefore this scenario covers 85 percent of delays.
- 2. Require passenger operators and Network Rail to reimburse passenger operators for enduser compensation for delays caused by them. Passenger operators cause 10 percent of delays.
- 3. Require all train operators and Network Rail to reimburse all passenger operators' enduser compensation for delays caused by them. Freight operators cause 5 percent of delays.

In modelling this option we used two key data series: the passenger compensation paid out in the 2014-15 as reported by gov.uk, and the attribution of delay in the year to 22nd August 2015 as reported by Network Rail. We mapped the data across to the indicative operators using relative train miles from the CP5 Final Determination to provide estimates of missing data.

As there is no available end-user compensation data for freight, we were unable to model a potential compensation system for them. Our modelling consequently focuses on passenger compensation.

The split of responsibility for delays (above three minutes) to passenger services in the year to 22nd August 2015 is shown in the figure below. We used this information to define three scenarios for our analysis:

- Require Network Rail to reimburse passenger operators' end-user compensation for delays caused by them; this would cover 85 percent of all delays to passenger services as passenger operators cause 26 percent of their own delays while Network Rail cause 59 percent.
- Require passenger operators and Network Rail to reimburse passenger operators' end-user compensation for delays caused by them; this would cover 95 percent of all delays.
- 3. Require all train operators and Network Rail to reimburse all passenger operators' end-user compensation for delays caused by them; this would cover all delays.

Figure A.3: Split of responsibility for delays to passenger services (above three minutes) and modelling scenarios

Delayed self 26%	Network Rail 59%	Passenger operator 10%	Freig opera 5%	ght ator %
Scenario 1: Delays caused b	y self and Network Rail		_	
Scenario 2: Delays caused b	y self, Network Rail, and passenger operators			_
Scenario 3: Delays caused b	y self, Network Rail, passenger operators, and freight operators			

Source: Network Rail's performance statistics for the year to 22nd August 2015, available here.³⁶

We then mapped the delay attribution data across our indicative passenger operators using relative train miles from the CP5 Final Determination. This gave the following split:

Table A.8: Attribution of delay to passenger services

	Passenger operator delays					
Type of operator that caused the delay	Franchised commuter	Franchised regional	Franchised inter-city	Open access		
Network Rail caused	59%	56%	63%	65%		
Self-caused	31%	29%	20%	11%		
Passenger train caused	6%	9%	10%	21%		
Freight train caused	5%	6%	7%	4%		

Source: Network Rail performance statistics and CEPA analysis

Next we mapped the government's published Delay Repay and Conditions of Carriage data³⁷ across to our three indicative franchised passenger operators, and determined a pro-rata estimate of open access operators' end-user compensation according to their relative mileage; this compensation is outlined in the table below.

	Franchised commuter	Franchised regional	Franchised inter-city	Open access
Sector total	£8.9m	£3.7m	£12.7m	£0.2m
N° of operators	10	9	6	2
Per operator	£0.9m	£0.4m	£2.1m	£0.1m

Table A.9: Passenger compensation paid out by indicative operator (2014/15 prices)

Source: ORR CP5 Final Determination and CEPA analysis

We determined, therefore, that for our four indicative passenger operators, the complete attribution of passenger delay compensation is as in the figure below.

³⁶ Network Rail also publishes annual data through the National Rail Trends (NRT) Portal, but that data does not include a 'freight' attribution category.

³⁷ Department for Transport, via gov.uk (2015) "Compensation paid by Train Operating Companies: Passenger's Charter & Delay/Repay 2009-10 – 2014-15."



Figure A.4: Overview of allocation of passenger compensation

Source: ORR CP5 Final Determination, Network Rail performance statistics, and CEPA analysis

Under the passenger compensation mechanism considered for this option, each indicative passenger operator would have the costs (grey) and incomes (blue) represented in the figure below. The gap between costs and income is largely due to compensation that the operator has to pay due to delays it caused itself. A smaller portion is due to compensation paid to other operators.





Source: ORR CP5 Final Determination, Network Rail performance statistics, and CEPA modelling

A.6. More frequent Access Charge Supplement calculation

The modelling examined the magnitude of the change in Schedule 4 over/under recovery that would result from an annual adjustment to the ACS based on an updated volume of works planned for that particular year.

Scenarios considered include:

- Lower Assumes a low proportion of Schedule 4 under/over recovery is due to variation in volume of works (33% of under/over recovery);
- *Mid* Variation in Schedule 4 costs due to volume of works based on 2014/15 data (57% of under/over recovery³⁸);
- Upper Assumes a high proportion of Schedule 4 under/over recovery is due to variation in volume of works (75% of under/over recovery).

The modelled option leads to estimated savings in the ACS paid by all types of franchise passenger operators with the average annual ACS declining by between 5% and 12%. Relative to their current profits, the biggest savings accrue to intercity services. However in the current SoW, some or all of these savings may flow to funders under specific franchise protection arrangements.

Our modelling involves estimating the changes in the ACS (and thus changes in the costs faced by train operators) from adjusting the ACS annually based on the volume of engineering works planned to be carried out during that year. We calculated an average annual ACS for each indicative passenger operator using figures from ORR's CP5 Final Determination. This was the baseline level of ACS used.

The modelling required an assumption about the level of actual Schedule 4 costs and how these may differ from the initial baseline estimate of the ACS. We calculated actual passenger Schedule 4 cost using historical data on the variation in actual Schedule 4 costs relative to the ACS, as shown in the table below. We used data from the last six full financial years (i.e. from 2009/10 to 2014/15). Based on this data, we assumed actual passenger Schedule 4 costs would be 17% lower than the baseline level used to set the ACS (equivalent to the average variation in the last six years).

Table A.10: Variation in actual passenger Schedule 4 costs relative to ACS

Year	2009/10	2010/11	2011/12	2012/13	2013/14	2014/15	Average
Passenger Sch. 4 costs (% of ACS)	-20%	-23%	-48%	-18%	+14%	-7%	-17%

Source: ORR data portal (Schedule 4 and Schedule 8 Costs and Income)

Next we applied scenario assumptions about the share of Schedule 4 under/over recovery resulting from variations in the planned volume of works. Starting with CP5, Network Rail's

³⁸ This estimate is based on information from Network Rail's Regulatory Financial Statements for 2014/15 about Schedule 4 costs variation relative to CP5 baseline due to deferral of activity.

Regulatory Financial Statements include estimates of the financial outperformance due to deferral of activity. Based on this information, we calculated that 57% of the outperformance in Schedule 4 costs was due to deferral of activity in 2014/15. We applied this figure as the mid scenario assumption in our modelling. The other two scenarios involved our own assumptions designed to test the possibility that a lower or higher share of Schedule 4 costs variation is due to changes in the planned volume of works.

Depending on the scenario, the ACS was adjusted by a percentage of the difference between the initial ACS and the actual Schedule 4 costs. For example, in the mid scenario, the ACS was adjusted downwards by 57% of the difference between the initial baseline ACS and the actual Schedule 4 costs. The remaining over-recovery of Schedule 4 costs is assumed to be due to better possessions management by Network Rail.

Overall the modelling results show a downward adjustment in the total ACS of between £11m and £25m per year with the mid estimate around £19m. Relative to the profits calculated in the common calculations of the simulation model, the biggest financial impact is felt by franchised intercity passenger operators followed by commuter and regional passenger operators. As open access operators have not opted in to pay the ACS and the freight Schedule 4 regime does not involve an ACS, there is no financial impact on these types of operators.

It is important to note that our modelling considered changes in the ACS at a network wide level. This resulted in an equal percentage adjustment in the ACS for all operators. In practice however the ACS is calculated based on estimated activity volumes at a route level such that the ACS adjustment for different operators could vary depending on the variation in activity volumes on the routes on which they operate.

A.7. Reform Schedule 4 discounts for notice period of possession

The modelling undertaken for this option explored the impact of different potential notification discount structures on Schedule 4 compensation payments. The scenarios considered involved:

- Scenario 1: Keeping the current notification thresholds but reducing the discount factors applied to early notifications by 10-15%;
- Scenario 2: Removing discounts for early notification; and
- Scenario 3: Adjusting the notification thresholds as well as discount factors.

The modelling shows that reducing or removing the discounts would lead to (potentially significant) increases in total Schedule 4 compensation paid to passenger train operators (by almost 60% if discounts for early notification are removed). This would lead to a potentially large positive financial impact on train operators although some of these gains may be transferred to funders under profit sharing mechanisms depending on the provisions of existing franchise agreements.

Our modelling of this option involved estimating the impact of changing the notification discount structure with a view to incentivise Network Rail to book possessions when it is most efficient to do so rather than focus on booking possessions very early. The notification discount factors applicable for CP5 are shown in the table below. These factors depend on the timing of the notification as well as on the type of service groups categorised by their average late time multiplier.

Average late time multiplier	By 26 weeks before possessions	By 22 weeks before possessions	By Applicable Timetable
4.3 or higher	40%	63%	85%
3.4 to 4.2	45%	65%	85%
2.8 to 3.3	50%	68%	85%
2.7 or less	55%	70%	85%

Table A.11: Passenger Schedule 4 CP5 notification factors (% of MRE payable under Schedule 8)

Source: ORR, CP5 Final Determination

To determine the share of possessions booked within each notification bracket and the value of these possessions, we used data provided to us by Network Rail on the number of possessions and the amount of revenue compensation paid out to train operators in 2014/15 in each of the notification discount brackets shown above. We establish a baseline by calculating total Schedule 4 compensation received by each operator. We also calculated the share of Schedule 4 compensation received by service group category (based on the average late time multiplier) for each operator.³⁹ This is relevant because changing the

³⁹ Depending on the types of train services operated, some train operators received Schedule 4 compensation only for one category of service group (e.g. service group with average late time multiplier between 3.4 to 4.2) while other train operators received compensation for two or more service group categories.

relative discounts for the different service group categories will have a different impact on train operators depending on the types of services they operate.

The next step in the analysis was to calculate the impact of changes in the notification discount structure under different scenarios. We conducted the analysis assuming no dynamic changes in the number and distribution of possessions across the different notification timeframes. We thus captured solely the impact of changes in the compensation rates payable under the Schedule 4 regime. Potential behavioural effects in the way Network Rail books possessions are however discussed in the detailed assessment.

The first scenario considered reducing the discounts applied for early notification of possessions by 15% for the earliest notification timeframe and by 10% for the middle notification timeframe. The rationale for these assumptions is that reducing the relative prices for booking possessions in different timeframes would reduce the incentive for Network Rail to book possessions too early. The notification discount factors applied in this scenario are shown in the table below. The Schedule 4 compensation paid out was then scaled up proportionally according to the new discount factors (e.g. for the smallest notification factor, a change from 40% to 55% represents a 37.5% increase in compensation paid in that discount bracket). Based on these assumptions the total Schedule 4 revenue compensation paid out increases by around 26%.

Average late time multiplier	By 26 weeks before possessions	By 22 weeks before possessions	By Applicable Timetable ⁴⁰
4.3 or higher	55%	73%	85%
3.4 to 4.2	60%	75%	85%
2.8 to 3.3	65%	78%	85%
2.7 or less	70%	80%	85%

Table A.12: Scenario 1 notification factors assumptions (% of MRE payable under Schedule 8)

The second scenario involved removing all discounts for early possession notification. The current notification factor for the latest notification window (85% of MRE payable) was thus applied to all possessions. This would remove the financial incentive on Network Rail to book possessions early (although requirements related to timetabling processes would still apply). This assumption leads to an increase in the total Schedule 4 revenue compensation of around 59%.

The third scenario involved a change in the notification discount thresholds as well as the discount rates applied. The potential notification thresholds considered and the rationale behind these assumptions were:

• 22 weeks before possession – this gives Network Rail an extra four weeks to consider its possessions timetable compared to the status quo but still allows significant time

⁴⁰ By 10pm, the day before the timetable comes into effect

for revisions and consultations with train operators before the Informed Traveller timetable is ${\rm published}$;⁴¹

- 12 weeks before possession this allows possessions to be agreed before the Informed Traveller timetable is published and advance tickets are generally offered to passengers;⁴² and
- By Applicable Timetable.

These changes required some assumptions to be made about the way possessions would be distributed within the new notification windows. As the earliest proposed notification timeframe (22 weeks before possession) involves merging the first two existing notification timeframes, we assumed the total number of possessions booked in this bracket would be equal to the sum of possessions currently booked in the first and second notification timeframes. The remaining possessions (booked after 22 weeks but before the Applicable Timetable) have been split equally between the second and third proposed notification timeframes. In this scenario we have also applied the notification discount factors used in scenario 1.

For scenario 3 the total Schedule 4 revenue compensation increases by around 21%. The increase compared to the baseline is due to the higher notification factors (i.e. lower discounts) applied in this scenario. The increase however is lower than in scenario 1 because some possessions are booked in a higher discount bracket compared to the baseline (e.g. possessions currently booked in the middle bracket are captured in the first discount bracket in this scenario).

⁴¹ The Informed Traveller timetable is generally published 12 weeks before it comes into effect

⁴² For simplicity, the 12 weeks threshold has been used, however in practice the deadline would probably need to be a few weeks before that in order to allow sufficient time for possessions to be included in the train operators' timetabling systems.

ANNEX B FIXED CHARGE BASED ON AVOIDABLE COST DETAILED ASSESSMENT

Option 1: Fix	Option 1: Fixed charge based on avoidable cost										
Overall performance against the RDG Vision in each SoW											
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers				
=	++	+	=	=	+	+	=				

Avoidable cost charging is an option to set charges used to recover Network Rail's net revenue requirement in a way that is more reflective of underlying costs than the existing Fixed Track Access Charges (FTACs), and with greater clarity of purpose.

Industry participants supported this option being selected for further investigation given its potential to send more informative price signals to train operators and their funders in the context of anticipated changes to money flows, i.e. a significant reduction in the Network Grant. We estimate that changes in money flows could result in an increase in charges to recover the net revenue requirement, putting them at almost ten times their current level⁴³ and increasing the influence of the approach used to calculate them. It was also selected for detailed assessment to investigate freight sector concerns about the implications of using avoidable cost information in charges.

For this option, we envisage that the current FTACs would be replaced with a new charge based on long-run incremental cost (LRIC) principles. This would produce a set of tailored charges based on causal links between train services and infrastructure costs.⁴⁴ They would be highly disaggregated by geography and by the specific demands train services place on the network. We assume in this detailed assessment that this option could be implemented as a "mark-up" consistent with the terminology of Commission Implementing Regulation (EU) 2015/909. Further legal analysis is required in this area but given its potential role in protecting the financial viability of non-franchised operators, we note that this is potentially a crucial assumption in the analysis.

The current FTAC methodology, which allocates costs based on traffic metrics (e.g. train km), results in charges that recover costs in areas where traffic is greatest. As shown in the table below, an avoidable cost methodology has the potential to be much richer.

	FTAC	Avoidable cost charge
Increase with own traffic	Yes	Yes
Capture utilisation of assets	No	Yes
Capture costliness of assets used	No	Yes
Capture cost of user-specific demands	No	Yes

Table B.1: Comparison of FTAC and avoidable cost charging

Indicative modelling completed in support of this assessment uncovered potential for large shifts in charges between operators. In particular, it might be possible to see shifts between train services operating on busy lines where economies of scale are realised to those where they are not. We also saw potential to attribute avoidable costs to freight operators, which could be substantial, but found they would require significant protection from them, with estimated charge increases (before considering any potential "mark-up" protections afforded by EU legislation) easily outstripping current profits.

⁴³ £4.4bn per year (2012/13 prices).

⁴⁴ We envisage that this option could be implemented as a "mark-up," potentially affording "ability to bear" protections to more vulnerable operators but further legal analysis is required in this area.

Option 1: Fixed charge based on avoidable cost

Overall, we found that while there could be informational benefits from investigating avoidable costs, this should not be conflated with those benefits that might arise through charging. In the current SoW, we anticipate minimal benefit from putting avoidable cost information into charges, given the nature of current franchising arrangements and decision making.

It is not clear that there would be sufficient benefit from this option to outweigh the non-trivial burden of the supporting calculations, and to calibrate the level of the mark-up to ensure the viability of open access and freight operators. Despite the weak overall performance of this option against the RDG Vision in the current SoW, we did identify potential for this option to be beneficial in four alternative SoWs (indicated above), making this option far more attractive if the sector were to move in those directions.

Limiting the scope of the charge to franchised passenger operators, it is possible to envisage some elements being introduced at PR18. For wider application, we anticipate far more time being needed to get the supporting framework right, at which point we might be in quite a different SoW.

Key characteristics

Description of option

Under an avoidable cost approach, a causal link would be established, using long run incremental cost (LRIC) principles, to allocate "fixed" costs to operators.

In general, LRIC approaches capture the cost resulting from "non-marginal" changes (i.e. more than just a one-unit change) in output over a timeframe where network capacity could be modified to most efficiently meet the new level of demand. This might involve enhancing existing capacity or abandoning assets at the end of their economic lives.

An avoidable cost approach is a form of LRIC that examines reductions in demand and considers what expenditure could be avoided in the long run. The HS1 "OMRCA2" charge is currently set on this basis.⁴⁵ The freight only line charge also has features of an avoidable cost charge.⁴⁶

Under this option, we envisage that the ORR would set charges based on the avoidable costs of train services (both passenger and freight) by examining what elements and features of the current network could in the long run be avoided at lower levels of traffic, for example:

- in a "minimal traffic" scenario, such as only running one train per day; and/or
- when groups of trains services are removed entirely.⁴⁷

We recognise that a number of options exist for implementing avoidable cost charging. However, for the purpose of this assessment, we envisage that the charges would be set using avoidable cost information calculated in a manner similar to that explored in the ongoing Brockley Consulting study, allocating costs in Network Rail's revenue requirement to train services with:

- a high level of geographic disaggregation;
- costs attributed to train services at each location where they would in the long run be avoided if that type of service were not to run;
- traffic metrics would be used to split attributed costs between all users to which they are attributed;

⁴⁵ ORR (2014) "ORR's Approval of HS1 Ltd's Five Year Asset Management Statement" available on the ORR website here p92.

⁴⁶ For information on the freight only line charge see RDG (2014) "Charges and Incentives User Guide" available on the RDG website <u>here p14</u>.

⁴⁷ NR (2015) "Network Rail's review of the existing approach to cost attribution and cost allocation for the GB rail network" available on the ORR website <u>here</u> p51.
- non-attributable costs at each location would be allocated across all users at that location based on traffic metrics (ensuring full cost recovery at each location); and
- remaining central costs, would be allocated across all locations based on traffic metrics (ensuring full cost recovery overall across the network).⁴⁸

As the revenue requirement allows Network Rail to recover both historic and forecast expenditure during the control period, we expect that the charges would be based on the attribution of current costs but also the attribution of forecast capital expenditure due to occur during the control period.

We assume in this detailed assessment that this option could be implemented as a "mark-up" consistent with the terminology of Commission Implementing Regulation (EU) 2015/909. Further legal analysis is required in this area but given its potential role in protecting the financial viability of non-franchised operators, we note that this is potentially a crucial assumption in the analysis.

Description of counterfactual

The current regime allocates portions of "fixed" costs to franchised passenger services for recovery through the fixed track access charge (FTAC). The allocation process is relatively simple but is seen by some operators as being "arbitrary." Fixed costs are defined as Network Rail's residual revenue requirement after deducting income from other charges and sources. Given the DfT's stated intention to remove or significantly reduce the Network Grant, we anticipate that a larger proportion of fixed costs will be recovered through track access charges (instead of Network Grant) from the start of CP6.⁴⁹

Each eligible operator's share of fixed costs is determined based on its share of total traffic metrics. Fixed costs are not allocated to freight operators but certain costs beyond wear-and-tear are recovered from freight as mark-ups in special cases.⁵⁰

Relevant factors impacting the form and/or the effectiveness of the option

- Franchising (Factors Report Section 3.2) In general there is a pass-through of track access charges in franchise agreements, which weakens the price signals provided by the charges and incentive regime thus limiting its impact over the franchise period and reducing its effectiveness.
- Track access arrangements (Factors Report Section 3.3) The charges regime needs to reflect the different needs of the network's users. Cost allocation is also made more complex in a mixed use network and changes are likely to be resisted by those who lose under any change.
- Approaches to specifying future outputs (Factors Report Section 3.4) The current central
 planning approach to investment may generate conflicts with a charging structure based on
 market signals.
- Economic viability of freight/ open access operators (Factors Report Section 4.4) These operators have limited ability to pass cost increases on to their customers and non-discrimination rules limit special treatment. It is desirable that these operators are not excluded and so this creates restrictions on the overall level of charges, (however distributed) that can be borne by them.
- Data availability, measurement, and billing (Factors Report Section 4.7) This charging option

⁴⁸ For example, in addition to the OMRCA2 charge, HS1 has a separate charge to allocate common costs. ORR (2014) p92

⁴⁹ See "Written question – 7552" available on the Parliament website here.

⁵⁰ For example, the freight-only charge is applied to terminal lines with segments only used by freight and which would be closed if freight services ceased to operate. See NR (2013) "Freight Only Line Charge" available on the NR website <u>here</u> p1.

relies on data, measurement, models and billing systems to set charges, measure consumption and charge the operators. Changes need to be proportionate and feasible, and may be time consuming both to establish and to embed.

Implementation	
Information requirements	The information required to calculate avoidable cost charges might depend on the approach adopted but we expect it to require a wide range of information, including on:
	 existing network capacity and capability;
	 asset information;
	• traffic (demand);
	 unit costs of activities; and
	 usage/capability standards.
	The granularity of information required might also depend on the level of disaggregation of the charges.
	We anticipate that while there could be diminishing marginal returns from the accuracy of information (especially if it were to require major changes in industry processes) information would need to be captured at a relatively fine level of disaggregation in order to meaningfully attribute costs. Based on our understanding of the ongoing Brockley Consulting cost allocation study for Network Rail, we expect that the information might need to be broken down by "constant traffic section" or "strategic route section" but it is possible that these areas defined for operational purposes are not suboptimal from a charging point of view.
Drivers	Ultimately, the level of avoidable cost charges should be linked to the underlying causal relationships between operators' use of the network and costs incurred by Network Rail. Each variable noted above under "information requirements" would drive charges for train services.
	We understand that the ongoing Brockley Consulting study for Network Rail is examining the traffic characteristics that drive costs. A selection of the potential longlist drivers considered for examination include:
	 track categorisation;
	 speed and stopping patters;
	 physical train characteristics e.g. wheel hardness;
	 use of electrification assets;
	 terminus occupancy; and
	 peak / off-peak usage (under different definitions).
	We anticipate that under avoidable cost charging, operators should expect to be charged for investments made primarily for their benefit but reflecting the useful economic life of the asset in line with the calculation of Network Rail's revenue requirement. They should also expect their charges to reflect how well utilised the assets are.
Calculation	The key principle behind the avoidable cost approach is the identification of (and

calculation The key principle behind the avoidable cost approach is the identification of (and principles translation into charges of) Network Rail costs that would be avoided if an operator or group of similarly demanding operators were to partially or entirely reduce their

use of particular parts of the network.

Alternative approaches include determining what elements and features of the current network could in the long run be avoided at lower levels of traffic, for example:

- in a "minimal traffic" scenario, such as only running one train per day; or
- when groups of train services are removed entirely.

We understand that the "minimal traffic" approach is likely to be preferable given that it is not affected by corporate structure but even within that approach there are different levels of detail that could be used to allocate costs based on competing cost drivers. Furthermore, there are questions regarding how common costs, including those of the "minimal traffic scenario" or of head office costs are factored into charges. For the purpose of this detailed assessment, we assume that they are allocated based on traffic metrics. Alternatively, they could be recovered as an equiproportional mark-up on all charges or paid for directly by Government, removing them from charges.

There are also key methodological decisions that must be made such as for if there is any differentiation between peak and non-peak traffic. This is important as it is the peak that normally drives the maximum capacity of the network and the peak may only be meaningful for trains running in one direction in the morning/evening. Therefore, it may not be fair to attribute peak-specific costs to train services running at times/directions where network utilisation is lower. A challenge to doing this in practice is to estimate the attributes and costs of a hypothetical off-peak only network in a consistent and mechanistic manner. However, this process could be informed by comparison with other parts of the network with similar characteristics but lower levels of demand.

There are also potential issues arising from the mixed-use nature of the railway. For example, freight operators use passing loops or take diversionary routes to allow the passage of faster passenger trains. In the absence of such passenger trains, some of that infrastructure could be avoided. Therefore, while the infrastructure is used by freight, it is caused by the usage of passenger services. The cost of such infrastructure therefore may best sit with passenger services not freight, and the charge might need to capture this level of complexity in the system.

These issues are not exclusive to freight. For example, if alternative, less-well utilised routes exist (e.g. the Hertford loop) current patterns of usage that reduce congestion elsewhere might be penalised. The effect might worsen as services move away from the less well utilised route as the cost would need to be recovered from a decreasing user base. Such issues might be partly resolved through a complementary scarcity charge or if non-attributable costs were covered by a direct grant from Government. Nonetheless, these effects underline the complexity of charging on a non-linear multi-use network.

We envisage that once calculated, avoidable cost charges would be implemented in the same manner as the current FTAC. They would be set based on forward-looking demand at each periodic review and consistent with the calculation of the overall revenue requirement, they would reflect the useful economic life of the asset, smoothing the cost over time and using the allowed cost of capital to protect its net present value.

Practical considerations	We envisage this charge being implemented as a replacement of the existing FTAC (and potentially the freight-only line charge and freight specific charge, which are both contributions to the current FTAC). Assuming the data collection issues, which are discussed in more detail below, can be resolved we do not envisage major practical implementation issues from a pure charging perspective. As indicated above, greater challenges exist in calculating the level of charges (see sections above) where imperfect data and use of hypothetical decrements in traffic require a significant degree of judgement/discretion from the ORR and Network Rail. These challenges are greater in rail than in other regulated sectors where avoidable cost charges are used such as in telecoms. In simple terms, in other sectors, network usage is a more homogenous continuous flow (of electrons, molecules or data packets), such that assets can be treated in a less specific manner. Network usage in rail by contrast is not homogenous, it is intermittent and costs are highly specific to different geographies through which trains pass. The abovementioned Brockley Consulting study is examining what data is currently available. Usage of existing sources might reduce the cost of introducing this charging option but it is possible that new databanks and processes might need to be put in place, and new charging models developed. These could take time to establish and achieve comprehensive coverage. There may also need to be greater involvement from operators and stakeholders in this process in terms of developing forecast demand so the element of the revenue requirement relating to the capital expenditure due to take place in the forthcoming control period could be allocated to train services. The 2006 GNER v ORR case found that one important reason that the existing FTACs were acceptable was that open access operators did not have equal access to the network such that they were constrained in the downstream markets they could access. This could mean tha
	included in our analysis, but this could be a significant change in the nature of the current centrally controlled model if required.
Lead time	As noted in Brockley Consulting (Mar 2015) there are some ways in which the current FTAC could be brought closer into line with an avoidable cost approach. Therefore, there may be potential for some elements of the approach, primarily those affecting the franchised rail operators, to be achieved in time for the next periodic review, particularly as they might (at least in the current SoW) be broadly held harmless to the charges. If the objective is to bring all operators' train services into the charge, a longer time period might be required to ensure:
	 consistent data sets and processes across train services and regions;
	 appropriate changes are made to the periodic review process to bring in attribution of expenditure (i.e. there may be a greater role for operators to agree to expenditure if they are going be charged for it);
	 relevant forecasts, particularly of enhancements, can be developed and attributed; and
	 funders can establish processes to neutralise or adapt funding flows in a way that does not preclude entry of new operators or materially adversely affect existing operators.

Given the above considerations, we anticipate that PR23 might be the earliest point at which full avoidable cost charging could be put in place. However, given the potential implementation complexities and the potential informational benefits of avoidable cost studies, there might be value in ensuring that all information is in place for a "dry run" at PR23, leaving implementation in charges as an option for PR28.

Resources As reinforced in Brockley (2015), while similar approaches have been put into place in other industries "the complexity and scale of effort required to estimate LRIC in a network as complex as rail should not be under-estimated, particularly if the analysis is conducted at a highly disaggregated level."

We expect that marginal changes to the current FTAC inspired by the avoidable cost approach could be implemented by PR18 but that minor revisions might take place in any case as part of the normal process of the periodic review.

Moving beyond marginal changes to existing arrangements, we expect that significant resources for the ORR, Network Rail, funders and operators might be required. Regulatory, technical, legal and policy workstreams would be needed to develop the links between access policy, funding/investment policy framework and the structure of charges. We expect this to require extensive consultation, development of charging and attribution models, integration of the process with the periodic review, and the development of processes to ensure that the charging and evidence base is refreshed regularly on a consistent basis.

The resources required to do this could be minimised if part of a wider package of reforms but given the "zero-sum" nature of the charging process, such that any change creates both winners and losers, the starting methodology needs to be developed with care. Subsequent changes would need to be robust to challenge.

Performance	against crit	eria						
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
System safety	=	=	=	=	=	=	=	=
	There is no approach t this area.	o material ir to the fixed	npact of t charge, th	his charging his option she	approach ould have	on system s limited ince	afety. As an entive prope	า erties in

Consistency with law	=	=	=	=	=	=	=	=
with law	Article 31.3 package an	of Directi	ve 2012/34 ss to infrasi	1 requires t	hat "[T]he onnecting se	charges for rvice facilit	the minim	um access
	cost that is	directly in	curred as a	result of o	perating the	e train serv	ice." ⁵¹	
	Commissio	n Impleme	enting Regu	lation (EU) 2015/909	of 12 June	2015 appe	ears to set

⁵¹ Directive 2012/34/EU of the European Parliament and of the Council of 21 November 2012 establishing a single European railway area Text with EEA relevance available on the EUR-Lex website <u>here</u>.

out a minimal notion of "cost that is directly incurred" using a very short-run approach⁵² but the allocation of fixed charges using LRIC appears to be permitted as a basis for setting mark-ups by use of the wording of Article 31.3: "In order to obtain full recovery of the costs incurred by the infrastructure manager a Member State may, if the market can bear this, levy mark-ups on the basis of efficient, transparent and non-discriminatory principles, while guaranteeing optimal competitiveness of rail market segments."

We note that the GNER v ORR (2006) case established that from a legal perspective, the FTAC was a mark-up but that it being levied on franchised passenger operators only was not discriminatory as:

- the charges were a "wholly artificial construct" to recover residual fixed costs rather than being attributed; and
- open access operators did not have the same protection from changes in charges as franchised passenger operators.

This raises important questions about the extent to which an Avoidable Cost mark-up could be levied on operators other than franchised passenger operators without extending protections to them.

One freight operator raised a concern that use of avoidable cost techniques to attribute costs might result in them being classed as costs directly incurred. Such a classification might result in charges without ability to bear protections. In market segments with limited ability to bear charges such as freight, full or partial direct subsidy might be required to cover the charges. There could be concerns regarding how that subsidy level would be calculated and also if there would be any state aid issues. One open access operator expressed the view that the type of direct subsidy required in such a case would be unlikely to be forthcoming.

Additionally, there is a question about whether if information on attribution of costs were simply to exist, the ORR might be obliged to introduce such "direct" costs into charges. The discussion of whether the charges reflected the best information available in the GNER v ORR (2006) suggested that it might be difficult to defend a charging approach that was not based on the best available information.⁵³

Whether or not this would be automatic, such a situation might arise if an operator were to identify that they might win from the change in methodology, and were to push for such a change through legal avenues.

Clearly, further legal analysis is required in this area to ensure the above issues are addressed before any changes to the charges structure were made.

⁵² Commission Implementing Regulation (EU) 2015/909 of 12 June 2015 on the modalities for the calculation of the cost that is directly incurred as a result of operating the train service, available on the EUR-Lex website <u>here</u>.

⁵³ The transcript of this case is available on the British and Irish Legal Information Institute website <u>here</u>.

Option 1: Fixe	ed charge based on avoidable cost										
Funding of											
Option 1: Fix Funding of Network Rail's efficient costs Allowance for market conditions	This approach relates to how fixed costs are allocated to (and thus paid for by) different operators' train services. As a result, it is not expected to affect the overall level of funding of Network Rail. It is assumed that any costs that cannot be attributed to particular train services would be allocated using traffic metrics, achieving full cost recovery. A further assumption is that any subsequent decisions for operators to exit the market or reduce particular train services would not have an impact on Network Rail's funding, at least in the long term.										
Allowance for											
conditions	Commission Implementing Regulation (EU) 2015/909 of 12 June 2015 makes the following provision that "the level of charges must not [] exclude the use of infrastructure by market segments which can pay at least the cost that is directly incurred as a result of operating the railway service, plus a rate of return which the market can bear." This implies that operators or perhaps even funders of public service obligation contracts (particularly in the "Regional powers" SoW) who could not afford to pay these charges might have to be protected from them. These protections might apply to at least some freight and open access operators but it is not clear to what extent the protection could extend to small increases in charges										
	it is not clear to what extent the protection could extend to small increases in charges as there is not a clearly defined procedure or robust evidence base in place.										
	It is important to note that if the avoidable cost approach were determined to be a cost directly incurred, rather than a mark-up, the abovementioned protections might not be binding and the assessment of this option against this criterion would be more negative. For example, this charge is similar in many ways both to the FTAC and freight-specific charges, which are both mark-ups. However, the use of processes to attribute costs using cost drivers suggests that it is more of a "direct cost" than the Capacity Charge, which falls into that category. Freight operators indicated that if this were the case, it might only be possible to sustain a viable sector under a "protect freight" SoW, in which there might be some difficult questions to answer regarding its legality under state aid legislation, and it might be necessary to consider other options to offset the impact to provide compensation for wider economic benefits not considered in the avoidable cost charge.										
A single	+ + + + + + +										
approach for the network as a whole	While the detail of the implementation (for example, in considering the nature of the relevant decrements in demand) might vary by location and funding arrangements (for example, in the SoW "protect freight," it might be possible to recover greater costs from freight operators through mark-ups) we envisage that the avoidable cost methodology would be applied across the network. As the current allocation is only applied to franchised passenger operators' train services, this criterion has been marked positively. However, we note in the "lower" scenario used to examine this option, there are ways in which the charges might not flow through to freight an open access. If that were the case, the option would only be graded as amber.										

Option 1: Fixe	Option 1: Fixed charge based on avoidable cost										
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers			
Service costs	=	=	=	=	=	=	=	=			
recovery	We do not anticipate that this option would have any impact on Network Rail's abilit to recover the total efficient costs of providing and improving all services. However, might provide a mechanism through which additional capacity could be funded be passengers and freight users if they are willing to pay.										
Efficient	=	++	+	=	=	=	=	=			
whole-system whole-life industry net costs	This option operators change mig passenger changing tl Avoidable commercia powerfully potential t certain cos were to fa during the their servi recovered one percer can be ach of annual delaying er workshop, spending b Despite th service pro signals cou a pass-thro be indifferent to franchis might be er that assum provision.	n might res to freight (ght lead to a versus freig ne mix of se cost chargi ally viable e given the to make op ts could be ce the cost morning pe tice pattern from their nt reduction ieved. At it savings on nhancemen the ORR ir y one year of e large pot by sion are of ld influence ough of acce ent to the le se bidding of even weake the grading in franchis	sult in a r and poter a change i ht, the pa rvices pro ng has th nhanceme advanced erators d avoided c of assets eak that d to avoi services c in renew s July 201 this basi ts if it con ndicated t could delive ential sav centrally p ess charge evel of cha processes r. As a re that might is most p e protect	ebalancing ntially to op in the mix o ttern of serviced by ea- ne potential ents could by d state of of ecisions mo- or postponed that only t loes not occo d triggering or later whe rals expendi 5 charging v s of £121m uld mean th hat efficien ver a presen ings that co olanned, it is rovision uno s during fra arges) such . In the "sp sult, this cr nt enable th ositive for t	of fixed c en access f services provi ch operato to provic be funded developme re reflecti d in the lor hey use (e cur for oth g addition n the asse ture has th workshop, n. Furthern hey would tly delayir t value cos buld be ur s not clea der the cur nchise con that any in becified fra iterion is r e rebalanc he "dynam	osts from f operators provided, al ded within or. le a mecha by operato ent of the ve of costs ng run. For e.g. to expa er operato al expendi ets need to ne potentia the ORR pr more, oper be better of ten perces t saving of allocked, as r to what e rent regime tracts (i.e. f npact of pri anchises" S marked mo ing of char ic rail" SoW	franchised . In princip tering the l each sector unism throu rs but perh network, i is in such a example, if ind networ rs), they mi ture that be renewe to be signi- resented ar ators migh utilised. At ent of enh £1.2bn. the key ele- extent avoid e. In addition train operation cing signals oW, the pri- st positively ges to mod / given the arges and	passenger le, such a balance of and even ugh which haps more t has the way that operators k capacity ght adapt could be ed. Even a ificant if it n estimate t support the same ancement of dable cost n, there is tors might is limited rice signal y in SoWs ify service combined increased			
	link from c	harges to ou	utcomes is	s less certair	1.						
Efficient long	=	=	=	=	=	+	=	+			
investment decisions	As discusse reflective t term inves regulatory	ed above, a han the sta tment deci regime that	n avoidal tus quo. T sions but t is fully al	ble cost allo his has the p providing e igned with t	ocation of potential to effective in hat object	fixed costs o encourage nvestment ive.	might be e more effic incentives i	more cost cient long- requires a			
	This optior features t	n has been r hat might	narked as reduce t	neutral for he effectiv:	the currer eness of	nt SoW as the invest	here are a r ment ince	number of ntives. In			

particular, the central planning nature of the investment decision making process. If decisions are taken centrally and are based on a wide range of variables, price signals may not make a difference or may only have a limited impact.

The SoWs "beneficiary pays" and "regional powers" have been marked positively as they might reduce the central planning features of the regime and thus more likely to enable effective investment incentives.

One freight operator suggested that the greatest benefit from the avoidable cost approach is the information that it provides. This could inform better decision making even in a centrally planned system. However, they argued that going the next step and using the information to set charges would not be a good idea.

Efficient	=		=		=	=			=	=	=		=
management	There manage	is no emen	ot a it.	clear	direct	impact	of	this	chargin	g appro	ach on	perfo	ormance
Efficient use	=		++		+	=			=	+	+		=
capacity	As disc This ha effectiv with th	As discussed above, this option should be more cost reflective than the status quo. This has the potential to encourage efficient use of network capacity but providing effective use of capacity incentives requires a regulatory regime that is fully aligned with that objective.											
	feature criterio contrac of price	etion es that in "E ctual i e sign	has t at re Efficie natur als.	been duce Int lo re of t	marked the eff ong rur he capa	as neut ectivene invest city alloo	ral f ess o mer catio	for the soft of th	e currence e of ca ecisions, ocess ma	nt SoW a pacity ir " the c ay limit o	as it has ncentive entral r elimina	s a nu s. As planni ate the	mber of for the ing and e impact
	The So' allocati plannin allocati the cor increas	Ws "(ion" ig an ion in mbine ied fra	dynar have id co icenti ed re anchi	nic ra beer ntrac ves. ductio sed fl	iilway," marke tual fea The grae on in fra exibility	"on-rail ed posit atures, a ding is n anchise p	com ively and nost prote	npetit /, as thus posi ectior	tion," "b these are mo tive for n from o	eneficiar would re ore likely the "dyr changes i	y pays" educe c y to en namic ra in acces	and " ertain able il" So' s char	capacity central capacity W given ges and
Judgement criteria	Curren	nt	Dynar railwa	nic ay	On-rail comp	Speci franch	fied lises	Pr fr	otect E eight	Beneficiary pays	Capaci allocat	ty l on	Regional powers
Predictability	=		=		=	=			=	=	=		=
	It is not less pr "arbitra of use result i of oper	t clea edicta ary" t of ca n cha rator.	r if o able han t pacit irges	nce ir over hey c y or c that	nplement time the urrently operatory would b	nted, an nan the vare, as rs' ability e more	avo curi they y to or le	idable rent / sho bear ess pr	e cost ap approac uld be d charges redictabl	oproach y h. Move riven by s but it is e for all	would b ments changes s not cle or any p	e any might in the ear if i particu	more or be less e nature it would ılar type
Simplicity	=		=		=	=			=	=	=		=
	We en current	visage t FTA(e tha C.	t avo	idable d	ost chai	rges	wou	ld applie	ed in a r	nanner	simila	r to the
Transparency	+		+		+	+			+	+	+		+
	This op links be	tion s	shoul en use	d ma e of th	ke the one two	harging: ork and t	regi :he r	me n ecov	nore tra ery of co	nsparent osts.	by basi	ng it i	n causal

Low transaction costs

We expect that avoidable cost charges would be more complicated to determine than the current FTAC, creating additional set-up and ongoing transaction costs. It is possible that some of the information required for this approach is not currently available and it may not be straightforward to determine how charges might respond to a given change in usage.

While significant set-up costs are anticipated (in the range of a small number of millions of pounds (see discussion on stakeholder impacts for Network Rail)), the level of ongoing transaction costs is not clear. However, once up and running there could be significant disagreement between parties about which methodology to use, updates to the methodology over time or in areas where judgement is required to determine the appropriate allocation of costs. These points will be important for operators who feel the impact of charges. In the current SoW, that may primarily be freight and open access operators. However, in SoWs such as "Dynamic railway" and "Beneficiary pays," there might be greater scrutiny from franchised passenger operators.

One passenger operator anticipated that such discussions could present an opportunity for operators to "game" the regime. However, that may already exist to some extent under the current regime, where there may be opportunities for operators to contest aspects of the more simplistic approach currently in place to determine the allocation of fixed costs.

Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
Network Rail	=	+	=	=	=	+	=	=		
accountability	There is no regimes in implement which Net result, the decisions "benefician	ot a clear d which inv ting avoidal work Rail r SoWs that are marke ry pays."	irect imp vestment ble cost c needs to i could pot d positiv	act of this cl decisions a harging coul dentify the entially invo rely. These	harging ap re made d improve benefician lve Netwo are the	oproach on by the go e accountal ries of capit ork Rail mak SoWs "dy	accountab vernment. pility in a s cal expendi sing more in namic rail	ility under However, cenario in ture. As a nvestment way" and		
Non-arbitrary	++	++	++	++	++	++	++	++		
allocation of costs	The avoida fixed costs costs migh	able costs a to operato t retain a se	pproach s ors' train sparate sir	should repre services but mplistic alloc	sent an ii it is ackr ation met	mprovemen nowledged 1 hodology.	t in the all that certair	ocation of 1 common		
	In the "protect freight" SoW, any greater burden of costs might be reversed for freight but the changes could still flow through to other operators. Therefore, as for the other SoWs, the "protect freight" SoW is still marked positively.									
Optimal traffic	=	+	+	=	=	+	+	=		
growth	In principle, in the SoWs allowing greater responsiveness of operators to charges, the greater cost-reflectivity of avoidable costs might support efficient long-run investment decisions and efficient use of network capacity. One freight operator noted however, that it is important to understand that this greater cost-reflectivity might not reflect broader net benefits to society. There could remain a role for public funding to ensure traffic growth reflects societal considerations. As we envisage it, avoidable cost charging would only capture costs to Network Rail.									

Option 1: Fix	ed charge ba	ased on avo	oidable cos	t				
Aligning	+	+	+	+	+	=	+	+
industry incentives	This chargi avoidable of charged for services. The network ple without avo Avoidable of from scarce when othe such routine might be of route.	ng approad cost chargir r capital exp his might in anning. Thi dable cost cost alone c ity. Avoida rs are avail ng options outweighed	ch should I ng would cr penditure i crease the s effect mi t charging. cannot how ble cost cl able. It is p exist. How I by signals	ead to bet reate a clea ncurred by role for op ght be achi vever captur harging dis possible tha ever, the r s for peak/	ter aligned rer causal Network R erators and ieved in the re all impac courages u t this could magnitude doff-peak u	l incentives link where ail for the k d their fund e "Beneficia cts; particula ise of less d worsen so of this effe isage withir	. We en operator penefit of ers to ta ary pays" arly thos well-utili carcity iss ect is not n day or	visage that s would be f their train ke a role in SoW even e that arise ised routes sues where t clear and t the same
Value for	=	+	+	=	=	+	+	=
funders, taxpayers and users	For the reasignificant could have where they behaviour. Based on C terms of ef the tens or these are "Dynamic H reforms in that these H The costs millions of maintain th option outs The ORR understance impact for beneficial e expect that However, e might be re- operators' there migh demonstra	sons discus impact on potential t re is greate DRR analysi ficiently av r even hund unlikely to Rail" or "O the "Benefits con of impleme pounds giv he regime. side of a So has argued ling of expe Network even without some aspe establishing equired to p were expect viability be t remain a te value for	ssed above, value for r o deliver ir er ability f s, the bene oiding or d dreds of m be realise on-rail com iciary pays" uld be unlo enting avo ven the tim Therefore W that cou d that sin enditure co Rail) to th ut the end ects of this the inform out the opti the opti the opti the opti the opti the opti	, we do not noney in the nproved va or pricing se efits that av- lelaying exp illions of po- d without p" SoWs of or "Capaci cked via cha idable cost ne and tech , it might f ld facilitate nply the in- uld be very ne extent f product of f option mig nation is just ion into pra ticipate, wi ke. Therefo hurdle tha funders, tax	expect aver ne current lue for more signals to voidable co- benditure h bunds a ye the additic r the great ty allocation arging in th c charging inical resound the benefit information r significant that avoid feeding intent that avoid feeding intent that avoid feeding intent that significant the pursu to ne piece ctice, and p th significant re, even if t this optic gravers and	oidable cost SoW but w ney in certa affect oper ost could ur ave the pot ar. Howeve onal franchi ter decentr on" SoWs. W e Current Si could quite urces requir alue for mo ts. al benefits cold quite urces requir alue for mo ts. al benefits could quite urces requir alue for mo ts. al benefits could quite urces requir alue for mo ts. al benefits could quite urces requir alue for mo ts.	t chargin ve do exp in altern ators' ar alock, par tential to tential to er, we co se flexib ve do nor oW. e possible do nor oW. e possible red to es oney to p soin of s modelling therefore the cour ele. Signif so if non le risk sh ation we eed to ov	g to have a bect that it ative SoWs id funders' rticularly in reach into nsider that ility of the and other t anticipate y run into tablish and pursue this ring better stakeholder g could be e, we might nterfactual. icant effort -franchised nould those re in place, vercome to

Option 1: Fixed charge based on avoidable cost								
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
	=	++	+	=	=	+	+	=

The approach used to allocate fixed costs affects the price signals faced by funders and operators. Given the anticipated changes in money flows, the approach adopted will have greater importance than it has had to date. Moving to an approach based on avoidable costs presents a clear alternative to move beyond the current simplistic arrangements, with the potential to send more informative price signals for the use of scarce resources.

At this stage, it is difficult to separate out the benefit of being able to take advantage of the greater knowledge that Network Rail, operators and funders might have about the network through conducting avoidable cost analysis and the benefit that might arise from using the avoidable cost information to set charges. There is nearly complete consensus in the industry regarding the informational benefits of this approach⁵⁴ but less so on going the next step and using that information to set charges.

In the current SoW, we anticipate that the benefits of this approach might be predominantly informational, with minimal incremental benefits from putting that information into charges given the limited incentive for most operators to change their use of the network. These may or may not outweigh the significant burden of the calculations and the expectation that certain open access or freight operators might not be able to bear the additional charges. The potential for this more sophisticated approach to provide its greatest impacts lies in alternative SoWs where price signals are stronger. The grading for avoidable cost charging is most positive for the "dynamic rail" SoW given the combined reduction in franchise protection from changes in access charges and increased franchised flexibility.

Impact on stakeholders

As discussed in the "Performance against criteria" section above, the long-term prize offered by the avoidable cost approach in certain SoWs is a network that better meets users' needs, with investment and operational decisions being justified based on causal relationships transposed into charges. In the short-run however, the avoidable cost approach has its greatest impact on the distribution of charges. This means that there will be winners and losers but the extent to which that is felt will depend on how charges are tempered by ability to bear tests or grants.

In support of this detailed analysis, we have conducted some indicative quantitative analysis using existing industry information to understand the potential distributive changes under different scenarios. This analysis conducted is predominantly static as the way in which operators might adapt their behaviour is subject to even greater uncertainty than the level of charges.

For the purpose of this exercise, drawing on work completed prior to PR13 regarding potential geographical disaggregation of variable usage charges, we assume that the presence of increasing returns to scale will result in lower costs being attributed for more highly utilised parts of the network. We also draw on work completed prior to PR13 on freight avoidable costs and assume that train services requiring particular adaptations (e.g. high-speed inter-city services) could be attributed those costs. As with the other detailed assessments, we consider the implications of three scenarios:

⁵⁴ One freight operator highlighted the risk that attributing costs and having that information might obligate the ORR to charge for them, something on which there is far less consensus.

- Lower avoidable cost information is only implemented in charges for franchised services,
- **Mid** as "lower" but all operators face charges open access in line with the ratio of fixed to variable charges modelled for intercity services and freight in line with lower bound of freight avoidable costs identified for PR13.
- Upper as above but using upper bound of freight avoidable costs identified from PR13.⁵⁵

The static impact that the change in approach might have on operators is analysed for the set of six stylised operators using a set of charges calculated using average CP5 charges and industry financial data. Further information on this approach is provided in Annex A to the main report. However, the primary objective of this exercise was to gauge for which types of train services the change in approach might have an impact, whether it could be large, and how it might shift the distribution of charges in the sector. We also compared the changes to profit margins to examine whether the shift might affect the viability of certain operators and require additional measures to support the viability of those market segments. In this analysis, we have considered the portion of increases in charges that might need to be passed on to users or funders to maintain half the modelled counterfactual profit margin for the stylised operator. However, that is certainly not to say that such a reduction in profits could either be desirable or sustainable. In the long run, franchised operators can protect their profit margins through the bidding process (it is inconceivable that an operator would ex ante bid a negative profit margin) but operators may simply chose to drop unprofitable services or withdraw completely from the market.

Franchised The impact of avoidable cost charging on any particular operator's payments to Network Rail would be a function of the methodology chosen and even then would require detailed technical analysis to gauge the impacts for the train services it runs. Any analysis at this stage is highly speculative but previous work on differences in renewals costs by line type,⁵⁶ while based on marginal rather average costs, may provide a way in to understand some of the key effects. That study indicated that for the "London & South East" (which we assume to align closely with this representative operator group) costs were only three percentage points higher than the network average. Therefore, it is conceivable that on average, franchised commuter passenger operators might not see a major change in charges relating to fixed costs under the avoidable cost approach.

Under the "lower" scenario, we estimate that this type of operator could maintain half their current profit margin if it is possible to pass on just over half the increase in charges. However, it is possible that in the "mid" and "upper" scenarios, if some charges were displaced to freight, their total charges might even slightly fall.⁵⁷

Overall, we anticipate that there could be a relatively small increase in charges for this group (assuming there is no direct grant to Network Rail), although that might mask significant differences within each operators' services. Whether or not that would feed through to their bottom line or change their actions, this could depend on the SoW in question. However, franchise contracts are likely to give protection to franchisees for such a large shift in approach within the franchise contract period. At the subsequent retendering process, the updated charges could be reflected in the

⁵⁵ L.E.K. (2013) "Estimating Freight Avoidable Costs Final Report" available on the NR website here.

⁵⁶ Halcrow (2008) "Independent Reporter A: Reporter Mandate – Variable Usage Costs Final Report" available on the ORR website <u>here</u>.

⁵⁷ We note that this effect could also occur in the "lower" scenario if freight operators' avoidable costs were covered by a direct grant to Network rail, removing them from charges.

Option 1: Fixe	d charge based on avoidable cost
	level of financial bids. Therefore, the burden of the charges should ultimately lie with funders.
	We note that the modest operator-level change in charges identified for this stylised operator is likely to mask differences in charges within the different train services that an operator runs. In particular, lightly used routes and demanding services such as peak services might face significantly higher charges than those running along the same section at off-peak times. Operators therefore may seek to rebalance their service patterns towards the times of day routes where charges are lower if it is profitable for them to do so. In the current SoW, this is unlikely to be the case. However, in the "Dynamic rail" SoW such effects could be more likely to be achieved.
Franchised regional passenger operators	As for franchised commuter passenger operators, except we assume that avoidable cost charges could be higher than for commuter operators due to less intense usage. Setting the level allocated to franchised passenger operators consistent with full allocation of costs given the small rise for commuter operators above and the fall for intercity discussed below, indicates an increase in charges of around 10 percent (assuming there is no direct grant to Network Rail). This is consistent with the assumption that less intense use of assets would result in higher charges. If the change were to be in line with that indicated above, and we were in a SoW with less protection for franchise holders such as "Dynamic Rail," we estimate that they could need to pass on approximately 80 percent of this increase in order to maintain even just half their profit margin. However, as explained above, we estimate that much of this impact is likely to rest with funders.
Franchised inter-city passenger operators	As for franchised commuter passenger operators, except we assume that avoidable cost charges would be lower than for the passenger operators above. This is based on the assumption that this stylised operator group mainly runs on "primary" lines where the marginal costs previously modelled were 27 percent lower than the network average. If this difference were to exist for average costs, such a reduction in fixed costs could reduce total charges for this stylised group by around 20 percent, with the move in charges being offset by the increases for other operators. One open access operator expressed the view that the estimate of lower charges for inter-city services and higher charges for regional services was counterintuitive. While detailed analysis would be required to establish avoidable cost charges in each case, with a number of opposing factors at play, at a review of charges workshop held on 17 th September 2015 we found that this finding (at least in directional terms) was in line with those of the ongoing Brockley Consulting cost allocation study for Network Rail. We understand that one of the key factors in the disaggregation work is the effect of intensity of use but we do not expect it to capture effects around costs related to any modifications that permit operation at high speed, which would not be made in their absence.
Open access passenger operators	Open access passenger operators do not currently pay FTACs but we assume that they would in the "mid" and "high" scenarios modelled. Given limited information in this area, we make a simplified assumption that the fixed costs for open access could be as high as those for inter-city operators, and model the same ratio of fixed costs to variable costs for this sector as an upper bound. We find that implementing that could be more than enough to wipe out the stylistic operators' profits, resulting in operators exiting the market if they are unable to pass

on charges. For our stylised open access operator, we estimated that they could need to be able to pass on approximately 75 percent of the rise in charges simply to maintain just half of their profit margin. This raises questions about the role of ability to bear charges protection for open access. However, it also raises a questions about the extent to which avoidable cost attribution could place costs onto open access operators, if it is not supported by grant funding.

If they were able to bear the increase in charges modelled, all else being equal, we estimate that the sector could recover approximately half of a percentage point of total sector charges. While this is not obviously a material sum it might have potential to be larger in the "dynamic" and "on-rail comp" SoWs. At a minimum, it might at least provide a mechanism through which open access operators could pay for commercially viable enhancements to their benefit.

Freight As freight operators do not currently pay FTACs, if avoidable cost information were used to set charges for freight, all else being equal, there is likely to be an unambiguous increase in charges prior to consideration of mitigating mechanisms such as:

- the ability to bear test if the charge is a mark-up; and
- potential offsetting grants if the charge were considered a directly incurred cost.

Freight operators might respond to unmitigated increases in charges by adjusting their routing to where their charges might be lower or by withdrawal from unprofitable markets. However, in this section we focus on potential static impacts to gauge the magnitude of impact that might stimulate such a change in behaviour.

Multi-customer freight is the more price-sensitive of the two freight flows modelled, so it might not be able bear much or any charge. We modelled between 2.2 and 5.2 percent of fixed costs being allocated to the freight industry as a whole based on the level of net freight avoidable costs estimated by L.E.K. prior to PR13. All else being equal, the indicative modelling suggests that the sector could double or triple its share of industry charges but that even the lower bound might be sufficient to wipe out the market segment's profits many times over. We estimate that in the "mid" scenario the stylised multi-customer freight operators could need to pass on around 80 percent of the increase in charges to customers to maintain just half their profit margin. We estimate that this could increase to approximately 90 percent in the "upper" scenario. Although there is potentially a significant amount of Network Rail's cost that could be attributed to this sector, our modelling suggests that only a relatively small portion could be translated into charges without some serious impacts on the viability of operators.

While we understand that they are less susceptible to competition from road, bulk operators already face the freight-specific charge and some the freight only line charge and it the extent to which this sector can bear further mark-ups is unclear. Our static modelling indicates that all else being equal, bulk operators could need to pass on similarly high proportions of the charges to their customers to maintain just half their profit margins; approximately 80 percent in the "mid" scenario and around 90 percent in the "upper" scenario. While here we simply examine the static impacts of changes in charges, such significant changes in charges could result in reductions in freight volumes and potentially the withdrawal of some operators from the market.

One freight operator highlighted that in the consideration of freight costs, it is important to take into account Network Rail's own freight traffic, operated for operational purposes, such that the costs of such usage of the network are not

Option 1: Fixed charge based on avoidable cost	
--	--

mistakenly attributed to the commercial operators in the sector.

Network Rail As a fixed charge, the incentive properties of this option for Network Rail are limited. While Network Rail could be able to take advantage of better information on the network, such information may be available to it in the future independent of whether it is used to set charges.

As with other aspects of charging, we envisage that Network Rail would play a major role in designing, implementing and running the approach on an ongoing basis, with peaks in activity occurring in the run up to each periodic review.

We anticipate that these would primarily be realised as an increase in central support costs. Network Rail's support costs are substantial. Core support costs (excluding group) of £1,860m⁵⁸ were allowed by the ORR for CP5. However, the regulatory and planning aspects of it are a small element of this. It is possible that the cost of implementing avoidable cost could run into a small number of millions of pounds given the requirements to dedicate full-time teams to develop the methodology, consult with stakeholders, collate data, and calculate the charges over a number of years. The set-up period might be the most costly stage but also the most informative.

While the work required to establish and calculate avoidable cost charges is clearly additional to current activities, it is possible that the additional information could allow Network Rail to operate more effectively and even displace some existing activities focussed on improving understanding of assets and costs.

At its July 2015 Charging Workshop, the ORR presented case study evidence that indicated that the benefit of simply having better information on costs could be £200m or higher over the course of a control period, regardless of whether such information were put into charges. While work might be required to examine whether avoidable cost information is the best type of information to unlock such benefits (is information based on allocation of the regulatory asset base of value outside of charging?), the potential scale of benefit indicated might justify serious investment to unlock such significant benefits. However, if that is indeed the case, it is something that should be taken forward independent of charging and is surprising that it has not been pursued on a large scale to date.

Funders While the overall pot allocated through charges is unlikely to change in the short-run due to this charge alone, its allocation between funders might. This could for example occur if ScotRail were found to have lower charges than under the FTAC allocation and consequently require less subsidy through Transport Scotland.

With the potential for such winners and losers at funding level, it is possible that were full cost attribution made available, there might be tension between such potential winners and losers as to whether the information should be used to set charges.

We envisage that avoidable cost charging would remove the current division between England and Wales, and Scotland in terms of how fixed costs are allocated. Currently, English services pay no FTAC for the use of the Scottish network and vice versa. Fixed charges would instead be based upon track usage, not origin of funding.

As noted above, the availability of avoidable cost information is somewhat separate

⁵⁸ Table 5.1, ORR (2013) "Periodic Review 2013: Final determination of Network Rail's outputs and funding for 2014-19" available on the ORR website <u>here</u>.

to using it for charges so the greater transparency it could provide funders on what they are paying for on behalf of tax payers, and how that might affect their decisions is somewhat out of scope, but could be a key benefit of the approach. It is funders and through the HLOS process that key investment decisions are made. Depending on how it is implemented however, funders might have a new tool with which they could recover commercially viable investment costs from the passengers and freight users that might benefit from it in cases where they have demonstrable willingness to pay.

PassengersFare policy is currently separate to the access charging process. Therefore, greaterand freighttransparency on the level of costs incurred by the passenger operator they travel withusersis not likely to be apparent.

The extent to which passengers might benefit will depend on the degree to which Network Rail, operators and funders change their behaviour in response to the charges. The impact of specifically using the avoidable cost information in charges (as opposed to simply having the information available) is directly related to the extent to which operators respond to the charges. In the current SoW, we do not anticipate a material change in their behaviour. This might even be the case under all SoWs for public service obligation services, which are likely to continue to require tightly specified patterns of service.

The charging benefits of this option therefore, are most likely to be realised for commercially viable services. The extent to which a service is commercially viable will to some extend be a function of how avoidable costs are defined (i.e. a service that currently appears to be commercially viable may no longer be considered to be so if avoidable cost charging results in higher charges and vice versa). However, one of the key benefits of avoidable cost charging in the "Dynamic rail" SoW could be that it has the potential to empower operators to pay for commercially viable enhancements, improving the level of service passengers receive.

The potential benefits for freight operators are similar for freight users in terms of in the long run being able to benefit from use of a network that better fulfils their needs. However, the impact on the prices they face is more direct. Oxera (2015) examined the impact of a cost-driven reduction in freight volumes.⁵⁹ That study found that, assuming a cost elasticity of -1.35, a ten percent reduction in freight volumes could result in an annual consumer surplus loss for freight users of £60m for those who continue to use rail and £3m for those that can switch to road.

While avoidable cost charging might give users greater clarity over what is included in their charges, avoidable costs is clearly an avenue through which a greater share of costs could be charged to the freight sector. Avoidable cost charging is unlikely to result in lower costs for these users. For some users at least, this higher level of costs might result in a switch towards transporting their goods by road freight. The ability to do so will be greatest for customers in the multi-customer flow category. The impact in the bulk sector might be greater, with fewer alternatives available to them. This might have an impact on the competitiveness of those users in the industries within which they operate, resulting in lower profits and lower volumes of freight being demanded.

The existence of, and magnitude any such impacts might be tempered by decisions

⁵⁹ Oxera (2015) "What is the contribution of rail to the UK economy?" A report prepared for the RDG. Available on the RDG website <u>here</u>.

regarding the extent to which attributable costs are put into charges for the freight sector or the extent to which they are offset by grants.

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was some consensus on the investigation of avoidable cost information as a route to improving understanding of Network Rail's costs but there was only limited support for using such information in charges, particularly in the current SoW. Despite this, some passenger operators saw this as means to make the FTAC more rational, particularly if combined with the removal of the capacity charge, which was seen to allocate fixed charges in a similarly arbitrary manner.
- Passenger operators expressed the view that it is important to distinguish between informational benefits and those from using avoidable cost information in charges. They saw minimal benefit from avoidable cost charging in the current SoW particularly given the burden of calculating the charges and ensuring that operators without public service contract income would be able to pay the charge and remain financially viable.
- The issue of ensuring the viability of freight operators was shared by all industry participants, with
 some passenger operators expressing the view that this issue must be explicitly addressed as it is
 an inevitable requirement. They also expressed the view that it is essential for further work to be
 done to close down the legal uncertainties of this option either at this stage of policy development
 or at least before implementation. A particular legal issue for many parties was the status of the
 charge, a "mark-up," "directly incurred cost" or something else.
- Passenger operators considered it important to address the impact on service sponsors and funders, particularly those associated with regional public service obligation (PSO) contract.
- Open access operators expressed the view that there is a strong link between the existing charges structure and processes for capacity allocation and Network access processes. Existing Open Access operators do not pay FTAC with one rationale being that this reflects the different markets served and the fact that they access to the network in a different way. In particular, Franchised Operators are able to access the whole of the relevant passenger market while the Open Access operator access is controlled by the Not Primarily Abstractive (NPA) test amongst other mechanisms. If Open Access operators were to face an avoidable cost charge, there would need to be a parallel adjustment to the Capacity Allocation process.
- Transport Scotland explained that this option is only significant to Scotland if there is a move away from the current model where all Scottish fixed charges are allocated to the ScotRail franchise.
- Network Rail considers that the current approach that is used to allocate Network Rail's 'fixed' costs is too simplistic; it could be improved to better reflect the actual underlying railway economics. It considers that the informational benefits of this could be powerful, even if it were concluded not to fully reflect this in charges.

ANNEX C ADMINISTERED SCARCITY CHARGE DETAILED ASSESSMENT

Option 4: Administered scarcity charge								
Overall performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
+	++	++	-	+	+	++	+	

The purpose of a scarcity charge is to encourage the release of lower value uses from scarce capacity and enable higher value uses to take over that capacity.⁶⁰ Many important parts of the railway network are used to full capacity, and in some cases there are valuable additional uses that are prevented, or required to use inferior timings or routings. An additional variable charge, applied only to 'scarce' parts of the network, could encourage better use of that scarce capacity, e.g. by discouraging low value uses of the network or by encouraging Network Rail to find a way of scheduling more trains.

We have carried out a detailed assessment of an administered, value-based, scarcity charge because it appears to be the most practical form of this type of charge. The alternative to administered is market-determined, but that appeared impractical given the difficulty of devising workable mechanisms consistent with the complexity of railway scheduling. The alternative to value-based is cost-based, but that appeared too complex and less relevant.

A reservation charge, which has also assessed in more detail, is a different type of charge related to efficient use of network capacity. A reservation charge and administered scarcity charge could both exist in the same charges regime.

The scarcity charge would be an additional variable charge, i.e. levied in addition to the variable usage charge. Our interpretation of European legislation suggests that it permits a scarcity charge, which is not a mark-up, in addition to the cost directly incurred within the charge for the minimum access package. It would therefore not be legally limited by ability to pay. It does not have to be restricted to locations formally declared as congested infrastructure.

The definition of value we are using is the commercial opportunity cost, the net income lost by putting a scarce resource to one use rather than another use that has been excluded. We initially suggested the opportunity cost could reflect full economic value, which would include non-financial socio-economic values. However, our further work suggests that it is more practical and appropriate, at least in the first instance, and particularly in the present state of the world, to use the 'commercial' opportunity cost. There is precedent for such an approach in other industries, and was the basis of earlier studies for ORR on its application, and presents opportunities to keep it simple and practical, through being narrow in application.

To focus on the key issues and minimise potential issues of feasibility and complication, we have envisaged a simple scarcity charge that would apply as a path charge to only the busiest and most commercial corridors, with a high peak charge, a much lower off-peak charge, and no charge at times of day when there is spare capacity. We assume that the path charge would apply to all services passing through the key bottleneck of the path. In this simple vision of the charge, many, or even most, locations in the network, where the infrastructure is used to capacity, would not attract the charge, because such is the level of non-commercial services operating there that capacity would not be said to be the prime constraint on expanding commercial operations. It would not correspond to the technical EU definition of "congested infrastructure."

⁶⁰ What is meant by "value" in this context is discussed later in this section.

Clearly a more detailed scarcity charge could be implemented, charging many more nodes according to their relative scarcities. For example, the definition of opportunity cost could be extended to include social value but would complicate the calculation and application of the charge. Nevertheless, we can note that there are options for a more sophisticated charge if the simple charge works well and is deemed worthy of extension.

We have assessed that in the current SoW such an administered scarcity charge could have positive benefits in incentivising more remunerative use of existing capacity among commercial users of that capacity. Whilst discouraging uses of low commercial value, it would not entirely resolve policy issues in relation to competition between open access and franchised operators for commercially profitable services. This is because it does not address issues relating to abstraction and crosssubsidy of subsidised services.

This charge would also have important effects at the boundary of commercial and subsidised services, where they share the commercially valuable capacity, making funders decide whether the social benefit of their funded service justified paying the premium. Some might consider this an inappropriate way of making trade-offs between commercially beneficial and socially beneficial uses of infrastructure. For services that provide social benefits but are not procured by funders e.g. freight services, they may not be able to accessing funding to offset the effect of this charge.

In alternative SoWs which introduce greater on-rail competition, or which allow train operators to play a greater role in capacity allocation, this option could have greater positive impact. In SoWs where train operators are more protected from charges, or where funders play a greater role in capacity allocation, it would be less relevant.

Any 'premium' charge has the effect of increasing the price of access and reducing demand. Therefore, if the current capacity charge remained, we would not expect the scarcity charge to be added to the capacity charge, and it would be inappropriate to do so. In locations where both a capacity charge and a scarcity charge applied, the actual charge would be the larger of the two.

A scarcity charge potentially works well with a geographically distinguished variable usage charge (VUC). Previous work by ORR on a geographically disaggregated VUC suggested it would be low in intensively used areas, and thus a countervailing charge representing that intense use would be useful to avoid further concentrating use in those areas. In practice the simple and focused scarcity charge we have envisaged as most practical to implement would only have this countervailing effect in a few locations of particularly high scarcity. Ultimately a more sophisticated scarcity charge would work best with a more granular VUC.

Key characteristics

Description of option

Scarcity costs arise where the presence of a train prevents another valuable train from operating or requires it to take an inferior path. Here we consider a charge, administratively set, which aims to reflect a scarcity value of a path, based upon the opportunity cost of that path. The purpose of the charge would be to create a higher level of respect for scarce rights, and discourage some lower value uses from occupying them. We believe such an approach, if it proves practical, could make some steps in this direction, but it would be unlikely entirely to resolve the issue of excess demand for paths.

An administratively set charge can be contrasted with a market-determined charge, where a market mechanism would reveal the value. A charge (necessarily administered) based upon the opportunity cost can also be contrasted with a charge based upon long-run marginal cost, where the measure of scarcity value is the cost of providing additional capacity, rather than the opportunity cost of that

capacity not being available.⁶¹ We considered these alternatives in the initial list of options, and the administered charge based on opportunity cost was selected for detailed assessment. This is because we believed it was, or at least could be made, more practical than the alternatives. In brief summary, our reasons are as follows.

- Auctions: we believe that market mechanisms for path allocation such as auctions are likely to be impractical for train paths they have not even been implemented for airport slots, for example, which is a simpler and more commercial case.
- Long run marginal cost: it is difficult to measure,⁶² and because expansion costs of railways are frequently high and lumpy, the opportunity cost is likely a more relevant measure of scarcity most of the time.

Whilst administratively determined prices are often criticised as being likely to be imperfect measures of value, because administrators are unlikely to be able to predict a market outcome accurately, there are number of potential advantages of the administrative approach to price setting in this context:

- The prices can be adjusted over time to move towards a more desired outcome.
- The prices are much less subject to market volatility, thus facilitating investment more easily.
- One can start with a simple and narrow conception of it, and if that proved to deliver net advantage, one could gradually expand its application and increase its sophistication so long as that appeared to deliver further advantage.

For this reason, we envisage implementing it in a simple way for the present evaluation. If the simple form of it turns out to be feasible and to generate some marginal advantage, one can consider making it more sophisticated.

Explanation of choice of commercial revenue

There are different measures of opportunity cost:

- the opportunity cost to a specific operator, reflecting what it might bid for a slot at auction, if an auction were feasible;
- the net commercial opportunity cost to the industry, reflecting the additional commercial value, taking into account reallocation of passengers and operating costs; or
- the net social opportunity cost to society, reflecting not just financial benefits of revenue and cost, but also social benefits which are considered in transport appraisal, such as journey time benefits, crowding relief, road decongestion and environmental benefits.

We do not find the first of these approaches appropriate in the context of the GB railway industry, because it is not well-defined or inherent to the slot itself. This is because it would depend considerably upon the identity of the operator, their present market situation, and the revenue abstraction opportunities available to them.

The third of these would represent the full opportunity cost to society of the path. However, we suggest it is not suitable for implementation as a scarcity price, at least not initially, and would be

⁶¹ In a market where capacity expansion was available in small increments, and without such long term commitments, then capacity should be expanded to the point where the marginal cost of capacity expansion and the opportunity cost of scarcity are equalised. This is the rationale for considering LRMC and opportunity cost as comparable measures of the same thing.

⁶² You need to decide what the appropriate capacity expansion measure is, which is likely to be very specific to each location, and it appears to be difficult to know what projects are in practice feasible to deliver; then you need to cost it, and it appears to be very difficult to cost railway infrastructure projects.

more relevant in other states of the world. This is because:

- Such a charge would be more complex to assess, and would result in applying the charge to a much greater proportion of paths than the remaining alternative. It would be inconsistent with the practical aims of our, initially simple, charge we have envisaged. These are sophistications that could come later if thought appropriate.
- This approach would be inconsistent with the manner in which funding decisions are currently
 made in the railway industry. Funding decisions are typically agreed where a cost-benefit ratio is
 substantially higher than one. Therefore, including the full social cost in the scarcity charge, and
 therefore in the costs of providing a particular service, would overstate the willingness to pay of
 society for these social benefits, although this could be dealt with by applying a discount factor to
 the social benefits reflecting the hurdle cost-benefit ratio.⁶³
- Increasing the marginal cost of provision of funded services, to use infrastructure that was publicly funded, (even though the price paid returns to the public sector), may be politically controversial, and we find it hard to recommend it in the first case. It may become more relevant in a different state of the world where services are subsidised by different mechanisms that facilitate operators to make decisions on how much to provide.

The remaining option is the second approach, which attempts to produce a scarcity charge similar to a price that would be determined by an auction as it reflects only the commercial value. Commercial value, avoids the issues of abstraction that exist in railway competition in GB. It is also well-defined and not context dependent. There are likely to be a relatively small number of paths that have a positive marginal commercial value, which helps to simplify the implementation. It is also similar to the approach taken by ITS (2005)⁶⁴ in its scoping study of scarcity charges for ORR.

In EU legislation, Directive 2012/34 provides that a scarcity charge, can be a charge included in the minimum access charge for the path. Thus we interpret it as a charge that is additional to the variable usage charge (the cost directly incurred), but is not a mark-up based on ability to pay. Nor is there any necessity to declare a location to be congested infrastructure to apply it. This is consistent with how scarcity charges are implemented in the several other countries that apply them.

Rail access charges in several major European countries include an economic scarcity charges in various forms, all administratively determined, as opposed to market determined. From various sources^{65, 66} we find documented scarcity charges in:

- Austria: train-km charge on bottleneck sections;
- France: path reservation charge, charges differing by line type;
- Germany: train-km charge on bottleneck sections, and differing rates by line type;
- Italy: charge for passing a congested node;
- Spain: charge per seat-km on certain lines; and
- Switzerland: charge for passing a congested node, and a path cancellation fee.

We do not assert that these charges have been specifically calculated as an opportunity cost or

⁶³ See for example the general manner of analysis in *A* theory of incentives in procurement and regulation, Laffont, Jean-Jacques; Tirole, Jean, (1993), where a shadow price is attached to public funds, an arithmetically equivalent method.

⁶⁴ Scoping study for scarcity charges, (2005, revised 2006), C Nash, D Johnson, Institute for Transport Studies, University of Leeds, and J Tyler, Passenger Transport Networks, York, report commissioned by ORR.

⁶⁵ European Conference of Ministers of Transport, Charges for the Use of Infrastructure in ECMT Railways, 2005

⁶⁶ Railway Access Charge Systems in Europe, M Vidaud, 10th Swiss Transport Research Conference (2010).

indeed by any specific method. However, they demonstrate the usage of administratively determined charges applied to represent the scarcity of the capacity. This is precisely what the scarcity charge we envisage here does, but we have also envisaged a specific calculation method which may differ from these European examples.

Administered scarcity charges have also been applied in the UK in other industries. Ofcom applies administered scarcity charges for parts of the radio spectrum. It has also had market determined prices – the Global System for Mobile Communications (GSM) auctions – but more generally applies administered charges because it believes that market mechanisms are impractical or unreliable in most parts of the spectrum, which requires a management process with some parallels to railway scheduling.

Application

To maximise simplicity and feasibility of this option, we envisage that the application could be confined to rail corridors which are dominated by uses of high commercial value e.g. the three main intercity corridors, and possibly some additional corridors if it was found that there was in practice true commercial scarcity on those corridors.

Again for simplicity, we envisage charges would be path charges rather than node charges, but path scarcity would be identified in relation to a key bottleneck or section that acted as the prime constraint on the number of available paths. Services using that critical section, during time periods of scarcity, would attract the charge.⁶⁷ We envisage the charge would not apply – at least initially – in locations where capacity is fully used but there is a substantial presence of funded services, as the substantial presence of such services implies that the capacity is not scarce in a commercial sense, but it may be scarce in another sense.⁶⁸

We would envisage that paths allocated late or on an ad hoc basis would not attract the charge, as such applicants are accepting only what capacity is available after other requests have been satisfied. This exemption would make it easier for freight and charter operations to remain profitable, while not limiting the ability of other operators to obtain paths.

We would envisage potentially several levels of charge for a particular corridor, carefully defined by time period and direction having regard to the specific demand characteristics of the corridor. A typical approach might, for example, have two charge rates for periods of the day where the corridor is used at capacity⁶⁹ but with different relative values – broadly peak and off-peak – and a zero charge for times of day when there is spare capacity on the corridor. We would also envisage the charge would not apply when a path was made available under late or ad hoc arrangements.

In locations where both a capacity charge and a scarcity charge applied, the actual charge would be the larger of the two. This is because we do not think that it is appropriate to charge both 'premium'

⁶⁷ In a sophisticated approach to scarcity charging, one could have bottleneck or node charges, and a single path might attract several charges. But it would be difficult to value scarcity at such bottlenecks, as complex techniques may be required, as in the electricity industry. By making it a path charge, and as far as possible focusing on a single key bottleneck for that path, we attempt to retain simplicity of application and calculation. ⁶⁸ This test would probably have to be made more precise in a practical implementation.

⁶⁹ It is observable that the number of paths scheduled in the peak on a main intercity line such as WCML or ECML is several more than during the interpeak period. Nevertheless ORR has not required NR to satisfy applications for additional paths in that interpeak period, accepting NR's view that the timetable is full, for whatever reason NR has given for this. Thus we are bound to accept – as has been argued by stakeholders - that there is scarcity in the off-peak period also, potentially commercially valuable, even if the number of movements is less than during the peak, at least during periods where scheduled paths are at the interpeak plateau level. During periods of the day, typically late evening and early morning, when the demand falls below this plateau level, we would argue that there is not scarcity.

charges where they would both apply. However in modelling it we have not taken this set-off into account. As we have envisaged it, it seems likely that the scarcity charge, where it would apply, would normally be much larger than the present capacity charge, thus given the size of the uncertainties we believe it is a fair approximation to ignore the set-off effect.

A scarcity charge potentially works well in tandem with a geographically distinguished variable usage charge (VUC). Previous studies on geographic disaggregation have found that the more lightly used parts of the network often, for good reason, have higher variable operating and maintenance costs. Therefore, a geographically disaggregated VUC risks encouraging traffic from lighter used sections towards the more densely used parts of the network.

A scarcity charge, applied in locations where such capacity was in fact scarce, would provide a countervailing incentive to avoid counter-productive traffic transfer. In practice the simple and focused scarcity charge we have envisaged would only have this countervailing effect in a few locations of particularly high scarcity. Ultimately a more sophisticated scarcity charge of greater sophistication would work best with a more granular VUC.

Description of counterfactual

The current regime contains a variable charge to recover short-run marginal wear and tear costs only. Variable charges currently neither capture short-run nor long-run marginal scarcity costs. Furthermore, there is no direct link between the need for enhancements and charging at present. A variable capacity charge exists: but this is related to the fact that additional traffic increases the risk of performance shortfall, and compensates NR, at the margin, for this effect. Franchised operators are protected against changes to track access charges.

Timetabling is managed administratively, and related to the franchising regime and new capacity is predominantly funded by fixed track access charges, the Network Grant and NR's commercial income.

Relevant factors impacting the form and/or the effectiveness of the option

Practical issues relating to the imposition of a opportunity costs based scarcity charge (by reference to the 'Factors' report) include:

- 2.2 EU legislation and 2.3 UK legislation which raise issues requiring clarification about the legal approach to applying a scarcity charge in rail.
- 3.2 The franchising regime. This would limit the impact of any such charge on franchised operators, at least in relation to services that they are required to operate.
- 3.3 A mixed use network. The varying business models of UK operators are not necessarily compatible with a scarcity charge which would not value wider social or environmental benefits, at least in the absence of other complete mechanisms which take them fully in to account.
- 4.4 Economic viability a scarcity charge may have a detrimental impact on smaller operators and freight operators. This could potentially be mitigated if there was a funding source or system within the charging system to reflect social benefits these operators provide which are currently unremunerated.

Implementation	
Information requirements	The information requirements for application of the charge, assuming it is implemented in the simple form we discuss, are modest. The design of the charge would identify which paths attract the charge, and then it would suffice to bill those rail operators who are allocated such paths at the relevant rate. The size of the dataset would be small, the charges well-defined, and not based on any detailed track usage. Our view is that this charge would be simple to apply.

Option 4: Admin	nistered scarcity charge
	The calculation of the charge would be very similar, both in its delivery and in the informational requirements to support it, to the calculations that are routinely made in support of franchise specifications. Thus these are calculations of the type that the government routinely procures today.
Drivers	The charge would increase as the commercial value of the use of corridors excluded due to scarcity increase. The charge would be levied on users of identified commercially valuable corridors who occupy capacity in a way that excludes such valuable users. The charge would tend to be much higher at peak times, defined in terms of the times of day which, for that corridor, carry the most valuable traffic.
Calculation principles	First, the valuable corridors and the quantity of available paths need to be determined. The quantity of available paths needs to be understood in practical terms, and may vary by time of day. ⁷⁰
	Next a standard planning tool such as MOIRA, in tandem with an accepted costing tool, would be used to identify the most valuable use of an additional path on the corridor, if such an additional path was made available. The appropriate increment would probably be one path per hour rather than one single path, in order to be able to identify values of paths averaged across suitable periods of the day, and also since train and personnel costs are probably more suitably considered over the course of a day. The appropriate measure of revenue is revenue of all passenger operators, taking into account abstraction, not a single passenger operator's revenue or train revenue. The DfT holds passenger train operator cost models from franchise bids, and these would be appropriate tools to estimate the marginal cost of additional train movements to service these paths, though probably some judgment on practical diagramming of stock and staff would be required. As noted above, we envisage its use only for paths which are scarce for usage of commercially profitable services. Where the line between "not many" and "many" subsidised services sharing the relevant capacity bottleneck lies is bound to result in some difficult intermediate cases, and probably a uniform test might be required. One possible test would be if all the relevant capacity could potentially be used for commercially valuable services.
	The scarcity charge would be a supplement above the total charge presently levied, and therefore the continuation or cessation of present capacity charge will considerably affect what is the appropriate magnitude of the scarcity charge.
Practical considerations	Some careful analysis and judgment would be required to determine which are the valuable corridors that should attract the charge, identify the essential bottleneck, and manage practical capacity considerations at different times of day. There is bound to be an element of opinion and lobbying of the body responsible for concluding on it.
Lead time	Once it was determined that such a charge was to be introduced, the lead time for determining its value would be similar to the lengths of time in making existing regulatory price control determinations. As we envisage it, with only a very limited number of corridors attracting the path charge, the practical implementation of the

⁷⁰ As noted above, NR the number of paths in a "full" timetable may vary materially from hour to hour. To the extent that ORR has not required further path applications to be accepted, we have to accept that this can be a "full" timetable.

Option 4: Adr	ministered scarcity charge
	charge would not be onerous.
Resources required for implementati	The main burden of implementation would lie on ORR, in determining where the charge would fall, and calculations of the level of charge. The assistance of NR and on input from operators would also be required in indicating relevant pathing issues. Some thought may also be needed on access management of other "lesser bottlenecks" that would not be charged, to determine that track access management methods were done in a way consistent with the charging philosophy.
Performance	against criteria
Axioms	Current Dynamic On-rail Specified Protect Beneficiary Capacity Regional railway comp franchises freight pays allocation powers
System safety	
	There is no unambiguous material impact of this charging approach on system safety, but charging for the use of scarce capacity might decrease pressure on assets in constrained areas and therefore have some positive benefit on safety. However since we are proposing to apply the charge only to relatively limited areas of the network, it would likely also be small in extent.
Consistency	
	Paragraph 4 of Article 31 (directive 34/2012) states: <i>'The infrastructure charges referred to in paragraph 3 may include a charge which reflects the scarcity of capacity of the identifiable section of the infrastructure during periods of congestion.'</i> Some stakeholders have argued to us that law only permits that a scarcity charge may be levied only in locations which have been declared as "congested infrastructure" (Article 47). At present only a handful of (rather unlikely) locations on the GB network meet this test, although a number of others are expected to be added. The Article 47 test requires that access requests have to have been turned down in order declare it as congested infrastructure. In the UK, there are a large number of likely congested areas where operators would not put effort into an access application because the constraints are already widely known, hence are not likely to exhibit an explicit denia of a request. We would however observe, in opposition to this view, that there is not cross-reference between Articles 47 and 31, nor is the defined term "congested infrastructure" used in Article 31. The directive recognises that infrastructure may cease to be congested if scarcity charges apply. Other EC countries have applied scarcity charges widely across their infrastructure.

⁷¹ The Commission Implementing Regulation (EU) 2015/909 of 12 June 2015 on the modalities for the calculation of the cost that is directly incurred as a result of operating the train service

Option 4: Adr	ninistered	scarcity cha	rge						
Funding of NR	=	++	++		+	=	+	=	
This option would be a new charge through which NR could recover in requirement. Forecast receipts would be matched by a corresponding re- fixed charges so overall we do anticipate any impact on NR's aggregate func- lif funds raised from the charge were regarded as a contribution to en- costs, then there may be a positive impact on this criterion but this would the value attached to the charge. In SoWs where there is more comp- capacity, this option would have greater impact. Where there competition/greater protection the option would deliver little if any added We note that the overall level of receipts will be a function of the gap betw and demand and operators' price sensitivity. Where capacity is truly const demand outstrips demand, there could be an increase in overall receipts fro- charges. However, if the charges were poorly targeted and levied on s where there were no constraints, any subsequent reductions in demand of in a net decrease in variable charge receipts.								is revenue duction in ing. nancement depend on etition for e is less value een supply rained and om variable ome trains ould result	
Allowance for		_	_		=		_	_	
conditions	A scarcity freight by that such times, only the charge The charge would refl of freeing Many ope demand of types of f rights regi capacity. S would be likely requ A cruder f operate at study a si corridor c that capaci	charge levi pricing freig a charge wor y a relatively by operatin e would affe ect the high up more cap rators requi reight. Curre me which ra carcity char accommoda ire detailed form of the t capacity and papacity, and city.	ed as a c ght usage uld only a small pro- ag more fl ect open value of pacity the re some irements ent access inge from ges raise ated. The analysis a charge w ould mak practical of manages	charge coul e of valuabl apply in limi oportion of lexibly, for e access ope the capacit y can profit additional f as they ari ss rights may fixed rights the issue of ere are way and charges thich sough e this criter charge whi s demand c	could detrimentally affect financial viability uable corridors off rail. However to the ext limited locations, and be very high only at p n of freight would be affected. They could av for example with non-firm rights. operators at an apparently high rate, but to vacity they were using. It should have the effer of itably apply to use. nal flexibility to manage, for example, seaso y arise. This is particularly the case for cert is make allowance for this through the acc ights through to options to use additional sp ie of how this required flexibility in traffic lev ways of managing this issue but these wo rges set on a granular basis. ught to price many locations which apparent riterion red across all SoW. We have sough which only identifies the key determinant and other locations administratively to prese				
A single	=	=	=	=	=	=	=	=	
approach for the network as a whole	This approach, as per the current planning led mechanism, could be applied to all operators irrespective of type.								

as a whole

99

Option 4: Administered scarcity charge									
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Service costs	=	=	=	=	=	=	=	=	
recovery	A scarcity charge levied as a part of the charge for the minimum package of access rights would reduce the size of the fixed cost that needs to be recovered elsewhere. The overall level of contribution would depend on the scale of the charge. We have estimated the amount the charge might recover as lying in the range of £800m to £2,000m. Although this is a wide range, the present capacity charge, which it would likely replace, lies in this range. It does therefore likely succeed in replacing the capacity charge, but not in making a more fundamental contribution than that								
Efficient	=	=	=	=	=	=	=	=	
whole-system whole -life industry net costs	This approa the costs o railway ind funding of I	ach could be f future ad ustry are a Network Ra	e beneficia ditional ca rranged, il and has	al, at least a apacity. Alt more likely little impac	it the mar hough as f it is simp t.	gin, in contr the funding ly treated a	ibuting in a flows in th as part of t	dvance to ne present he overall	
Efficient long	=	+	+	=	=	=	+	=	
investment decisions	This approach would be beneficial, at least at the margins, as the setting of the charge would describe clearly where capacity is truly constrained and expansion would be valuable. But arguably the method of setting the charge is just the same as the existing planning based approach to deciding where expansion would be useful, after all it is an administratively set charge. In a SoW that places greater overall emphasis on efficient use of capacity, this option might have benefit over current arrangements, which rely on historic access rights.								
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Efficient	=	=	=	=	=	=	=	=	
performance management	This approach has no direct impact on the occurrence of disruption.								
Efficient use	+	++	++	=	=	=	++	=	
of network capacity	A scarcity c required to recognise t The effect competition given to op	harge provi operate se he potentia of this op n, but woul erators.	des a fina rvices rat I for impro tion migh d have lit	ncial incent her than rel ovements to t be greate tle or no ef	ive to mor y on pre-e o capacity er in thos fect in So	re carefully existing right allocation. e SoWs wh Ws where v	consider th ts that perh here there wider prote	e capacity aps fail to is greater ctions are	
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Predictability									
	A scarcity of constraints potentially with traffic, is administe	charge may will change quite sensit , and to fluc ered.	adversely e over tim tive to tra ctuate wit	y affect pre ne, and the ffic levels. T hin or betw	dictability opportun his could een reviev	of charges ity cost of t cause charg ws dependir	generally a those const es to increa ng on how t	s capacity raints are use rapidly he charge	

Option 4: Adr	ninistered	scarcity cha	rge					
Simplicity	-	-	-	-	-	-	-	-
	Establishing the basis of the charge for the first time would be time consuming and potentially complex versus the current planning based approach – and the present access planning would still have to continue since the charge would be unlikely to make excess demand go away, nor simplify the task of scheduling trains to get best usage of capacity. There would likely be a contested decision on where the capacity was deemed scarce, exactly how it should be defined, and how much capacity it was wise to schedule. But the administered nature of the charge would make the process more straightforward thereafter, in that the methods would be set up and become easier to replicate thereafter, although changes would be required every time there was a change in the capacity or location of the key determinant of corridor capacity.							
Transparency	+	+	+	+	+	+	+	+
	Current track access arrangements are established in bilateral contracts that are no necessarily publicly available or easy to review. The imposition of a scarcity charg would bring greater transparency to poor use of capacity and result in overall near optimal use of the network, at least in areas where poor network use was a constrain upon commercially valuable use of the network							at are not ity charge rall nearer constraint
Low	=	=	=	=	=	=	=	=
transaction costs	Imposition of scarcity charges would imply a degree of monitoring and management that does not currently exist. Offsetting this would be the benefit of identifying capacity that could be allocated to better commercial value.							
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
NR	=	=	=	=	=	=	=	=
accountability	This option has no direct impact on NR's accountability, any more than any of the capacity charge mechanisms previously employed.							
Non-arbitrary	=	=	=	=	=	=	=	=
allocation of costs	Unlike a reservation charge, a scarcity charge would be imposed on top of other charges i.e. it would be an additional charge on top of the variable usage charge rather than an up-front contribution to costs that would likely be incurred in any event. The charge would be set administratively to recover the proportion of costs considered to represent opportunity costs. We find no compelling reason that it is more or less arbitrary than existing charges.							
Optimal traffic	+	++	++	=	+	+	++	=
growth	A scarcity more posi capacity fo additional	charge cou tive in SoW or use by o protections	ld contribu / where th ther opera	ute to all th nere is grea ators. Its im	e objective ter compe apact is lik	es of this cr atition since ely to be in	riterion but it should nhibited in	would be free some SoW with

Option 4: Administered scarcity charge								
Aligning industry incentives	+	++	++	=	+	+	++	=
	This option recognises that industry incentives may not be aligned and places weight on industry participants giving up unused access, or access that does not reflect the scarcity costs it imposes. Current planning based approaches seek to achieve the same thing, but the imposition of a mechanism that creates a common basis for discussions and should improve transparency. It will work more effectively in SoWs which have encourage more flexibility and thus responding to those incentives.							
Value for	=	+	+	=	=	=	+	=
funders, taxpayers and users	We do not implement other cha models. Fr would be computati made in re	believe that Measurem rges. It wo urther the r a charge p ons to calcu lation to inv	it a capac nent woul uld not number o per path late the o estment o	ity charge i d be relativ require rea f chargeab and applyi charges are decisions an	n this form ely straigh Il-time me le events ng to a s related to d franchise	a would be tforward in asurement, would be r mall propo calculatior e specificati	expensive compariso , or comp relatively s ortion of p ns currently ons.	to bill and on to most lex billing mall, as it paths. The routinely
	A scarcity of scale of im not an issu retain a si demand. T competition likely to an	charge could apact may bue on many gnificant de The impact on, and ther ise.	l contribu e small in parts of t gree of fl would be n mainly	te to all the the curren he network lexibility in larger in S on the part	e objectives t state of t and beca order to r oWs which ts of netwo	of this crit the world, a use freight nanage inh h encourag ork where	erion. But t given that would like erent fluct e a larger more com	he overall capacity is ly need to uations in degree of petition is
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers

+++++++++++In the current SoW an administered scarcity charge as described above, and if legal,
could have positive benefits in incentivising better use of existing capacity among
commercial users of that capacity. Whilst discouraging low value uses, it would not
entirely resolve policy issues in relation to competition between open access and
franchised operators for commercially profitable services, since it does not address
issues relating to abstraction and cross-subsidy of subsidised services. It would also
have important effects at the boundary of commercial and subsidised services, where
they share the commercially valuable capacity. It would make funders of subsidised
services pay to occupy this commercially valuable capacity, considering therefore

more carefully whether the social benefit that this brings justifies that loss of commercial value. Some might consider this an inappropriate way of making trade-offs between commercially beneficial and socially beneficial uses of infrastructure. We can also note that not all sources of social benefits on the railway network are capable of accessing funding to defend them, e.g., freight, as things stand.

Our modelling indicates that it could adversely affect freight services. Although, as we envisage it, it would likely only affect a small proportion of freight movements, the size of the charge is likely to be sufficiently large to amount to a material impact. There would be some scope to design it to limit that impact, and for freight operators to take action to reduce the impact.

Impact on franchised passenger operators under the current regime are likely to be limited given that current arrangements provide protection from change and also limit the scope to change service levels. But it may encourage funders to consider whether

their use of scarce capacity gives value for public funds, when they are excluding commercially valuable services.

Assuming adverse impacts could be managed e.g. that freight operators retained the flexibility required etc., the overall impact could be positive but may be small overall in financial terms. The main areas of uncertainty would lie over which parts of the network really do have a high load of valuable commercial services through bottlenecks, as opposed to a high load of taxpayer funder services, and are therefore genuinely scarce in economic terms. More detailed analysis, involving detailed capacity, revenue and cost analysis of specific corridors, would be required to assess where the charge should be introduced and clarify its likely level in different corridors. This may also help make clearer whether the benefits outweigh the disadvantages.

However, in alternative SoWs which introduce greater on-rail competition, or which allow operators to play a greater role in capacity allocation, this option could have greater positive impact. Conversely, in SoW where operators have increased protection from change, or capacity allocation is more strongly driven by funders, the option would have less impact, and maybe overall would not be worth implementing.

Impact on stakeholders

The broad effect of this option is to charge the users of valuable capacity, and incentivise them to consider whether it is of sufficient value to them to pay the charge. For franchised services, in practice it is funders rather than operators who would ultimately feel the effect of the charge in the current SoW. The users of this valuable capacity are mostly intercity operators (franchised and open access). Freight operators and regional passenger operators make some small use. Regional operators are likely to, overall, be net beneficiaries as a result of reductions in other charges. The impact on commuter operators is unclear.

A high level understanding of this charge is that it works most strongly at the boundary between funded services and commercial services, in areas where capacity is particularly commercially valuable.

The desirability of introducing the option is depend on the extent to which it is considered appropriate to influence the allocation of scarce capacity in this way. It will also work to some degree at the boundary where franchised and open access operators compete for access to paths for commercial services. However, it does not resolve the issues related to abstraction and the extent to which it is seen as desirable to raise profits that can be used to cross-subsidise PSO-type services.

Franchised Taking into account an assumed reallocation of FTACs, the overall impact of this commuter charge on the commuter sector is unclear - in some scenarios there is a net passenger reduction in charges to the sector, in others an increase. This is because there are operators options for how extensive the charge would be, and this particularly affects the commuter sector. Some wide variation is also likely in the impact on individual franchised commuter operators, for similar reasons. At one extreme, some would likely not face the charge at all, because they simply do not use the corridors with commercially valuable scarce capacity. Where the other extreme would lie would depend upon precisely which corridors were determined to be scarce and commercially valuable. At one level, it is clear that some commuter services explicitly take a share of the capacity of commercially valuable intercity corridors, and such path usage can reasonably be expected to be chargeable, though we would expect parallel "slow lines" would not to attract the charge. If that was the extent of it, it would likely be of relatively minor impact on commuter operators, aided by the observation that

Option 4: Adm	inistered scarcity charge
	the longer distance intercity peaks – which would define peak charge timings – do not always coincide with commuter peaks.
	But there are also corridors which are intermediate in characteristic between commercial intercity and longer distance public service commuter, and where even the some commuter services are commercially valuable, and it could be argued that these would merit a scarcity charge – for example the mainlines from Waterloo and Liverpool Street might be considered in this category.
	In a sense, this would be the point. Highly commercial paths are identified, and proportionate charges would aim to discourage the use of them. Funders of PSO services may find it appropriate to release paths to commercial uses, and thus avoid the charge.
	As a result of these uncertainties, we have scenarios where charges are relatively high and impacting some commuter operators, and other scenarios where they are much less impacting on these operators.
Franchised regional passenger operators	These operators would likely pay only a small amount of these charges. Their usage of intercity corridors is mostly away from the critical bottlenecks on those corridors, and we would anticipate management rather than charging would mostly be used to discourage them from impacting on valuable commercial traffic. Since the operators which would mostly bear this charge would make a material contribution to industry fixed costs, the overall effect on the regional sector, when the FTAC is taken into account, would be a reduction in charge. Nevertheless detailed examination, following detailed definition of the charge, may find cases where an individual operator found itself materially liable for it. It would then be a question for service funders whether they wished to avoid the charge by reducing occupation of the paths.
Franchised inter-city passenger operators	Franchised inter-city operators will bear the main impact of the charge, in terms of total quantity of the charge. They are the operators which substantially make use of the commercially valuable and scarce paths on the network.
Open access passenger operators	Open access operators will likely experience the charge in relation to many of their services. Although this amounts to a relatively small amount of money for the sector as a whole, it is likely to be a material amount in relation to the level of access charges currently paid by open access operators. As a percentage of their present charges paid, it would have the largest impact on them.
	However, the charge aims to discourage occupation of valuable paths by lower value traffic, and would therefore potentially free additional valuable paths which open access operators could apply for. The difficulty they might have is that the Not Primarily Abstractive test may materially restrain their ability to use those paths.

Freight operators	Our estimates suggest that freight operators would only rarely incur a charge. However, as the charge is relatively large per movement, even during off-peak, it potentially represents a large proportionate increase in access charges for freight operators. It may encourage operators to seek to re-time movements to avoid the charge, or make more use of contingent and spot access rights at those locations to avoid the charge.
	Our modelling appears to suggest that the charge is more significant for bulk freight movements than multi-customer. This is because bulk freight typically moves shorter distances than multi-customer freight, and thus occupy more paths for a given train-km output. Thus it risks incurring the charge more frequently. In practice it would depend upon detailed pathing considerations which is beyond the scope of our modelling.
NR	This option would introduce a new variable charge. At a periodic review, the expected income from the charge is likely to be set off against fixed charges. Network Rail's total revenue requirement will not be affected.
	However, this would introduce more variability to network Rail's actual income. Therefore, it should incentivise Network Rail to seek to find more paths to receive the income from the charge, or even to make some small investments to facilitate this.
	The charge is likely to cause a reallocation of the use of paths rather than reduction in use of paths. In principle, following a capacity expansion, the charge might reduce at the next review. However, this would be done in tandem with reviewing the fixed charge.
Funders	Funders will find that some of their funded services attract the charge. In the case of some less commercial services, they may decide to economise on the use of paths in order to avoid funding operators having to pay it, thus releasing paths for more commercially valuable services.
Passengers and freight users	To the extent that the charge encourages a reallocation of scarce capacity for more commercial services, there will be winners and losers among passengers. Funders of PSO services have the ability to retain capacity for the benefit of less remunerative services if they believe that is on balance in the public interest, but will pay more to do so.
	Freight customers whose services occupy valuable paths risk either paying more for these services, or having to accept that their freight service provider runs the service on ad hoc or late applications, with greater uncertainty of delivery window.
Summary of inc	lustry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Several industry representatives suggested that socio-economic as well as commercial value should be included in the measure of scarcity value. However, freight and passenger operators raised a some potential risks with the introduction of a scarcity charge:
 - Some operators argued that there was some uncertainty regarding the assumption made in this report that a scarcity charge could be a "mark-up" such that ability to pay would be taken into account in its application. This uncertainty in part come from scarcity not being included as an admissible cost in the implementing regulation 2015/909, which defines costs directly incurred. Some also expressed a

degree of uncertainty regarding the assumption made in this report that under Directive 2012/34, a scarcity charge could apply beyond locations formally declared as congested infrastructure. Both these points of legal uncertainty would need to be clarified ahead of implementing such a charge.

- Using price to encourage reallocation of capacity between funded and commercial services would be politically controversial.
- The binary nature of the proposed scarcity charge could lead to a high degree of instability as the route flipped between being "full" and having a spare path.
- It could lead to gaming by a dominant operator filing a route to make it scarce and impose charges on competitors.
- If the scarcity charge applies to freight and Open Access operators, there would need to be a parallel adjustment to the capacity allocation process to make them freer to obtain paths they are willing to pay for.
- There would be boundary problems if the charge were to change by a large amount at a sharp time boundary, and difficulties if a path were re-timed to the expensive side of the boundary.
- Network Rail considers that there could be merit in carrying out more work to better understand the full economic value (i.e. societal and commercial) of each train path. They would be concerned about pricing off traffic of high social value if a purely commercial value approach were adopted.
- Transport Scotland observed that the application of a scarcity charge on the West Coast Main Line (WCML) and the East Coast Main Line (ECML) could result in disproportionately increased service funding costs for Transport Scotland, given the funding treatment of cross-border services and allocation of the FTAC. Transport Scotland has a tightly defined public service specification and does not anticipate changing that policy.

ANNEX D PATH RESERVATION CHARGE DETAILED ASSESSMENT

Option 7: Path reservation charge								
Overall performance against the RDG Vision in each SoW								
Current Sow	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
+	+	+	+	=	+	+	+	

A reservation charge is non-refundable charge for booking capacity. It is an alternative kind of scarcity charge whose purpose is to discourage users from booking capacity that they do not in practice require. Such a charge might encourage operators to manage their network usage more efficiently and discourage booking capacity beyond what they predict will be required, except to the extent a firm option of being able to run is sufficiently valuable to them to pay the charge.

The reservation charge we considered for the purpose of this assessment was a **deposit** (or obligation) based scheme payable only for **firm access rights**. It can be defined equivalently either as a non-returnable deposit which is then set off against charges for paths actually operated, or else as a charge for rights not used. We assume the reservation charge being assessed as a fee **per train km**.

While it is not what EU law would define as a "cost directly incurred," basing the reservation charge on the expected VUC would be a simple way to calculate a per km charge, which is explained in the *Implementation* section of the detailed assessment template. We have therefore used the VUC as the basis for our modelling and have assumed a separate rate would apply to passenger operators and freight.

This charge most strongly impacts those train operators that have low utilisation levels for their booked capacity, and evidence suggests that this is most significant for train operators running bulk freight services. Other train operators tend to have high levels of utilisation of allocated capacity. The chart below shows the modelled relative impact of the reservation charge on total charges paid



by stylised operators (compared to bulk freight).

In the current SoW, reservation charges could have some positive benefits in incentivising better use of existing capacity but in some forms could significantly adversely impact freight. on Impacts franchised passenger operators under the current regime are likely to be small given that the current arrangements provide protection from change and limit the scope to change service levels. This form of scarcity management has some precedent given that it is in use on HS1, in France and Germany and is being considered for Crossrail.

Assuming adverse impacts could be managed, such that freight (and potentially other smaller operators) retains the flexibility required, the overall benefit could be positive but may be small. More detailed analysis would be required to assess whether likely benefits outweigh issues such the transaction costs involved in introducing and managing a reservations system.

In alternative SoWs that introduce greater on-rail competition or which place greater emphasis on the value of capacity this option could have greater positive impact (although it would need to be



Option 7: Path reservation charge

weighed against other options for value based charging). In SoW where operators have increased protection from change, the option would have less impact.

Key characteristics

Description of option

In this option, Network Rail would levy a capacity reservation charge based on capacity that is booked. This could be in the form of an upfront deposit that would not be returned if capacity were unused. Equivalently, it could take the form of an obligation to pay the charge if capacity goes unused, which may mitigate potential cash constraints of paying deposit fees upfront while still providing incentive to book capacity efficiently. This charge would be the same as ORR's 'failure to lose' reservation charge considered in its 2006 charging workshop.⁷²

The charge itself could be calculated based on a per unit rate, for example:

- Charge per train km;
- Charge per train service minute; or
- Charge per gross tonnes train km or gross tonnes train minute.

The charge can then either be applied to units represented in either access rights or timetabled train paths. There is precedent for both in the UK (HS1) and internationally for levying reservation charges, which supports the use of a reservation fee based on per-unit charges and on timetabled train paths. A few cases are summarised further below. Previous work by ORR has also considered the pros and cons of using either timetabled train paths or access rights as the basis for the charge.⁷³

As explained further in 'calculation principles' below, we consider here a **per train km** charge applied to **firm access rights**. It is envisioned as applying to all services with firm track access contracts (e.g. including freight services carrying out work for Network Rail).

International precedent

- High Speed 1⁷⁴ HS1 levies such a charge to discourage block booking of capacity that might be used by other operators and to promote competition on the network by encouraging efficient use of capacity. On HS1, the charge is set for train operators at 25% of the applicable investment recovery charge and for freight at 25% of OMRC. The mechanism also provides an incentive to return unused capacity where this capacity is then used by another operator.
- SNCF Réseau⁷⁵ SNCF Réseau, the French rail infrastructure manager, levies reservation charges on both passenger and freight services. Reservation charges are applied to all network users who are allocated capacity (i.e. booked timetabled train paths). For freight and passenger services this is levied as a per-km charge applied to the timetabled train path and includes various adjustments (e.g. peak/off-peak). The per-km charge is calibrated to different network segments based on defined rate categories, and additional adjustments are made depending on the network segment (e.g. freight has an additional speed and length based adjustment on conventional lines).
- DB Netze⁷⁶ DB Netze, a German rail infrastructure manager, allows operators to classify up to 15% of their booked train paths (in train km) as optional. In cases where optional

⁷² ORR (2006), Structure of Charges Workshop.

⁷³ See for example, ORR Structure of Charges Workshop (2006) slide deck.

⁷⁴ HS1 Freight Access Terms (2015) and HS1 Passenger Access Terms (2015).

⁷⁵ SNCF National Rail Network Statement (2016 timetable), section 6.2.1.

⁷⁶ The Train Path Pricing System 2016 of DB Netze AG.
paths are not used, or are only partially used, a reservation fee equal to 10% of the path fees is levied. Path fees are on a per-km basis and are calculated using the train path pricing system that captures three components: user-dependent components, service dependent components and other components.

In addition, DB Netze charges a cancellation fee for cancelled paths/portions of train paths. These increase as the time to timetabled service draws nearer. They also offer discounts to the train path fee on sections of the network that are uncongested.

Description of counterfactual

Planning processes in the rail industry such as ORR's Track Access Policy, Route Utilisation Strategies (RUSs) and the Network Code currently guide industry decision making regarding the level, type and pattern of traffic on the GB rail network. They are used as the principal mechanisms for incorporating utilisation of allocated capacity and path reservation/ holding into industry decision making. Currently there are three main types of access rights that are granted to operators, all of which are time-limited as specified in respective contracts:

- Level 1 rights. The most specified access right that is firm in respect of quantum, origin and destination, equipment, timing (with Network Rail right to flex), etc. and in some cases routing.
- Level 2 rights. Firm rights that entitle the operator to a quantum of service over a particular time period (e.g. day or week). They do not specify timing of service or route.
- Level 3 rights. Rights contingent on Network Rail being able to satisfy bids for track capacity. Otherwise can share same service characteristics as level 2.

Part D of the Network Code sets out the rules governing the translation of access rights into timetabled paths and specifies the priority given to different levels of rights for inclusion in the timetable. Level 1 and 2 rights are considered as firm access rights and take priority over all other rights, apart from any rights Network Rail has for Network Services. Otherwise all firm rights are accommodated in the working timetable for which an Access Proposal to exercise those rights has been submitted by the operator. Once all firm access rights have been accommodated, then Level 3 and rights are granted timetabled paths. Part D also specifies the timeline for timetable development and may be an important determinant of how capacity is used. For example, freight operators would need to submit access proposals at least 26 days before the release of the timetable at which point there may still be volatility in demand for their services.

Part J of the Network Code specifies conditions under with Network Rail may request that rights be surrendered and for the transfer of rights from one operator to another. Part D governs Network Rail's approach to timetabling including the priority given to firm and contingent rights in the timetable. Quarterly review meetings are also held between Network Rail and freight operators in which they discuss the reasons why FOCs need to retain unused capacity, timescales for when they will use this capacity and when this will be next reviewed.

Since 2014 Network Rail has also made use of a new system called the F-CDM that allows them to produce quarterly reports on utilisation of scheduled train paths, and therefore captures the 90 day contractual period for using a path, after which Network Rail may challenge the company's retention of the right to that path under Part J of the Network Code. These reports are then used by the Capacity Management Review Group to make decisions on whether or not schedules have strategic value. Paths that are deemed not to have strategic value are then removed from the timetable, while those that do have strategic value are retained by Network Rail as strategic capacity.

Currently there is no explicit price incentive approach to the use of allocated capacity and route allocation, only the planning approaches.

Relevant factors impacting the form and/or the effectiveness of the option

Practical issues relating to the imposition of a reservation charge (by reference to the factors report) include:

- 4.4 Economic viability a reservation charge may increase costs for those operators that have need for some degree of flexibility to accommodate seasonal demand etc.
- 4.7 Data Billing etc. we are advised by Network Rail that there may be issues with updating the current billing system to allow for billing a reservation charge deposit.

Implementation

Information The information required would depend on the complexity of the reservation charge. To apply a simple reservation charge based on access rights the requirements are quite minimal:

- Number of firm access rights allotted.
- Number of firm access rights exercised.
- Units represented by each firm access right.
- Per-unit charge.

The above pieces of information would be required for the simplest type of perunit reservation charge based on access rights. As seen in the European examples above it is possible to add additional complexity that would bring the reservation charge closer to a scarcity charge, but would require additional information. For example:

- **Time-of-day adjustment**. Some access rights may specify the time of day for services. In such cases it may be possible to make an adjustment if, say, these rights fall at peak or off-peak times. This could be done as a scalar adjustment (i.e. multiplied by a pre-determined value), as is the case in the French regime.
- **Track-type adjustment**. It may be possible to incorporate a degree of differentiation by track type. In theory, track types that are more congested (i.e. more scarce) should have higher reservation charges attached to them. The granularity of track type could be quite high level. For example, the French system has 19 classifications grouped into 4 broad categories (suburban, mainline intercity, high speed lines and other). This would require the classification of Network Rail assets and, to the extent that it is specified in access contracts, the types of track that are reserved under those contracts.
- Efficient headroom. If it is deemed appropriate to allow freight to retain spare capacity (i.e. headroom) without being penalised (due to freight's requirement for some degree of flexibility) then it would require one to calculate what the efficient level of headroom is for freight operators. This would likely differ across commodity types.
- Adjustments for other elements of quality. In theory a reservation charge could be calibrated to capture additional elements of path or access right quality. Some of the potential adjustments above, such as for peak-services and track type, capture characteristics associated with quality of access rights. This could be extended further depending on the extent to which additional quality elements are specified in track access contracts. For example, if an access contract specified a minimum time buffer between its services and

other services (to minimise the risk of delays) then this could be captured in an uplift to the reservation charge applied to that right. Generally speaking, the ability to adjust for quality will be greater if access contracts are more specified.

The more detailed information requirements discussed above are examples of how a simple reservation charge can be made to be more refined. In general, these additional elements would bring the reservation charge closer to a scarcity charge. In practice, basing the charge on access rights rather than the timetable may make some of the additional adjustments less feasible as access rights tend to be more flexible in nature.

Drivers This is would be a charge/obligation applied to all operators and would not replace any of the current charges. In practice those that will see an increase in charges will be those who hold the most spare capacity. While there may be several reasons that operators hold spare capacity such as inefficient planning or strategic holdings of rights to limit the capacity of new entrants, one key driver of spare capacity is the requirement for some operators to have flexibility in their services. This is typically understood to be bulk freight carriers as the commodities they transport may be seasonal in demand or subject to short notice periods for loading from sea terminals (e.g. some operators may be informed only the day prior to ships arriving in port).

> If additional refinements are captured in the charge, such as time-of-day or tracktype adjustments, then it may be possible to subject some more specified access rights (i.e. level 1 rights) to additional adjustments to the extent that their contracts capture relevant information.

Calculation While there are varying degrees of complexity of reservations charges, the key principles principles behind their calculation are quite simple – operators pay for capacity they have booked but not used.

As discussed above, there are several possibilities of units when specifying the charge, as well as a choice between applying the charge to access rights or timetabled train paths. The key choices are elaborated below.

Choice of units. The use of capacity on the network can be considered to have two dimensions, both in terms of time spent on the network and the length of network used. Therefore the natural choice of metrics would be either train km, train service minutes, or a combination of the two. HS1 passenger services are charged based on train minutes, while freight services are charged based on train km. While a similar split could be made for the network as whole, the application of a time element may carry significant complexities versus the HS1 example due to the relative complexity of Network Rail's network. It would, for example require adjustments to be made for stopping at stations so that stopping services are not unduly prejudiced. Furthermore, operators do not necessarily have control over timing of their trains as Network Rail requires a degree of flexibility timetabling purposes. This issue is even more relevant if applied to access rights. By contrast, train km are relatively simple to apply to both the timetable and access rights. Therefore, basing the charge on train km is likely to be the most feasible option.

Access rights vs booked (timetabled) paths. In its 2006 Charges Workshop ORR

consider whether to use timetabled paths or access rights as the basis for a reservation charge with the initial preference being for access rights.⁷⁷ There are pros and cons for both access rights and timetabled paths.

Using timetabled paths has precedent internationally and is relatively straightforward to apply by routes as they are clearly defined. However, the timetable includes level three rights/ spot bids for capacity which are inherently more flexible as there is no guarantee of access to the network.⁷⁸ Charging for capacity obtained in this way may be unfair as those routes are not retained as spare capacity, though it may be possible to adjust for these types of access rights. Network Rail also needs to retain some flexibility over routing and timing in developing the timetable and therefore operators may not have control over the length of their route (and therefore their per km reservation charge). In either case operators should not pay the reservation charge if they are unable to use their path due to Network Rail possessions.

Using access rights as the basis for charging is relatively straightforward as well, but has an extra level of detail in that you must define what is meant by 'used' (discussed below). With access rights it is possible to differentiate immediately based on the flexibility of rights as this is clearly set out in access rights contracts. Partial usage of firm access rights (where perhaps half the services are run) would therefore still be subject to the reservation charge. Access Rights for which an Access Proposal was made to Network Rail in the timetabling process, but Network Rail was unable to accommodate in the working timetable should be adjusted for (i.e. not charged).

We consider the most appropriate approach is the use of access rights over timetabled paths as using access rights specifically targets the legal right to track access. It would also not impinge on the flexibility required for Network Rail in creating the timetable, nor on operators' desire for more certainty over routing/timing of services and charges.

Definition of 'used'. We propose basing the reservation charge on usage of firm access rights. There are two possible ways we could define this:

- a) exercising firm access rights by booking timetabled paths; or
- b) exercising the firm access right and actually running the service.

Under option a) operators would have an incentive to exercise their rights even if they were intending on running the actual services, which would exacerbate any potential capacity constraints that exist. Option b) is more directly related to actual usage and therefore preferable as it targets translation of access rights into actual services. We therefore use option b).

Calculation of unit charge. The charge per-km itself does not have to represent costs directly incurred, as reserving capacity does not directly incur costs on Network Rail. However, precedent appears to point to using variable charges as a basis for the per-km reservation charge. One could use current VUC rates and a notional equipment type (and tonnage for freight) to calculate per-km charges.

⁷⁷ ORR (2006), Structure of Charges Workshop.

⁷⁸ FOCs have the ability to bid year round through a rolling spot bid process. Therefore some schedules in the timetable are not attached to specific access rights contracts. However, Network Rail has told us that most FOCs will apply for access rights for most spot bid traffic once they have a Working Timetable path.

While one could envisage a uniform charge for all operators, our assessment has assumed a separate charge for passenger operators and for freight. For simplicity our modelling has assumed a charge equal to 25% of the VUC divided by the number of train km (separately for passenger and freight services) as a central case.

Calculation of distance. In terms of the train km value used for each right, level one and two access rights would define routing characteristics to different degrees. It may be possible in some cases to specify total route km quite well (e.g. 'Contract miles' are specified in column 26 of Schedule 5 of freight track access contracts). However, in some cases, especially where routing is more flexible, an average or minimum distance may need to be used. For example, where access rights specify only origin and destination one could use the shortest likely route possible.

Calculation of total charge. We proposed a simple version of the charge, on a per train km basis applied to unused firm access rights. We do not suggest the addition of any time-of-day or track-type adjustments, but in theory these could be applied simply as scalars.⁷⁹

The calculation is therefore quite simple. For a given unused firm access right i and associated number of kilometres, operator j pays:

$RC_{i,j} = Unused_firm_right_km_{i,j} * Charge_per_km$

For a single operator j with n_i unused access rights, the total charge over a given period is:

$$RC_j = \sum_{i=1}^{n_i} RC_{i,j}$$

Practical We describe here some of the practical considerations and, where possible outlines of potential solutions.

• Adjustments for possessions/ cordon caps. Paths that are seized by Network Rail will have to be adjusted for as this implies an effective revocation of rights if services are unable to be rerouted. The adjustment becomes less clear with quantum rights. A simple example serves to demonstrate the difficulty with quantum rights: say a freight operator has rights for 10 trains per week and maintenance works mean that the train is unable to operate on 4/7 days in a week. The operator then runs 3/10 of their allotted rights. How many of the unused 7 rights are should be adjusted for given the quantum right covers the whole week?

Adjustments for cordon caps will also have to be made but should be more straight forward as Network Rail specifies the total number of paths it is able to allocate.

- Partial usage. Adjusting for partial usage of capacity is a practical issue regardless of whether rights or timetables are used. Operators should only pay for the portion of rights that go unused. This would likely require some dispute mechanism and potentially negotiation with companies.
- Flexible rights. Some operators retain a large degree of flexibility in their contracts. Y-paths are one example where operators in practice have the right

⁷⁹ That is, by a predetermined value as in the French system.

to use one route up to a certain point, then have the choice between two routes. Q-paths are another type of flexible contract that allow operators a certain quantum of services over a time period. These types of access rights may present some difficulties in determining the appropriate level of charge.

- Updating Network Rail billing system. An ex post billing system⁸⁰ would need to be developed and implemented. International examples may help in designing the system but it may be costly to update the current system. However, it is likely that Network Rail already collects a large amount of the information required for the calculation of the charge.
- Efficient headroom allowance. Some types of customers may have a legitimate claim to holding excess rights. In particular, the efficient holding of rights for bulk freight operators is likely to be higher than what is actually used due to potential seasonality of commodities, short notice at ports, variable demand (e.g. construction materials may be weather-dependent) or the terms of end-user contracts.⁸¹ To some extent, it may be possible to incorporate seasonality into track access contracts themselves, but even so there may be a requirement for further flexibility in track access. Therefore, it may be appropriate to allow commodities some level of 'headroom' (holding of rights in excess of expected usage). The efficient level of headroom would in practice be difficult to determine and the incentives implied by this would need to be considered thoroughly. It may be possible to establish a benchmark level of headroom for bulk freight that has penalties/rewards attached to it.

Also, some flexibility may be socially desirable if, for example, it means that there is more security for freight customers that provide critical services. For example, contracts for supplying Drax coal power station (which supplies circa 7% of GB electricity) may need to be flexible to accommodate variations in demand.

- Additional charge adjustments. As mentioned previously the ability to scale the charge for additional factors such as time-of-day and track type may be limited with flexible rights. However, to the extent that rights a more prescriptive it may be possible, and justifiable, to charge a higher reservation charge. Another approach would be to develop a method of varying the reservation charge with the level of certainty embodied in track access contracts. In this case more prescriptive contracts would be more expensive. This would create an incentive for operators to consider trade-offs of cost and certainty while providing more options to holders of firm rights than a uniform charge would.
- Industry revenue neutrality vs freight revenue neutrality. The offsetting of this additional charge (such that Network Rail required revenue is met) can be done at either an industry level or at a more granular level, e.g. by operator, type of operator or type of commodity. To the extent that the charge may fall disproportionately on one group it may be a valid consideration whether this group should receive more benefit from the offsetting of other charges.

⁸⁰ That is, a system that calculates the charge after rights have been used of not.

⁸¹ Section 4.3 and 4.4 of freight track access applications is directly related to justifying the holding of any access rights that may be in excess of expected usage. However, it is conceivable that operators may still apply for more than an efficient level of rights.

- Rebates for paths used by others. The HS1 scheme includes a rebate of 75% of the reservation charge if the reserved path is eventually used by another operator. A rebate scheme should be considered for the charge discussed in this assessment as it would potentially soften the impact on some operators. However, basing the charge on access rights may complicate this calculation and allocation of rebates, especially when considering rights that are less specific such as quantum rights.
- Short term planning of paths. Operators which need flexibility but are unwilling to pay reservation charges (and therefore not hold firm rights) may be encouraged to plan paths on a more short term basis. From a purely operational sense freight operators may be able to manage more short-term booking of paths as they currently deal with a large amount of short-term train retiming (e.g. we understand that 40-50% of freight trains are retimed/re-rerouted, sometimes at very short notice). However, short-term booking of paths could add to the marginal cost of running a service due to having to retain additional staff to manage paths, contracting drivers at shorter notice and adding risk that paths will be unavailable. In some cases short term planning may not be feasible as some freight operators need to book paths at least 18 weeks in advance in order to avoid paths being taken for maintenance or ensure that they are not left with low quality paths. These result of these impacts could be a significant risk to performance.
- Lead time There are a number of factors that would have to be consulted upon before the implementation of a reservation charge. NERA summarised some of these:⁸²
 - deciding on the overall approach to be adopted;
 - deciding on any adjustments to be made for the charge (e.g. headroom, possessions, cordon caps, ...);
 - deciding the level of charge;
 - designing the rebate scheme (should one be adopted); and
 - defining contractual and legal implementation.

In practice it would also be pragmatic to first develop an indicator of allocated capacity usage and monitor this on an ongoing basis for some time before implementing any charge based on the metric. Simultaneously the billing system could be refitted.

As much of the consultation is likely to be controversial and sufficient time is required for calibrating a metric and refitting the billing system a rough estimate of lead time could be 3-5 years plus any years remaining before the start of the subsequent price control. NERA used a lead time for the scheme envisioned in their paper of 5 years.⁸³

Resources NERA estimated the setup costs of the billing system for their version of required for implementation spread over 5 years. The estimated ongoing costs of this approach were estimated to be £95,000 per annum and included costs for updating databases/ ongoing rights reviews and dispute resolution.

⁸² NERA (2007), The Impact of a Reservation Charge, p.31.

⁸³ NERA (2007), The Impact of a Reservation Charge, p33.

Option 7: Path reservation charge It was also estimated that freight operators would need to hire an extra full time staff for reviewing and monitoring charges and rebates, costing circa £25,000 per annum. Performance against criteria Axioms Current Dynamic On-rail Specified Protect Beneficiary Capacity Regional franchises railway comp freight pays allocation powers System safety = = = = = = There is no clear/direct impact of this charging approach on system safety. It is likely to lead to a reallocation of capacity with the result that this option will not materially change the status quo. Consistency = = = = = = = = with law Regulation 15(1) of the Railways Infrastructure (Access and Management) Regulations 2005 provides for a capacity reservation charge: 15.—(1) The infrastructure manager may levy an appropriate charge for capacity that is requested but not used, and the imposition of this charge must provide incentives for efficient use of capacity. Funding of + = = + Network Rail If funds raised from the charge were regarded as a contribution to enhancement efficient costs costs, then there may be some positive impact on this criterion but this would depend on the value attached to the charge. When these were considered previously ORR suggested that the charge would be low. However, there is also a potential benefit in reallocation of/ optimal use of existing capacity with consequential impacts on enhancement costs. In SoW where there is more competition for capacity, this option would have greater impact, although other capacity-based options might be even more beneficial. Where there is less competition/greater protection the option would deliver little if any added value Allowance for = = = = = = = = market Most operators require some degree of flexibility to manage seasonal demand or new conditions requirements as they arise. For example current freight access rights make allowance for this through differing categories of right which range from fixed rights through to options to use additional spare capacity. Reservation charges raise the issue of how this required flexibility in traffic levels would be managed. A single = = = = = = approach for This approach, as per the current planning led approach, could be applied to all the network operators irrespective of type. This is the case on HS1 and in France. as a whole Current Dvnamic On-rail Specified Protect Beneficiary Outputs Capacity Regional railway comp franchises freight pays allocation powers Service costs = = = = recovery This option has no direct impact on cost recovery.

Option 7: Pat	h reservatio	n charge							
Efficient	+	+	+	+	+	+	+	+	
whole-system whole -life industry net costs	This approa additional of approach cu total freight This process the timetab Given a sig recently it is freeing of Nonetheless rights beyon recent relea of rights in occurred in efficiently w The positive	ch could be capacity. T urrently in p paths). Of s is current le are eithe gnificant pr s unclear wh useful cap s, this exan nd that wh se of rights response t the past 10 yould bolste e effects of	e beneficial, he freight place, relea these, only ly continuin er currently roportion of hether a re- pacity that nple also s nich is nece s may be po- to falling fr years. Intre- er the existi	at least at industry sed approx 22% were ng and the under rev of total free servation cl would o hows that essary unde cossible due eight miles oducing a fi ng provisio n might be	the margin has recen- kimately 1, deemed to remaining iew or dee- eight access harge wou- utweigh to operators er the curre to operators (due to re inancial incon- ns under P more sign	n, in reducin tly, under 800 access o be useful (1,600 unu: emed to have ss rights have ld be able to rent regime ors not upd cunning long centive to m art J of the hificant in S	ng the nee the admi rights (circ for other p sed freight ve strategi ave been o achieve a of implem accumula e. For exar lating their ger trains) nanage acc Network C SoW where	d to fund nistrative ca 20% of purposes. t paths in c value. ⁸⁴ returned additional nentation. te access mple, the r holdings that has code. e there is	
Efficient long	greater com	petition.	_	_	_	_		_	
run investment decisions	This approach could be beneficial, at least at the margins, in identifying areas where capacity is truly constrained and ensuring optimal use of the assets before enhancement is considered. As discussed above, recent trends in releasing of access rights by the freight industry suggest that a large proportion of rights that were given up were not 'useful' (approx. 78%) for other purposes. Freight lines are also likely to be less congested in the first place. Therefore impacts under this criterion may be marginal. In a SoW that places greater overall emphasis on efficient use of capacity, this option								
Efficient	=	=	=	=	=	=	=	=	
performance management	This option	has no dire	ct impact o	n the occur	rence of d	isruption.			
Efficient use of network capacity	+ A reservation capacity recommended perhaps insu- The freight place, release these, only currently co- are either significant punclear when useful capacity	+ on charge juired to op industry h sed approx 22% were ontinuing a currently proportion ether a rese city that wo	+ provides a perate servi lynamic to nas recentl imately 1,8 e deemed nd the rem under revi of total fre ervation ch puld outwei	+ financial in ces rather f reflect ongo y, under th 00 access r to be use haining 1,60 ew or dee ight access arge would gh the cost	= ncentive to than rely o bing chang he adminis ights (circa ful for ot ful for ot con unused emed to he rights hav be able t s of impler	+ o more car on pre-existi es to deman strative app a 20% of tot her purpos freight pat nave strate ve been ret o achieve a mentation.	+ ng rights, y nd pattern proach cur cal freight p es. This p hs in the gic value. curned rec additional f	+ sider the which are s. rrently in paths). Of process is timetable Given a ently it is freeing of	

While this may suggest that the existing administrative approach is capable of

⁸⁴ Source: discussions with freight industry representatives and Network Rail.

managing access rights, it also shows that operators are able to accumulate access rights beyond that which is necessary under the current regime. For example, the recent release of rights may be possible due to operators not updating their holdings of rights in response to falling freight miles (due to running longer trains) that has occurred in the past 10 years. It is difficult to predict to what extent a reservation charge would lead to a release of useful rights, but it would nonetheless provide a financial incentive for the efficient management of access rights on an ongoing basis. Therefore, it may be more capable than the current administrative regime, which is reactive in nature, at managing capacity.

We have not undertaken additional detailed analysis on the opportunity cost of unused or lightly used access rights and therefore cannot predict the extent to which a reservation charge would free up rights. However, in the example above, while the majority of access rights returned were not useful it is clear that there are some useful rights that are retained in excess of what is needed by operators. Our modelling has not attempted to capture a proportion of unused rights that may be returned. We have calculated that the impact on bulk freight is likely to be the greatest and may be significant enough to incentivize bulk operators to relinquish some additional rights.

The effect of this option might be greater in those SoWs where there is greater competition. The impacts in the 'specified franchises' SoW would be similar to the current SoW as most impacts are felt by freight. In 'Protect freight' the impact of the charge would be diluted by the offsetting effects of additional freight protections.

Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
Predictability	-	-	-	-	-	-	-	-		
	May adversely affect predictability of charges, e.g. for freight which are inherently subject to some degree of volatility e.g. seasonal demand for coal.									
Simplicity	=	=	=	=	=	=	+	=		
	While this is to suggest which app simplicity in	is the most that it wo ear to be na SoW tha	straightfo uld be sin well und t places g	orward of the mpler to ad erstood. Thi reater emph	e scarcity minister f s option asis on th	related opt than the cu would hav e value of ca	ions there rrent arrar e the adva apacity.	is nothing 1gements, antage of		
Transparency	+	+	+	+	+	+	+	+		
	Current ac necessarily would brin optimal use	cess arran publicly av g greater to e of the net	gements ailable or ransparen work.	are establis easy to revie icy to use of	hed in b w. The in capacity	ilateral con nposition of and may re	tracts that a reservati esult in ove	t are not on charge erall more		
Low	-	-	-	-	-	-	-	-		
transaction costs	Imposition manageme not curren ability of t system. Off capacity the costs.	of reserv nt (both by tly exist an he current fsetting the at could be	ation ch Network d concer billing so costs of used more	arges would Rail and by p ns have bee ystem to im this would b re effectively	d imply bassenger en expres plement be the ber r and whic	a degree and freight sed by Net the option nefit of ider ch might off	of monito operators) work Rail a of a depo ntifying und set some ir	oring and that does about the osit based lerutilised ovestment		

Option 7: Pat	h reservatio	n charge							
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Network Rail	=	=	=	=	=	=	=	=	
accountability	Reservation charges would in principle place greater emphasis on Network Rail's management of capacity and its ability to obtain optimal utilisation of the network. Arguably existing efficiency incentives e.g. the efficiency assumed by ORR in its determination of costs, have a similar effect currently in incentivising Network Rail to make best use of existing assets before investing further.								
Non-arbitrary	=	=	=	=	=	=	=	=	
allocation of costs	This charge users, it is p that they d operators f capacity to allocation o	is not relation ourely an in lo not use. ail to show users, whe of costs.	ted to the centive m However , and boo ther they	e opportunity echanism to to the extended oked capacity use it or not	y cost of a discouragent that N y impinge t, it could	access, from ge operators letwork Rail is on the co not be desc	the point of s from bool loses inco sts of prov cribed as ar	of view of king paths me when iding that arbitrary	
Optimal traffic growth	=	+	+	=	=	=	=	=	
	A reservations scale of the flexibility and charge. The option could free of	on charge s e benefits nd given th could be m capacity for	should con may not e lower in nore position use by ot	ntribute to a be great or ncentive effe ive in SoW w her operator	all the obverall given verall given ect of a development where the rs.	jectives of t en the nee eposit based re is greater	this criterio d to maint d system ov competitio	n but the ain some ver a new on since it	
Aligning	+	+	+	+	+	+	+	+	
Aligning industry incentives	This option recognises that industry incentives may not be aligned and places weight on industry participants giving up unused access. Current planning based approaches seek to achieve the same thing but the imposition of a pre-payment mechanism creates a common basis for discussions and should improve transparency.								
Value for	+	+	+	+	+	+	+	+	
money for funders, taxpayers and users	A reservation charge should contribute to all the objectives of this criterion but the scale of impact may not be great overall given the need to maintain some flexibility for freight operators and given the lower incentive effect of a deposit based system over a new charge.								
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	

In the current SoW, reservation charges could have some positive benefits in incentivising better use of existing capacity but in some forms could significantly adversely impact freight if adverse impacts are not addressed (e.g. through grants or other protections). Impacts on franchised passenger operators under the current regime are likely to be small given that the current arrangements provide protection from change and limit the scope to change service levels. This form of scarcity management has some precedent given that it is in use on HS1, in France and Germany and is being considered for Crossrail.

=

+

+

+

+

+

+

Assuming adverse impacts could be managed such that freight retains the flexibility required, the overall benefit (in terms of more efficient allocation of capacity and

+

management of allocated capacity) could be positive but may be small. More detailed analysis would be required to assess whether likely benefits outweigh issues such the transaction costs involved in introducing and managing a reservations system.

In alternative SoWs that introduce greater on rail competition or which place greater emphasis on the value of capacity this option could have greater positive impact (although it would need to be weighed against other options for value based charging). In SoW where operators have increased protection from change, the option would have less impact.

There may be scope to augment the administrative mechanisms that govern track access and the release of unused rights. For example, better monitoring of access right utilisation could bolster the effectiveness of rights review meetings as well as help in administering the use-it-or-lose-it provisions in Part J of the Network Code. Indeed, the use of Network Rail's new F-CDM system appears to have allowed several paths to be removed completely from the timetable. However, a purely administrative regime would still lack a financial incentive.

Impact on stakeholders

Overall, we expect the impact on TOCs to be relatively small. FOCs will feel a relatively greater impact but it is still expected to be relatively moderate. Freight users may bear some additional costs to the extent that operators will be able to pass on costs of demand uncertainty/ seasonality (which are a source of low utilisation of allocated capacity). Basing the charge on access rights should not impinge on the flexibility Network Rail needs for the construction of the timetable and should not impact Network Rail's ability to recover costs.

Franchised operators typically have fairly rigid service level commitments (SLCs) Franchised commuter embodied in their franchise contracts and therefore have little room to adjust their passenger behaviour in response to a reservation charge. However, this makes little difference as the reservation charge would not be levied due to the franchisees requirement to operators fulfil SLCs. Evidence suggests that passenger operators are typified by relatively high levels of allocated capacity utilisation. Therefore, reservation charges would have little impact on franchised operators.⁸⁵ If the additional income from the reservation charge is offset against the FTAC (e.g. through the rebalancing of charges assuming Network Rail's revenue requirement is unchanged), franchise operators may in fact pay slightly less in charges (though this is likely to be marginal). However, the financial impact will be nil in existing franchise agreements as franchised operators are held neutral to changes in charges. The range of sensitivities of allocated capacity utilisation rates included in our model is consequently geared towards the lower end of non-usage. The financial impact is consequently very small and potentially offset by changes in other charges.⁸⁶

Franchised Same as above. regional passenger operators

⁸⁵ e.g. NERA (2007), The Impact of a Reservation Charge.

⁸⁶ The charge is assumed to be revenue neutral over the industry as there is no assumed changed in Network Rail's revenue requirement. It is not assumed to be revenue neutral for freight operators.

Option 7: Path	reservation	charge
-----------------------	-------------	--------

Franchised inter-city passenger operators	Same as above.					
Open access passenger operators	Evidence suggests t allocated capacity u on open access ope	hat passenger operators are typified by relatively h tilisation. Therefore, reservation charges would have rators. ⁸⁷	igh levels of little impact			
	Open access operat a result of the reser open access will be they are not bound	ors are likely to see a relatively small negative financ vation charge. However, the main difference to franc able to better respond to the charge if it does have a by the same types of service commitments.	ial impact as chises is that an impact as			
Freight Evidence from the 2007 Freight Route Utilisation Strategy suggests that inter operators freight services, which comprise a large proportion of multi-customer freight services, have high levels of allocated capacity utilisation at around 95%. ⁸⁸ More evidence from discussions with Network Rail suggest that this trend is largely today, though intermodal services to the London Gateway port may have seen level of utilisation recently.						
	Based on the relatively high level of allocated capacity utilisation the negative financial impact is likely to be quite small for these services. However, even small changes to costs in this sector have the potential to have large impacts because demand for these services may be relatively price elastic due to the ability to switch to road. Therefore intermodal shift (to road) may be a concern despite the small					
	Bulk freight services are likely to bear the greatest negative financial impact in the sector. This is because bulk freight services have much lower rates of alloc capacity utilisation than other types of operators. For example, the table below sh utilisation rates of booked paths (not access rights). ⁸⁹					
	Table 3.9: Path take-	up by key commodities 2004/05				
	Commodity Take-up					
	Intermodal	95%				
	Petroleum	56%				
	Metals	51%				
	Coal	45%				
	Construction	37%				
	Channel Tunnel	21%				

Source: Network Rail Freight RUS (2007)

One of the reasons for this spare capacity is the requirement for bulk freight to retain flexibility due to seasonal/ variable demand. While the table above suggests quite a

 ⁸⁷ e.g. NERA (2007), The Impact of a Reservation Charge.
 ⁸⁸ Network Rail (2007), Freight Utilisation Strategy, p26.
 ⁸⁹ The Network Rail Freight RUS.

Option 7: Path reservation charge bit of non-utilisation by some commodities, there has in recent years been significant relinquishing of unused rights and timetabled paths by the industry (through the provisions of Part J of the Network Code). Both Network Rail and representatives from FOCs have informed us that since 2014 over 1,800 train paths have been removed from the timetable. Currently, approximately 22% of freight schedules (across all FOCs) in the timetable are unused, capturing both schedules that are yet to be reviewed and those deemed to have strategic value. Our analysis, which has been based on current utilisation rates and informed by the table above, suggests moderate impacts on bulk freight charges.⁹⁰ The impact is therefore smaller that it would have been prior to the recent release of rights and schedules. Network Rail Network Rail would need to update its billing system to accommodate the charge. The costs of doing so would need to be weighed against the potential value of freeing up access rights. The reservation charge should not impact Network Rail's ability to recover costs. The reservation charge would complement Part J of the Network Code that provides the administrative mechanism for reclaiming access rights. The reservation charge may encourage more flexible track access contracts and potentially more spot bids for access. When constructing the timetable a small amount scheduling flexibility can be very useful and potentially valuable. However, more last minute planning of train paths by operators may also make it more difficult for NR to ensure that it retains enough space in the timetable for operators to bid into to meet operators' need for capacity. However, the practical consequences for timetable construction may warrant further consideration. Funders A reservation charge should not impact DfT or Transport Scotland directly, though may play a role in future franchise bids as operators build any expected charges into their bids. Since franchises are likely to have a high rate of allocated capacity utilisation (and thus incur minimal reservation charges) there is likely to be little cost passed through to funders due to the protections in franchise agreements. This of course may change depending on the State of the World. Passengers Passengers are unlikely to see a material financial impact as passenger services and freight typically have higher levels of allocated capacity utilisation and therefore will incur users fewer charges. Furthermore, in the current SoW a large proportion of passenger fares are regulated and exposure to any increase in costs to passenger operators would be reduced. They may however benefit if useful paths are freed up and additional services (which were previously not able to gain access to the network) are allowed to run. Freight customers may suffer from potentially less flexible FOCs if access rights are returned. However, if other FOCs are able to fill other timetabled paths then there is likely to be little change in the flexibility of services overall. Freight customers may also suffer from increased costs of services as FOCs may try to pass a portion of charges incurred to customers. Small increases in costs may cause some customers

⁹⁰ Our modelling assumes that utilisation rates of Access Rights are broadly similar to utilisation rates for timetabled paths. This will be inaccurate to an extent as the timetable includes spot bids and contingent rights, and it does not capture any rights that are held but not exercised. However, at the time of analysis information on utilisation rates of access rights was unavailable.

(in particular intermodal customers) to switch some services to road.

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was consensus in the industry and Government that a path reservation charge would almost exclusively 'bite' upon freight operators, raising a wide range of concerns that it could be impractical given the nature of the market within which they operate.
- Freight operators require some flexibility in their access due to variability in demand. However, freight operators noted that passenger operators have paths in expectation of demand, whereas freight operators react to demand, which is why freight operators have a lower utilisation rate.
- Reservation charges in other countries (e.g. France) are thought by freight operators to have led to a significant decrease in rail freight volumes and reduction in quality of service.
- Work on the holding and management of freight capacity has been ongoing since 2007. This has contributed to a significant release of access rights by the rail freight sector (information from Network Rail and freight indicate that approximately 20% of access rights have been released since 2014).
- A charge could encourage short-term booking of paths, which would add to operational costs. Short term planning may also pose a significant risk to performance
- The charge would have to be designed to ensure that:
 - o customers and operators are not punished for unmanageable variation in demand;
 - o there is not incentive for operators to run shorter trains; and
 - small operators and new entrants are not penalised.
- Freight operators noted that with more freight operators in the market there are likely to be more paths required but not used.
- Passenger operators did not expect that this option would have any material impact, as the real issue is capacity allocation and timetable development. Unless freight operators were charged for the paths "reserved" for them in the timetabling process, nothing would actually be changed by the introduction of the charge.
- Freight operators suggested that transaction costs are likely to be far too high for the level of benefit that could be realised.
- Network Rail commented that as the network becomes more 'full' it will become increasingly
 important that all mechanisms that could improve utilisation are considered. However, there has
 already been a significant amount of work undertaken by Network Rail and freight operators to
 determine paths that could be 'given back' for other uses. Network Rail would however be
 concerned if the charge led to freight operators making greater use of short term planning
 resources to gain access to the network, in seeking to avoid a reservation charge.

ANNEX E RECALCULATE BENCHMARKS FOR TRAFFIC CHANGES AND REMOVE CAPACITY CHARGE DETAILED ASSESSMENT

Option 15: Recalculate benchmarks for traffic changes and remove capacity charge										
Overall performance against the RDG Vision in each SoW										
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers			
+	+	+	+	+	+	+	+			

Schedule 8, the performance regime, is in place to provide train operators with compensation for disruptions to their services and therefore to their business. This option is envisaged to update Network Rail's Schedule 8 performance benchmark annually to take account of traffic growth (or decline), as an alternative to the current separate 'capacity charge' which aims to reimburse Network Rail for its traffic-related portion of Schedule 8 compensation payments. The capacity charge's current link to Schedule 8 is not obvious in its name: it is sometimes wrongly assumed to be similar to a 'scarcity charge'. The capacity charge is not understood well by the industry or considered to be cost reflective, and therefore rather than explore opportunities to make the charge more granular or adjust its calculation in any other way, this option envisages removing the charge entirely.

Network Rail's Schedule 8 benchmark is set at the Periodic Review preceding the control period. The Periodic Review also outlines updates to the benchmark each year of the control period to reflect the changes in Network Rail's performance targets. This option proposes that Network Rail's benchmark also includes an annual update to account for the actual increases or decreases in traffic (using the latest year of available data each time), which could allow for the removal of the capacity charge. The capacity charge is currently intended to recover the amount of additional compensation paid out by Network Rail through Schedule 8 due to increases in network traffic making it more difficult for Network Rail to recover services after a delay, and is levied per actual train mile at an ex ante tariff rate set at the Price Review.

There are a range of options for change to address the current issues with the capacity charge, and this option is envisioned with the following key characteristics:

- Annual updates under this option are intended to 'smooth' the effect of traffic growth, as the benchmark would be updated a little each year to take account of the level of traffic.
- The industry's understanding of the charge and its intentions could be improved by better integrating the recovery of Schedule 8 Congestion-Related Reactionary Delay (CRRD) costs into Schedule 8 itself.
- The method of updating the benchmark would ideally be simple and mechanistic, to ease understanding and calculation, therefore also minimising transaction costs.

There are some benefits to this option, mainly ensuring that the Schedule 8 benchmark is more reflective of the appropriate achievable level of performance and the likely improvement in industry understanding that could result. However, there will also be costs involved in the initial calculation of the appropriate adjustment to the benchmark, which would likely require a level of resources comparable to that required to recalibrate the capacity charge at CP5. Subsequent annual updates could be devised as simple mechanical calculations, reducing the complexity and cost of implementation.

This change would reduce the variable charges that operators pay, and hence Network Rail's income from track charges when new capacity is put into operation. This also reduces the marginal cost of running an additional train. The capacity charge reflects, when levied on additional trains (introduced after the start of a Control Period), the marginal performance costs of running additional

services. The charge does not reflect this perfectly as it does not adjust for time periods such as peak and off-peak, only for weekend vs weekday services, so the rate is averaged within those two. It also has a single flat rate across the network for freight and charter operators so does not reflect geographic differences in the marginal performance cost of running additional services. Nonetheless, the capacity charge at present, to some extent, discourages additional services running where it would worsen congestion, even though this is not its primary purpose. Removing the capacity charge might result in some additional trains that might not otherwise have run, which could be considered a less efficient outcome in terms of use of network capacity (the cost to other running trains of the congestion that additional train running will cause may be higher than the benefits the additional train itself will bring).

Without other changes to the regime, we expect that revenue previously recovered through the capacity charge, which is paid by all operators, would instead be recovered through the FTAC, paid only by franchised passenger operators. This would lead to a redistribution of charging income from open access and freight operators to franchised passenger operators, and also some redistribution between franchised operators. This is because the capacity charge is offset against the FTAC at present. This redistribution could also affect the allocation of cost across administrative boundaries, due to the current arrangement of cross-border services between England and Scotland.⁹¹ However, the redistribution across boundaries is likely to be relatively small.

We note that there are options, such as a scarcity charge, which could be implemented alongside this change to the regime to reverse some of the redistribution effects, and also to provide more explicit and targeted incentives around use of capacity than the capacity charge does at present.

Overall, we have rated the impact of this option as slightly positive in many SoWs, as we don't expect impacts to differ materially between SoWs.

Key characteristics

Description of option

NR's Schedule 8 benchmark for each five-year control period is set at the preceding Periodic Review (currently the industry is under CP5, running from 2014-19, which was set at PR13). The benchmark is adjusted each year to take account of the annual changes in performance targets that were set during PR13, without any further updates. This option proposes that NR's benchmark also includes an annual update to account for actual increases or decreases in traffic, which could allow for the removal of the capacity charge.

The capacity charge is intended to recover the amount of additional compensation paid out by NR through Schedule 8 due to network traffic making it more difficult for NR to recover services after a delay, and is levied per actual train km at a tariff rate set at the Price Review. The capacity charge is currently considered an issue within the charging and incentive regime, as its intention is not clearly understood by industry members and there is a broad consensus among participants that it is not fit for the purpose.

There are a range of options for change to address the current issues with the capacity charge, and this option is envisioned with the following key characteristics:

• Annual updates under this option are intended to 'smooth' the effect of traffic growth, as the benchmark would be updated a little each year to take account of the traffic. This option envisages that each year's benchmark would use the traffic data for the previous year. Therefore, there will be a small lag between actual traffic growth (or decline) and it being reflected in the

⁹¹ This arrangement means that English franchises pay the capacity charge but not the FTAC for use of the Scottish areas of the network, and vice versa. However Transport Scotland state that this is unbalanced against Scotland.

benchmark. However, this lag is likely to be small compared to, for example, updating only at each price control and therefore accounting for five years of traffic change in a single update, which might cause a larger lag which would be undesirable.

- The industry's understanding of the charge and its intentions could be improved by better integrating the recovery of Schedule 8 CRRD costs into Schedule 8 itself. While this option as envisaged may remove the 'incentive' (marginal cost) side-effect of the current capacity charge, an incentive to not use constrained areas of the network (where the capacity charge is likely to be highest) could be introduced through a separate scarcity charge or avoidable costs approach.
- The method of updating the benchmark would ideally be simple and mechanistic, to ease understanding and calculation, therefore also minimising transaction costs. To determine the appropriate increase to the benchmark, the minutes of delay caused by additional traffic on the network (congestion-related reactionary delay, CRRD) would need to be converted into Average Minutes of Lateness (AML)⁹², the measure that the benchmarks are currently set in.⁹³ A mechanistic approach could be similar to the "congestion factor" which updated the freight benchmark for traffic in CP4 (once traffic growth or decline reached a threshold) and CP5 (regardless of the magnitude of traffic change).⁹⁴

The traffic updates to the benchmark would be additional to the current updates which account for increases in performance targets, and therefore the upwards 'trajectory' of NR's expected performance level would increase slower if there is traffic growth, and faster with traffic decline.

This option does not envisage a wash-up to account for differences between the level of traffic used to calculate the benchmark and the actual traffic that year. Depending on the speed of traffic growth (or decline), the benchmark adjustment could cause some over- or under-estimate of the appropriate benchmark adjustment. An incorrect benchmark could mean NR pays out too little or too much compensation. With an annual update to the benchmark any lag should not be significant and therefore our analysis of this option does not assume a wash-up.

This approach differs from the proposal put forward by the Rail Freight Operators' Association (RFOA) for PR13.⁹⁵ RFOA proposed to retain the capacity charge close to its previous form but with a 'baseline' traffic level for freight below which no charge would be applicable. RFOA argued that the impact of implementing this for franchised passenger operators would be minimal since the reduction in capacity charge would feed through to an increase in their Fixed Track Access Charges and therefore even out.⁹⁶ This option does not apply a baseline level of traffic, it envisages that the CRRD from all traffic would be accounted for in the benchmark.

See RDG Phase 2b Report Features: 8.4, 8.5, 8.6, and 8.7.

⁹² AML is calculated by the product of the number of passengers expected to alight at main stations by the punctuality to the nearest minute at those stops. ⁹³ Source: Network Rail performance data, available <u>here</u>.

⁹⁴ The congestion factor is discussed in ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" pp. 784-785. The equation for the freight benchmark is as below

 $Updated benchmark = Original benchmark \times (1 + (traffic growth \times congestion factor))$ This factor increases the benchmark to account for the increased difficulty for a freight operator to meet its benchmark when overall traffic on the network increases.

⁹⁵ RFOA (2013) "Freight Capacity Charge – proposal on methodology", available here.

⁹⁶ Using CP4 data, the split of FTAC across the 'indicative operators' is quite different to the split of the capacity charge, and so while it may be net for franchised operators as a whole, operators will experience a different net impact of such a policy depending on the proportion of commuter, regional, and inter-city trains they operate.

Description of counterfactual

There is a Schedule 8 benchmark for NR, set at a service group level. This benchmark determines the level of performance that NR is required to meet; if NR performs better than this benchmark, it will receive a bonus payment and if it performs worse than this benchmark it will be required to pay compensation. This benchmark is set for the control period (five years), and is then adjusted each year within the control period in line with NR's performance targets; NR is required to improve its level of performance each year.

NR's benchmark is not updated to account for the impact that traffic growth (or decline) will have on its ability to meet a certain level of performance. All train operators also have a benchmark; passenger operators' benchmarks are not updated for traffic, but freight operators' benchmarks are. In CP4 and CP5 the freight benchmark has had a 'congestion factor'; the benchmark is updated each year by applying the congestion factor to the level of traffic growth and multiplying this by the original (pre-traffic growth, or decline) benchmark.⁹⁷

The capacity charge is a pre-determined charge levied per train mile, to make NR financially neutral to the addition of more traffic to the network.⁹⁸ It recovers the estimated additional Schedule 8 costs from increasing traffic on the network, and is therefore heavily linked to Schedule 8 at present – if estimated Schedule 8 payments increase then the capacity charge must also increase. The payment rates are higher for areas of the network where delays are more likely to have an effect on the operator's long term revenue (as end-users are more discouraged by a higher likelihood of delay). The figure below outlines the approach to calculating the capacity charge taken at CP5. Some industry participants view the capacity charge as an incentive to use the network efficiently i.e. as a form of 'scarcity charge'. While this side effect is recognised by ORR⁹⁹, it should not be considered an 'objective' of the current capacity charge.





CUI: capacity utilisation index, which is calculated per CTS for each 3-hour slot in the day. **CRRD**: Congestionrelated reactionary delay. Higher levels of traffic on the network make it more difficult for NR to recover services after a delay, this additional delay is recorded as CRRD. **CTS**: Constant Traffic Section, a high degree of geographical granularity (there are over 3,000 CTSs).

⁹⁷ The congestion factor is discussed in ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" pp. 784-785. The equation for the freight benchmark is as below

Updated benchmark = Original benchmark \times (1 + (traffic growth \times congestion factor)) This factor increases the benchmark to account for the increased difficulty for a freight operator to meet its benchmark when overall traffic on the network increases.

⁹⁸ This is in contract to the volume incentive which aims to encourage NR to assist traffic growth.

⁹⁹ ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" p.590 (¶16.189).

Relevant factors impacting the form and/or the effectiveness of the option

Data availability, measurement, and billing: there are unlikely to be any issues with gathering the correct data, however there will be a reasonable amount of calculation to undertake and there may be a period of delay between a year finishing and its data being ready to calculate the following year's charges – there might therefore be a delay to some billing and therefore a time lag between traffic increases and that traffic being reflected in NR's Schedule 8 benchmark. This lag is likely to be between a few weeks and a couple of months. There is some concern that this option would be difficult and costly to implement, however a mechanical calculation such as the 'congestion factor' currently applied to freight could be effective without incurring prohibitive transaction costs; the option is not envisaged as an exact calculation of the impact but rather some which is correct 'on average'.

Implementation	
Information requirements	The information required to implement this option should not exceed that already available, which includes:
	 CRRD data or traffic data, depending on whether the calculation will

- CRRD data or traffic data, depending on whether the calculation will depend on the actual congestion-related delay or on the growth in traffic. These two are inherently linked however a decision will be required on whether the calculation should use the actual traffic growth or the actual delay attributed to traffic.
- The relationship between delay minutes and the Schedule 8 performance metrics. This is required to update a year's benchmark with the delay data from the previous year.
- Drivers This option would remove the capacity charge, which would in turn remove one of the variable components of NR's revenue. NR's total revenue requirement would not change under this option, therefore the reduction in variable revenue would be replaced by an increase in the FTAC. All operators pay the current capacity charge but only franchised passenger operators pay the FTAC, therefore this would represent some redistribution (see the Table below). However, the magnitude of the redistribution would be small. This redistribution impact could be limited further, for example through the introduction of a scarcity charge.

Table E.1: Indicative Operators' shares of the capacity charge and FTAC

Charge		Franchised commuter	Franchised regional	Franchised inter-city	Open access	Multi- customer freight	Bulk freight
		10 operators	9 operators	6 operators	2 operators	4 operators	4 operators
Capacity charge	Share per operator	4%	2%	6%	0.5%	0.1%	0.1%
	Share for operator type	44%	21%	33%	0.9%	0.3%	0.6%
FTAC	Share per operator	3%	5%	4%	-	-	-
	Share for operator type	27%	49%	24%	-	-	-

Source: CEPA analysis using charges data from ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19." These figures assume a Network Grant has been applied.

Option 15: Re	calculate b	enchmarks	for traffic	changes an	d remove	capacity ch	arge			
Calculation principles	Calculat understa the PR13	ing the in anding the 3 process id	ncrease t relationsh entified a	o Schedule ip between relationship	e 8 targ traffic and that can	geted minu d CRRD. The also be used	ites would work unde for PR18.	involve ertaken in		
	The mai increase	The main principle of the capacity charge is to neutralise the Schedule 8 impact of increases in traffic, which is the intention of this option.								
Practical consideration	This opt s system. envision more ex would n	ion is unlike It does no ed The imp opensive an ot require	ely to be r ot need a lementati d time-co	nore expens ny more da on of this o nsuming to	sive and t ata than ption nee implemer	ime-consum at present, d not neces nt than the	ning than th and the c sarily be sig current sys	e current alculation ;nificantly tem, as it		
Lead time	to determine the appropriate relationship between known traffic (or CRRD) and NR's performance would depend on the complexity desired. This option is intended to be a mechanistic and simple calculation, which is broadly correct on average rather than precise and accurate at all times, and therefore a reasonably low leve of complexity is required.							il analysis RRD) and intended າ average low level		
Resources required for implementatio	The main burden of introducing this option would fall on NR, as it requires more frequent use of TOC data and modelling. tation									
Performance	against crite	eria								
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
System safety	=	=	=	=	=	=	=	=		
	This option impact NR caused by have a sign	his option is unlikely to have a negative impact on system safety, as it does not apact NR's ability to recover its revenue requirement. This assumes that the lag aused by using the previous year's traffic for the present year's benchmark does not ave a significant effect on NR's revenue.								
Consistency	=	=	=	=	=	=	=	=		
with Iaw	We do not to the Scl implication Directive 2 mirrored in In particula penalties, o improving	We do not anticipate that the removal of the capacity charge and the annual update to the Schedule 8 benchmark to include traffic growth, would have any legal implications. This is not intended as legal advice. The most relevant EU legislation is Directive 2012/34/EU (recast of Directive 2001/14/EC), and in the UK this is largely mirrored in The Railways Infrastructure (Access and Management) Regulations 2005. In particular, it is necessary to have a performance scheme (which can include any of penalties, compensation, and bonuses) which is targeted at minimising disruption and improving the performance of the network in particular the regime must:								
	• be non-c	liscriminato	ry across t	he network	;					
	• encoura	ge optimal (use of the	network and	d its capac	city;				
	• provide s	sufficient in	centives a	nd price sign	hals to par	rticipants;				
	• reflect the There is n compensate to update for In addition	ne costs inco othing that ion must ta reight oper	urred in pr implies a ike a spec ators' ben	oviding the a factor upo ific form. In chmarks and under the B	service dating for CP5 a 'co nually for ailways A	traffic and ngestion fac traffic. ct 1993 to r	l congestion ctor' was in	n related troduced		

Option 15: Recalculate benchmarks for traffic changes and remove capacity charge									
	vary mode that are pa clauses set	I clauses fo art of all tra out the cha	or track ac ck access arges and	cess agreer agreements incentives.	ments. Mo s of similar	del clauses type. In pa	are standa articular, the	rd clauses ese model	
Funding of NR	=	-	-	=	=	=	=	=	
efficient costs	This option might affect the funding of NR's costs if the time lag in bench adjustments is large and there is no wash-up. Time lags might arise as benchmarks are set in advance based on an estimate of traffic and there would delay between changes in traffic and updates in the benchmarks. This is likely to more of an impact in the SoW where there is more dynamic rail or more of competition since greater fluctuations in traffic might be expected here. This is la an issue in the current system as the capacity charge is levied per actual train mil If the lag is considered to have a significant effect, then a wash-up migl appropriate. However, it is likely that with annual updates to the benchmark errors in traffic growth to small year to merit a wash-up.								
Allowance for	=	=	=	=	=	=	=	=	
market conditions	This option would result in freight no longer paying capacity charges, but also receiving less Schedule 8 compensation. The impact of this should be approximately neutral.								
A single	+	+	+	+	+	+	+	+	
approach for the network as a whole	Re-calculating NR's benchmarks for traffic changes annually, as is done for the train operators, makes Schedule 8 closer to being 'a single approach for the network as a whole' in its methodological approach to the calculations.								
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Service costs	-			-	-	-	-	-	
recovery	There will be a time lag in this option, as the benchmark would be updated each year to account for the <i>previous</i> year's traffic. This might not allow NR to fully recover its costs if traffic increases (unless there is a wash-up arrangement in place, which is not envisioned in this option).								
	of traffic growth. The impact of such a lag on NR revenues would depend on the level of traffic growth. The impact is expected to be negative in comparison to the current system where a charge is levied per actual train mile so a large increase (or decrease) in traffic is automatically accounted for. However, given that the benchmark (under this option) would be updated annually it is unlikely that the magnitude will be large. Therefore this option is graded weakly negative (-) in all SoW apart from 'dynamic railway' and 'on-rail competition' where traffic might fluctuate more and therefore a lag might have a greater impact on NR.								
Efficient	=	=	=	=	=	=	=	=	
whole-system whole -life industry net costs	This option should not greatly impact NR's incentives to enable changes to the pattern of service. It might have a potential small positive incentive on train operators to run additional services (if there is not a wash-up) through removing the capacity charge that is explicitly levied per mile travelled, however there might also be a disincentive for NR to allow and encourage growth (without a wash-up).								

Option 15: Re	ecalculate be	enchmarks	for traffic	changes an	d remove	capacity ch	arge			
Efficient long	=	=	=	=	=	=	=	=		
run investment decisions	This option should not negatively affect this criterion. The traffic update would not take into account improvements to NR's performance, but would rather ensure that the benchmark accurately reflects the level of performance expected of NR given the increase (or decrease) in traffic on the network.									
Efficient	=	=	=	=	=	+	+	=		
performance management	This option works effici that year b estimates). under all So 'on-rail cor franchises' However, a the impact neutral apa positive imp	should pro- iently. At the put only h Therefore, pW, with a npetition', where ther small red of this incourt from for	ovide NR ne start o as an est this optic particular but less re might b uction in entive, th SoW 'be	with an incr f a year, NR imate of ca on might im rly large imp of an impa e less flexibi efficient use herefore the neficiary pay	reased inc is aware pacity ch prove effi act in the ct in the lity. e of netwo scoring c ys' and 'ca	entive to m of its Sched arge revent cient perfor two SoW 'd SoW of m ork capacity of most SoW	anage its u lule 8 bencl mance man dynamic rai ore highly would like / for this c ation' whe	nplanned nmark for on traffic nagement lway' and 'specified ely reduce riterion is re a small		
Efficient use	_	_	_	_	_	_	_	_		
of network	- It is not clea	- ar that this	- ontion w	- Juld bave a l	- argo offor	- t on this crit	- terion	_		
capacity	Some indus efficient us replacing th This is beca lower. How strong as m heavily affe	stry participse of current ne capacity nuse the m rever, it has nany do no reted by thi	oants view ent netwo charge a arginal co s been rep t understa s option.	w the capaci ork capacity as a variable st to the op ported (RDG and the char	ty charge , an effe e compon erator of Phase 2B rge. Overa	as also bei ect that wo ent of train running an report) tha all, this crite	ng to incen buld be rea operators' extra train t the incent rion is unlil	tivise the duced by ' charges. would be tive is not kely to be		
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
Predictability	+	+	+	+	+	+	+	+		
	NR knows i charge reve costs more encourage variable. Th	NR knows its Schedule 8 benchmark each year but only has an estimate of capacity charge revenue based on traffic forecasts. Therefore, this option would make NR's net costs more predictable compared to the current regime, assuming that it does not encourage an increase in traffic volatility. This is because the capacity charge is variable. This criterion is graded weakly positive (+) for all SoW.								
Simplicity	+	+	+	+	+	+	+	+		
	This option is intended to reduce the complexity of the charging regime overall through removing the separate capacity charge. However, the calculation of the benchmark will become more complicated – although this option is envisaged as a mechanistic calculation which should not create significant complexity in the benchmark. Therefore this option has a weak (+) rather than strong (++) positive impact on this criterion. This would be the same in all SoW.									

Option 15: Re	calculate b	enchmarks	for traffic	c changes ar	nd remove	e capacity ch	narge			
Transparency	+	+	+	+	+	+	+	+		
	Absorbing a separate performan	the capacit e charge v ce regime i	y charge r will aid u n all SoW.	nore obviou Inderstandir	isly into So ng and tr	chedule 8 ra ransparency	ther than h of the S	aving it as chedule 8		
Low	-	-	-	-	-	-	-	-		
transaction costs	This option would not require more data than is currently calculated or available under CP5, and should not require more resources than a recalibration of the capacity charge as at CP5. However, updating the benchmark annually to account for traffic would lead to higher transaction costs and therefore all SoW are graded negative (-).									
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
NR	+	++	++	+	+	+	+	+		
accountability	This option would slightly improve the accountability of NR as it ensures that its Schedule 8 benchmark more accurately reflects its required level of performance.									
	The capacity charge refunds NR for potential delay compensation in advance of a delay actually occurring and regardless of whether the delay and subseque compensation payments actually occurs. In contrast, increasing the benchmark proposed by this option will protect NR from any additional compensation it mighave to pay should any delay actually happen. This option will have a greater impact in the SoW where there will be great flexibility to train operators (dynamic railway and on rail competition), therefore these two SoW receive a stronger positive grading (++)							nce of any ubsequent chmark as n it might be greater therefore		
Non-arbitrary	-	-	-	-	-	-	-	-		
allocation of costs	Network Rail has benchmarks on a 'Service Group' basis while the capacity charge is on a 'Service Code' basis (more granular). Therefore replacing capacity charges with updates to the Schedule 8 benchmark might make the allocation of costs less directly reflective of the source of those costs thus making the charge more arbitrary. Network Rail has stated that the capacity charge tariffs at present still "represent an 'average' rate based on the geography over which a service code operates." Therefore, the loss of granularity is unlikely to have a strong impact and this option receives a weakly negative grading (-). ¹⁰⁰									
Optimal	+	+	+	+	+	+	+	+		
traffic growth	+ + + + + + + + + + + + + + + + + + +									

¹⁰⁰ Network Rail (2013) "Capacity Charge: SBPT3272". Part of the supporting documents for CP5 SBP, link <u>here</u>.

Option 15: Recalculate benchmarks for traffic changes and remove capacity charge									
	allow for growth to exceed the optimal level, so this criterion has been graded weakly positive (+) for all SoW.								
Aligning industry incentives	=	-	-	=	=	=	=	=	
	This option does not have a significant impact on the alignment of incentives for industry parties to cooperate.								
	A benchmark without a wash-up would likely reduce NR's incentive to encourage and facilitate new traffic, as NR would face higher potential Schedule 8 costs that year but the traffic increase would not be accounted for in that year's benchmark. Train operators would instead be incentivised to operate more services as the marginal cost of each is reduced. This impact is likely to be small (=) in most SoW, but larger in the two SoW 'dynamic railway' and 'on-rail competition' where rail is more competitive and flexible. These two SoW are therefore graded slightly negative (-). A form of scarcity charge could be introduced to realign incentives in this area, if required.								
Value for	=	=	=	=	=	=	=	=	
money for funders, taxpayers and users	The increase in transaction costs would negatively affect this criterion. However, there are potential benefits for operators if this option makes the Schedule 8 and surrounding regime clearer and easier to understand.								
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
	+	+	+	+	+	+	+	+	
	There are s	some clear	benefits t	o this optior	n, howeve	er there will	be costs i	nvolved in	

There are some clear benefits to this option, however there will be costs involved in implementing the option. The initial calculation of the appropriate adjustment to the benchmark to take account of traffic might require large amounts of effort and resources, but the subsequent annual updates would be mechanical calculations which need not be complex.

This option might have a positive impact on the traffic growth, if the lag in updating the benchmark does not have a large impact on NR's ability to recover costs – although it might also encourage train operators to run less efficient services through reducing the marginal cost of running an additional train (by removing the capacity charge).

Overall, the impact is likely to be slightly positive in all SoW.

Impact on stakeholders

This option removes the capacity charge as a separate aspect of the charging and incentives regime, which is a benefit to most stakeholders as the charge is not particularly well understood across the industry (see RDG Phase 2b Feature 8.7). In particular, the name 'capacity charge' is interpreted by many to indicate that this charge is supposed to be a form of 'scarcity charge'.¹⁰¹

The Figure below demonstrates the absolute change in charges that might be experienced by each operator, if the capacity charge revenue is distributed across the FTAC in the same proportions as the 'current' split of FTAC (in CP5).

Figure E.2: Total charges by indicative operator, current charging (plain) and with this option (striped)

¹⁰¹ For example, see: Rail Freight Operators' Association (RFOA) (2013) "Freight capacity charge – proposal on methodology", a letter to the Head of Economics of ORR, available <u>here</u>.



Source: CEPA analysis using charges data from ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19." These figures assume a Network Grant has been applied.

All train operators pay the capacity charge on every train mile rather than just trains above a certain benchmark. The capacity charge is a variable component of NR's revenue and of all train operators' costs. The reduction in NR's revenue would be offset through an increase in the FTAC, which is currently payable by the franchised passenger operators. This would represent a transfer of charges from freight and open access operators, towards franchised train operators. The redistribution from non-franchised train operators to franchised operators is minimal, £7.5m of the £400m capacity charge in the model.

However, the redistribution between the franchised operators will be larger given the difference in how the capacity charge and FTAC are split across them. Regional operators' charges would reduce while the others' charges would reduce. This redistribution of funds might be undesirable, in which case a form of scarcity charge might be introduced which would re-balance the charges by charging more for use of more congested areas (which would most affect franchised commuter and inter-city operators). The increase in charges for franchised operators would affect the franchise values and feed through to their funders.

Operator (N° of operators)	capacity charge (£m)	share of industry capacity charge	current FTAC (£m)	Current share of industry FTAC	Total current charges (£m)	Change in total charges (£m)	Change in total charges (%)
Franchised commuter (10)	18	4%	13	3%	57	-7	-12%
Franchised regional (9)	9	2%	26	5%	54	12	23%
Franchised inter-city (6)	22	6%	19	4%	74	-6	-8%
Open access (2)	1.9	0.5%	-	-	6	-1.9	-34%
Multi-customer freight (4)	0.3	0.1%	-	-	7	-0.3	-5%
Bulk freight (4)	0.6	0.1%	-	-	15	-0.6	-4%
Industry total	£400m		£476m		1597		

Table E.2: Outline of indicative impact of this option of the FTAC and share of charges, charges are per operator within each sector

Source: CEPA analysis using charges data from ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19"

One potential further impact is the Sustained Poor Performance (SPP). SPP allows passenger operators to submit a claim for additional compensation if performance is at least ten percent worse than the benchmark. This is because the calculation-based compensation scheme may not be appropriate for large deviations from the performance target, and might under-compensate

operators.¹⁰² An increase in the benchmark has the same effect as increasing the SPP threshold: ten percent of the benchmark with a traffic component is equivalent to a higher than ten percent threshold on the benchmark without a traffic component. It is unclear whether this option would require or influence a change in the SPP threshold percentage, particularly as there is some disagreement on an appropriate level. ORR has reduced the threshold in recent control periods, while some have recommended that it increase, including SDG who recommended an increase to 30 percent.¹⁰³

Another consideration with the removal of the capacity charge is that it lowers the marginal cost to train operators of running trains on the network. It may be necessary for ORR's control period assessments and any traffic forecasts to take account of this change to operators' incentives.

Many of the distribution, variable cost, and incentive related concerns regarding this option could be addressed through the introduction of a 'scarcity charge' or 'avoidable costs' charge, which would reintroduce a charge that is levied on all operators and on a per km or per service basis. This option does not currently envisage this and therefore the analysis does not focus on the impact that such an inclusion would have, however it is a potentially appropriate remedy for some of the perceived negative impacts of this option.

This option may lead to an increase in volatility of the charges and incentives regime, as the benchmarks could change annually for traffic while the current regime has a greater degree of upfront predictability as the tariff (per km) is set at the Price Review for the entire control period.

Franchised commuter passenger operators	Franchised commuter passenger operators pay a large amount on the capacity charge, since they operate mostly on constrained areas of the network where the capacity charge is likely to be highest. As they are allocated only a moderate share of the FTAC, the franchised commuter passenger operator's overall charges would decrease.
Franchised regional passenger operators	Franchised regional operators pay less on the capacity charge in comparison to other franchised operators, as they tend to operate in less restricted areas of the network. They also pay a larger share of the FTAC as compared to other operators. Overall, therefore, the franchised regional operators experience the <i>only</i> increase in charges under this option and therefore an increase in their share of the overall industry charges. If this redistribution effect is undesirable an appropriate charge can be implemented, such as a scarcity charge which could be applied to all operators. Further, if a franchised operator is likely to be significantly negatively impacted under this option it may trigger the (low) threshold at which the franchise contract can be renegotiated. It may therefore be necessary to introduce a new component to franchise agreements which allows for changes through this option to be brought into contracts in a mechanistic way.
Franchised inter-city passenger operators	Franchised inter-city passenger operators pay the most (in absolute terms) on the capacity charge, since they operate predominantly on constrained areas of the network where the capacity charge is likely to be highest. As they are allocated the lowest share of the FTAC (out of the franchised operators) but the FTAC is significantly larger than the capacity charge, their overall charges would be lower.

¹⁰² ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" ¶20.50

¹⁰³ SDG for Network Rail (2013) "Specification for Sustained Poor SPP Arrangements for CP5: Final Report" p.i

Option 15: Re	ecalculate benchmarks for traffic changes and remove capacity charge
Open access passenger operators	The capacity charge represents a large percentage of the open access passenger operators' current charges. As they are unaffected by the corresponding increase in the FTAC, overall their charges would reduce.
	Smaller operators such as open access operators might also experience an increase in volatility of their Schedule 8 compensation. As the benchmark changes annually to take account of changes in traffic, whether changes in that operator's traffic or that of other operators on the network, the benchmark for Schedule 8 compensation could feasibly change substantially year-on-year, although our expectation is that year on year changes would usually be small. Larger operators would also be subject to changing benchmarks, but the smaller balance sheets of these smaller operators means that they will be less able to 'absorb' such short-notice changes.
Freight operators	Multi-customer freight operators and bulk freight operators do not differ significantly in their capacity charge payments as a percentage of their total charge obligations – they would see a moderate reduction in their charge obligations. They also do not pay the FTAC, therefore the overall impact would be a reduction in charges. However, if this redistribution effect is undesirable an appropriate charge can be implemented, such as a scarcity charge which could be applied to all operators.
NR	The impact of this option on NR depends on how the revenue requirement takes account of the loss of the capacity charge revenue. If the reduction in revenue is redistributed to the FTAC as modelled, there would not be any impact on NR's revenue. The revenue may become more predictable as a variable aspect has been replaced with a fixed aspect. NR will also pay less compensation through the increases to the Schedule 8 benchmark. The net effect on NR's net revenue would likely be neutral, assuming no adjustment to the revenue requirement.
Funders	The removal of the capacity charge, levied on all train operators, would result in an increase in the FTAC, which is currently only payable by franchised train operators ¹⁰⁴ . When franchises are renewed their value will be adjusted to take account of the change in charges faced by these operators. Therefore, the funders might become responsible for a portion of the 'burden' of this option. If a franchised operator is likely to be significantly negatively impacted under this option it may trigger the (low) threshold at which the franchise contract can be renegotiated. It may therefore be necessary to introduce a new component to franchise agreements which allows for changes through this option to be brought into contracts in a mechanistic way. At present franchised passenger operators bid for the franchise based on their Schedule 4 and 8 assumptions. This option might result in potential franchisees increasing the risk premium included in their bid, if they expect to experience annual volatility in their benchmark (and therefore Schedule 8 compensation). This would lower the franchise values if volatility is expected to be large, however the intention of this option is to smooth changes accounting for changes annually rather than having a 'step-change' at every Price Review. Additionally, it might be possible to remove the need for any additional volatility risk premium by ensuring that changes to the Schedule 8 benchmark would be appropriately reflected in the franchisees' contracts.

¹⁰⁴ Noting that the level of the capacity charge paid be other operators is very small in the context of the FTAC

infrastructure, therefore cross-border services from England & Wales franchises do not pay any of the FTAC on the section of their cross-border services that operate in Scotland. Similarly, the Scottish franchised operator is not required to pay FTAC on the sections of their services that operate in England.¹⁰⁵ Transport Scotland stated at PR13 that "the amount of Fixed Charges foregone in each network is not equal, which results in an imbalance in the system," and a transfer from the capacity charge to the FTAC could worsen any imbalance. Cross-border services from England which were previously paying the capacity charge in Scotland would stop doing so, with the amount instead allocated to the Scottish FTAC (and similarly for Scottish cross-border services in England).¹⁰⁶

Passengers Passengers might benefit if there is a positive impact on optimal traffic growth and if the regime being simpler results in a more efficient service. Freight users may have less predictable costs if they are on a flexible contract which allows the freight operator to pass through any reduction in compensation revenue that results from a change in NR's benchmark.

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Funders and passenger train operators raised concerns that annual adjustments would take time and resources. This option has therefore been envisaged as a simple and mechanistic adjustment which should reduce the transaction costs involved while having the benefit of improving the transparency of the relationship between performance and variations in traffic levels. Network Rail noted that a similar mechanistic approach is already in place for freight operators' benchmarks, and that seems to reflect the impacts of changes in traffic well without being overly complex.
- The current capacity charge level depends in part on the use of the network by other operators, but the concern raised by the freight sector is that this option would increase that effect since the current capacity charge is levied per train mile actually travelled by the operator. In contrast, this option would adjust Network Rail's benchmark for all operators in that service group. A further concern raised by passenger operators is that changing Network Rail's benchmarks may have knock-on effects on TOC benchmarks given that it will adjust the level of delay that is considered 'efficient'.
- DfT were concerned that by creating more frequent opportunities to change the benchmarks and therefore the terms of the franchised operators contracts, there would be more frequent requests by franchised operators for renegotiation of the contracts. Assuming the impact is sufficient to be a Qualifying Change this would place additional resource requirements on both DfT and the franchised operators. It may therefore be useful to consider whether the mechanistic process employed for the charge could be extended to dealing with impacts on the franchise agreements.
- DfT and Transport Scotland expressed concerns that this option could cause a 'shift' in costs to Transport Scotland due to the arrangements that cross-border services currently have in place with regards to the capacity charge and FTAC. Currently, cross-border services pay the FTAC in the operator's country only, and pay the capacity charge relevant to the route regardless of country. Therefore, removing the capacity charge would remove the element of charges that cross-border services pay to use the infrastructure outside of its own country.

¹⁰⁵ ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" p.639-640

¹⁰⁶ Network Rail (2013) "Periodic Review 2013: Fixed Charges in CP5 – Conclusions" Section 3.5, pp.13-15

ANNEX F RECOVER END-USER COMPENSATION THROUGH THE PERFORMANCE REGIME (SCHEDULE 8) DETAILED ASSESSMENT

Option 18: Recover end-user compensation through the performance regime (Schedule 8)								
Overall performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
+	++	++	+	+	+	+	+	

Schedule 8 reimburses train operators for estimated revenue losses as a result of disruptions to its services, given that passengers may choose not to travel in future due to the reputational impact of disruptions to service. This option was selected for detailed assessment given that the performance regime does not provide compensation to train operators that are required to provide delay compensation to their end-users. This option particularly looks at passenger operators as these are required by contract to provide end-user compensation, and also recognises that it might be appropriate to include an element for freight operators.

A key objective of this option is to make the intention of the performance regime clearer to the public, as there is currently a high level of misunderstanding in assuming that the Schedule 8 compensation for passenger operators should be passed onto passengers as delay compensation, rather than its actual intention to compensate train operators for the longer term revenue impact of delay (and some short-term impacts such as replacement buses). Given that passenger compensation is currently undergoing a series of changes, it is an appropriate time to consider how passenger compensation requirements could be incorporated into the performance regime. It is important, at the same time, to consider whether an appropriate end-user compensation mechanism for freight could be implemented given that freight operators often compensate their end-users for delays.

Schedule 8 currently compensates train operators for the longer-term financial impact of disruptions to services. There is no component to reimburse train operators for end-user compensation (i.e. passenger compensation paid through Delay Repay, or freight operators' case-by-case contract); this option proposes introducing such a component in addition to the current Schedule 8 compensation.

This option aims to redistribute the risk associated with end-user delay compensation. Currently in passenger services all of the risk of delay compensation is on the passenger operators, despite each type of passenger operator causing only between 10 and 30 percent of their own delays. While franchised passenger operators do include a 'risk premium' for this into their franchise bid value, they are then at risk for any deviations from their estimate.

This option has several clear benefits. It would be likely to reduce incentives on passenger operators to limit compensation paid to passengers. It also aligns the cost of delays more closely to those who caused them by 'passing through' an element of passenger compensation to the at-fault party. The main concerns relate to its practicality and transaction costs. In particular, the effectiveness of this option relies on passenger and freight operators passing through compensation to their end-users. This might be particularly difficult to measure in freight where explicit compensation payments might be foregone to instead charge customers lower prices to account for the risk of potential delay. There is typically a low claim rate among eligible delayed passengers (no comparable freight data is available), often attributed to a lack of awareness driven by train companies not actively providing passengers with the relevant information. Use of delay claims processes, such as cash refunds and online forms, make it likely that there will be an increase in the rate of claims by those eligible. Therefore, an automatic refund might provide an effective delivery mechanism to ensure that there is not a large disparity between the amount claimed by passengers and the amount passed through Schedule 8 (at a lower level of transaction costs than some manual methods would

bring).

There are difficulties associated with applying this option to freight, since there are not standard compensation terms for freight operation, and it could be discriminatory to make use of the varying and confidential terms in contracts. In principle, standard rates could be applied according to some categorisation. But it would also impact upon the economic viability of freight if this resulted in increased payment rates to cover the cost of the scheme – freight operators may prefer to take the risk themselves and pay lower payment rates. Nevertheless, implementing this option will need to ensure that Schedule 8 is not unduly discriminatory against freight: freight should be included in the two-way payments if it is deemed appropriate. This option assumes that it is possible to determine an appropriate mechanistic calculation for freight end-user compensation, however it is not certain that freight would desire the additional complexity that even a simple mechanistic approach would bring when freight operators are currently able to negotiate the issue through their contracts individually with each customer.

While our analysis has showed an overall positive grading of this option, the magnitude is unclear, and further work would be required on reasonable practicality and cost. Train operators might also oppose the added complexity that this might bring to Schedule 8 (see RDG Phase 2b Feature 8.12). This option will, in several areas, have a stronger impact (negative or positive as appropriate) in the SoW where there will be greater flexibility to train operators (Dynamic railway and On-rail competition). However, these extremes have balanced each other out to mean that these SoW are given a small positive grading as with the other SoWs.

Key characteristics

Description of option

This option would involve an additional component of Schedule 8 compensation which would allow train operators to recover end-user compensation (passenger or freight) that they paid out as a result of delays which are attributed to Network Rail or other train operators. This compensation would be in addition to the current Schedule 8 compensation measure, which is intended to allow train companies to recover the long-term financial impact of unplanned service disruption.

The additional compensation would be included as an additional calculation in Schedule 8. This calculation would allow train operators to recover a specified amount of compensation according to the type and severity of delay. For passenger services the amount would also take into account the number of passengers and the value of tickets (as passengers on a single train may have paid very different prices, which affects the compensation they are eligible for).

A simple implementation of this option could involve having a set calculation which determines how much compensation a train operator should receive. However, receiving set compensation to pass through to end-users might create a perverse incentive for operators to avoid passing it on (and retain the compensation). Therefore, for passenger operators this option proposes that:

- for passenger operators which operate automatic refunds (Virgin Trains West Coast¹⁰⁷ and C2C from 2016¹⁰⁸), if they can obtain and share appropriate data on passenger refunds then actual data regarding the number of 'claims' will be used in the calculation; and
- for all other passenger operators, their compensation calculation would assume that the same

¹⁰⁷ Advance tickets for a specific train, delayed more than 30 minutes, are eligible for a refund if the ticket was bought through an account with Virgin Trains online.

BBC News (2015) "Virgin starts automatic compensation for train delays" available here.

¹⁰⁸ Delayed passengers who travelled using a smart card will be refunded for delays of two minutes or more. C2C (2014) "Passenger's Charter Nov 2014"

percentage of passengers claimed for eligible delays for the previous year, such that improvements in claimant numbers are taken account of but at a lag.

This dual approach is likely to help to ensure that those operators with the available data receive accurate end-user compensation while whose without the data do not have to wait the additional four weeks before the paper-based claims are complete.

This option aims to redistribute the risk associated with end-user delay compensation. Currently in passenger services all of the risk of delay compensation is on the passenger operators, despite each type of passenger operator causing only between 10 and 30 percent of their own delays. While franchised passenger operators do include a 'risk premium' for this into their franchise bid value, they are then at risk for any deviations from their estimate. This option would introduce more attribution of delay and better alignment of financial incentives – as those who cause the delay would be required to cover more of the cost of that delay to the train operator.

This option would not 'solve' the issue of passenger operators creating barriers to paying delayed passengers compensation, and that is not its key objective. However, it would be likely to provide passenger operators in particular with less reason to create such barriers and create the possibility for ORR to place more pressure on them to pay make compensation more routine and proportionate, and maybe on uniform rules. If the end-user compensation is recoverable based on the dual system described above, then operators should have greater incentives to make compensation routine, since doing so should have a positive impact on their reputation and passenger demand.

There is no published data on comparable end-user compensation for freight operators. End-user compensation in freight is part of the freight operators' private contract with its customers. Therefore, to include any component of freight compensation in any 'end-user compensation' recovery mechanism, freight operators would need to work together with ORR to determine an appropriate formula. The assessment of this option assumes that freight operators are included in the end-user compensation mechanism but does not include freight in the modelling due to lack of available data.

See RDG Phase 2b Report Feature 8.12

Description of counterfactual

Schedule 8 has two main components: ¹⁰⁹

- an incentive to maintain performance, by aligning the financial incentives to not delay other parties; and
- a compensation mechanism, to reimburse operators for some of the *long-term* financial impact of delays to services above a certain level (some amount of delay will have been included in the assumptions made when bidding for a franchise or route).

Long-term financial impact of delay refers to the impact of poor performance on an operator's longterm revenue, and Schedule 8 aims to reduce the extent to which this is factored into franchise bids with a 'risk premium'.¹¹⁰ The calculation of payment rates for takes account of the Passenger Demand Forecast Handbook (PDFH) values on the responsiveness of long-term passenger demand to service disruption. It is not designed to cover the on-the-day costs of delay, including the compensation paid to delayed passengers.

¹⁰⁹ RDG (2015) "Review of Charges Phase 2b: Assessment of the current charges and incentives regime." Features 8.2 and 8.12

¹¹⁰ The difference between Schedule 8 and passenger compensation is elaborated upon in: ORR (2013) "Final determination of Network Rail's outputs and funding for 2014-19" ¶20.15

Freight compensation is part of the freight operators' private contracts with their customers, and may take the form of forgone payments rather than explicit compensation. Passenger compensation is a requirement of franchise agreements, and is therefore more transparent. There are currently two passenger compensation schemes in coexistence, the voluntary Delay Repay (which pays from 25% of the ticket price for delays longer than 30 minutes) being more generous than the compulsory minimum set by the National Rail 'Conditions of Carriage' (which pays from 10% of the ticket price for delays longer than 60 minutes).¹¹¹

Relevant factors impacting the form and/or the effectiveness of the option

- Franchising (Factors Report Section 3.2): there is a low threshold of changes to the revenue/ costs
 of franchisees before they are allowed to request a renegotiation of the franchise. This option
 might cause franchisees to meet that threshold, incurring additional costs for ORR. Further, that
 franchises are staggered to start and end at different times would mean that operators may bid in
 a different amount of 'risk premium' for passenger compensation depending on when their bid
 process occurred in relation to this option being implemented. However
- Balance of risk and reward for asset-light companies (Factors Report Section 4.5): Asset-light industry participants have little ability to absorb risk, and this option would reduce the risk to an operator of delays by other parties, but would also increase the risk to an operator of its own delays.
- Data availability, measurement, and billing (Factors Report Section 4.7): This option will incur costs in calculating and implementing. In passenger compensation the information should already be available, however the delivery mechanisms may not. In freight

Implementatior	1					
Information requirements	To implement this option it is necessary to determine the value of compensation each operator would need to recover. For freight this calculation may be complex and much of the information required is commercially sensitive or confidential. For franchised passenger operators the data would be more readily available, and four elements would impact the total value of passenger compensation from a delay:					
	 the number of passengers on the service; 					
	 the percentage of passengers that will claim; 					
	• the length of the delay;					
	 the value of their tickets / compensation; and 					
	 the agreed levels of recoverable compensation. 					
	Currently only the length of the delay is recorded for the current Schedule 8 calculations, however with the introduction of automatic refunds the remaining information should become more readily available.					
	Automatic refunds make collecting the required data easier and faster, as compared to using estimates or waiting for the 28 day claims period to elapse so that the actual figure is known. Estimates are further complicated by the mix of types of ticket, such as first class or standard, and advance, season ticket, or full- price tickets.					
	If the cost of compensation is distributed in the same way as the current Schedule 8					
¹¹¹ For further d website.	etail on the differences between the two main schemes, please see the Transport Focus					

Option 18: Reco	over end-user compensation through the performance regime (Schedule 8)					
	payments, where the party that causes the delay is responsible for paying the compensation, it would be necessary to know which party caused the delay. This attribution data should already be available through the current Schedule 8, as should the 'monitoring point weightings' which provide estimates of passengers alighting at each (major) station.					
Drivers	There are five main components which would impact the size of the compensation that a party is obliged to pay to a passenger operator, and the compensation a party is due from a delay:					
	 the number of passengers, which is assumed to be out of control of the party at fault for the delay – however if the delay in question is due to Network Rail carrying out emergency works it might be possible for it to choose a less busy time of day; 					
	 the percentage of passengers that will claim, which is assumed to be partly out of control of the delayed operator although it is able to influence the percentage by making the system more or less difficult; 					
	 the length of the delay, which is assumed to be within the control of the party at fault for the delay; 					
	 the value of the tickets/ compensation per passenger, which will be affected by the ticket values and also by the compensation scheme that the passenger operator has in place (i.e. Delay Repay or Conditions of Carriage minimums); and 					
	• <i>the agreed levels of recoverable compensation,</i> which might differ from the passenger operator's standard or actual levels of compensation paid out, to standardise the option across the network and its operators.					
	Of these factors, only the value of compensation per passenger assumed to be within the control of the delayed passenger operator, and the <i>recoverable</i> level of compensation (as a percentage of ticket value) should be determined in advance to maintain consistency.					
	This compensation would be additional to the current Schedule 8 compensation, but it may be appropriate to adjust the original Schedule 8 compensation if this option were introduced. One of the considerations in the calculation of the long-term financial impact on passenger operators is the impact on future passenger demand of the negative reputational impact of disruption to services. If more compensation is paid out under this option, as train operators would have less of an incentive to avoid paying passenger compensation, then this 'future demand' aspect could be reduced and the current Schedule 8 compensation might be an overestimate of the actual impact.					
Calculation principles	The Figure below demonstrates a potential calculation that could be used to estimate the appropriate amount of passenger compensation to recover.					
	Figure F.1: Calculation of the passenger compensation to be included in Schedule 8					



It is important not to over-compensate train operators for delay (i.e. compensate for a level of compensation higher than that actually paid out), to avoid creating perverse incentives. This means that there should be some way to ensure that the compensation is being passed onto the end-users; in passenger services this could be through automatic payment ensuring that the amount is instantly paid to passengers, or through using the previous year's claim rate (% of eligible journeys claimed for) when determining the amount the operator should receive.

This compensation would be additional to the current Schedule 8 compensation, but it may be appropriate to adjust the original Schedule 8 compensation if this option were introduced. One of the considerations in the calculation of the long-term financial impact on train operators is the impact on future end-user demand of the negative reputational impact of disruption to services. If more compensation is paid out under this option, as train operators are likely to have less of an incentive to avoid paying end-user compensation, then this 'future demand' aspect could be reduced and the current Schedule 8 compensation might be an overestimate of the actual impact.

- Practical A simple implementation of this option could involve having a set calculation which determines how much compensation a train operator should receive. However, receiving set compensation to pass through to end-users might create a perverse incentive for operators to avoid passing it on (and retain the compensation), while waiting for all passenger operators to be on the automatic refund systems would create a very long lead time. Therefore, for passenger operators this option proposes a dual system, where:
 - for passenger operators which operate automatic refunds (Virgin Trains West Coast¹¹² from October 2015 and C2C from 2016¹¹³), if they can obtain and share appropriate data on passenger refunds then actual data regarding the number of 'claims' will be used in the calculation; and
 - for all other passenger operators, their compensation calculation would assume that the same percentage of passengers claimed for eligible delays

¹¹² Advance tickets for a specific train, delayed more than 30 minutes, are eligible for a refund if the ticket was bought through an account with Virgin Trains online.

BBC News (2015) "Virgin starts automatic compensation for train delays" available here.

¹¹³ Delayed passengers who travelled using a smart card will be refunded for delays of two minutes or more. C2C (2014) "Passenger's Charter Nov 2014"

Option 18: Re	cover end-us	er comper	nsation th	rough the p	erforman	ce regime (Schedule 8)		
	for the previous year, such that improvements in claimant numbers ar taken account of but at a lag.								
	The dual approach outlined above is likely to help to ensure that those operato with the available data receive accurate end-user compensation while who without the data do not have to wait the additional four weeks before the pape based claims are complete.							operators e whose ne paper-	
Lead time	d time This option also requires an appropriate formulaic approach for recovery of en user compensation in both freight and passenger services. This would require consultations and iterations, and therefore would have a likely minimum lead of 18 months before full implementation.						f end- ire ead time		
Resources To enable operators to required for delayed them, it is likely the supplementation current "STAR" model in provide the supplementation current the supplementati				r compensa his option w for Schedul	tion from yould use le 8 compe ibution ro	any other o a pooling sy ensation.	operators th stem simila	at have r to the	
	estimatio	ns as in the	e current S	Schedule 8 c	ompensat	ion.	assenger m	linder	
	Additiona compensa	l resources ation in ead	s would be ch case, ev	e required to ven with a fo	o determin ormulaic a	ne the appropriation of the product	opriate		
Performance	against criter	ia							
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
System safety	=	=	=	=	=	=	=	=	
	This option Network Rai be caused b incentive to compensatio ambiguous, a	This option would increase Schedule 8 costs to Network Rail. This could increase Network Rail's incentive to invest by increasing the potential cost of delays that might be caused by a lack of investment. Conversely, it could also reduce Network Rail's incentive to carry out short-term emergency works through increasing the compensation it will have to pay for the disruption. The overall effect is therefore imbiguous, and receives a neutral (=) grading.							
Consistency	=	=	=	=	=	=	=	=	
with law	We do not component legislation is is largely r Regulations	/e do not anticipate that introducing an additional 'end-user compensation' omponent to Schedule 8 will have any legal implications. The most relevant EU egislation is Directive 2012/34/EU (recast of Directive 2001/14/EC), and in the UK this largely mirrored in The Railways Infrastructure (Access and Management) egulations 2005. ¹¹⁴							
In particular, it is necessary to have a performance scheme (wh penalties, compensation, and bonuses) which is targeted at mini improving the performance of the network, in particular the regi							iich can include any of imising disruption and ime must:		
	• be n	on-discrim	inatory ac	ross the net	work;				
	• enco	ourage opti	imal use o	f the netwo	rk and its (capacity;			
	• prov	vide sufficie	ent incenti	ves and pric	e signals t	o participai	nts;		
	• refle	ect the cost	s incurred	I IN providing	g the servi eimburser	ice	luser com	ansation	
		ining that li	inhiies au		ennburser		i-usei comp		

¹¹⁴ We emphasise that this does not constitute legal advice.
Option 18: Re	cover end-	user compe	nsation t	hrough the _l	performan	ce regime	(Schedule 8)
	is prohibite	ed.						
	In addition vary mode that are at model clau	In addition, ORR has the power under the Railways Act 1993 to prepare, publish, and vary model clauses for track access agreements. Model clauses are standard clauses that are attached to all track access agreements of similar type. In particular, these model clauses set out the charges and incentives.						
Funding of	-	-	-	-	-	-	-	-
Network efficient costs	⁵ Under this option, train operators would recover some end-user compensation from Network Rail for all trains which Network Rail has delayed. This applies to disruptions. This is in contrast with the Schedule 8 compensation, which is benchmarked regime and therefore train operators are only eligible for compensati if overall performance is worse than the benchmark (and Network Rail is eligible for reverse 'bonus' if it outperforms its benchmark). This option would therefor represent a small negative impact (-) on the funding of Network Rail's efficient cos even disruption which is within Network Rail's benchmark would make Network R liable to pay for an amount of end-user compensation.						tion from ies to <i>all</i> hich is a pensation gible for a therefore ent costs; twork Rail	
Allowance for	=	=	=	=	=	=	=	=
market conditions	This option whether it specific for have comm however fr arrangeme determine negative in	would ref represents rm of the c mitted to j reight opera ints. Howe the appro npact on th	und an es a single a calculatior oin Delay ators have ver, by priate co is criterior	timation of approach for that is inc repay, the confidentia implementin mpensation n and has be	compensa the netwo luded in S erefore dir al and priv ng a simp amount, een graded	tion provic ork as who chedule 8. fferences a ately deter ole mecha this optior neutral (=)	led to end-u le will depen Passenger are likely to mined comp nistic calcu n should no	users, and nd on the operators o narrow, pensation lation to ot have a
A single	=	=	=	=	=	=	=	=
approach for the network as a whole	This option would refund an estimation of compensation provided to end-users, and whether it represents a single approach for the network as whole will depend on the specific form of the calculation that is included in Schedule 8. Passenger operators have committed to join Delay repay, therefore differences are likely to narrow, however freight operators have confidential and privately determined compensation arrangements. However, by implementing a simple mechanistic calculation to determine the appropriate compensation amount, this option should not have a pagative impact on this criterion and has been graded poutral (=).							
Objectives	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
Service costs	=	=	=	=	=	=	=	=
recovery	This optior	should not	: significar	ntly affect th	e efficient	service cos	sts recovery.	
Efficient	=	=	=	=	=	=	=	=
whole-system whole -life industry net costs	This optior	ı should not	affect ef	ficient whole	e system w	hole life in	dustry net c	osts.
Efficient long	+	+	+	+	+	+	+	+
run investment	This optio investmen	n increases t incentives	s Schedu in all SoV	le 8 costs V, by increas	to Netwo sing the po	rk Rail. Th tential cos	nis may inc t of delays t	crease its hat might

Option 18: Re	ecover end-ເ	iser compe	nsation th	nrough the j	performan	ce regime (Schedule 8)	
decisions	be caused b	by a lack of	investme	nt.					
	However, t short notic likely to be	his option o e unless es small.	could also sential, th	reduce Net rough incre	work Rail' easing the	s incentive t cost of doi	to carry out ng so. This	t works at impact is	
	There is a small risk that Network Rail could focus its investment in the areas we are likely to cost it more on compensation through the train operators having m generous schemes (rather than just because more end-users are delayed). Howe any such impact is likely to be small and could be mitigated by implementing a for each delay category. Therefore, the overall impact of this option is likely to positive in all SoW.								
Efficient	+	+	+	+	+	+	+	+	
performance management	This option should incentivise the efficient management of unplanned work through increasing the cost to Network Rail of causing disruption above its benchmarked level, as this regime is directly about Schedule 8. However, it might also have a positive impact on Schedule 4 if Network Rail is incentivised to become more efficient at delivering its planned works (covered by Schedule 4) to reduce the risk of overrunning and being subjected to Schedule 8 and the related compensation payments.								
Efficient use	=	=	=	=	=	=	=	=	
capacity	This option should not create large distortions to incentives for the allocation and use of available network capacity.								
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Predictability	=	=	=	=	=	=	=	=	
	If using a calculation/estimation method, this option does not impact the predictability of the Schedule 8 payments, as the payments can be calculated with a pre-determined formula shortly after the delay has occurred, as at present.								
Simplicity	-	<u>-</u>	-		-		-	-	
	This option has been gr	will make	Schedule Iy negativ	8 slightly n e (-).	nore comp	plex and the	erefore this	criterion	
Transparency	=	=	=	=	=	=	=	=	
	This option	should not	have any	impact on t	he transpa	arency of ch	arges.		
Low	-	-	-	-	-	-	-	-	
costs	Calculating the compensation amounts would incur transaction costs. However as the network adopts smartcards on a growing basis, automatic refunds will become more readily and widely available. Therefore, the same calculation as used for the automated refunds will be included in the Schedule 8 payments to ensure full recovery of agreed levels of compensation (i.e. if industry agrees on 50% passenger compensation for 30-60 minute delay and an operator pays out 80%, they only recover 50% and absorb the additional 30% themselves, and similarly for freight). Under this scheme, the increase in transaction costs (of implementing this option, assuming automated refunds are funded elsewhere as a separate scheme with its own benefits) is minimised in all SoW. Therefore, this criterion is graded negative (-)								

Option 18: Re	ecover end-ເ	iser compe	nsation th	nrough the p	performan	ice regime (Schedule 8	
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
Network Rail	++	++	++	++	++	++	++	++
accountability	This option more repre operators,	i increases esentative a in all SoW.	Network mount of	Rail's accou the financi	untability al impact	through ens that its dela	suring that iys have on	it pays a the train
Non-arbitrary	=	=	=	=	=	=	=	=
allocation of costs	This option enhanceme	n is unlike ent/service	ely to ha	ive an imp	act on t	he non-arb	oitrary alloo	cation of
Optimal	=	=	=	=	=	=	=	=
traffic growth	This option should have a positive impact on promotion of efficient long run investment decisions, without negative impacts for the promotion of efficient industry costs. The impact on this criterion is therefore likely to be neutral or slight positive, and receives a neutral grading (=).							
Aligning	+	++	++	=	+	+	+	+
incentives	the party r Including th to further in may be str are more e franchises'	responsible ne short-ter mprove this onger in th exposed to where fram	for the f rm costs i alignmer e 'dynam risk (++), chisees ar	Financial impresentation of the second secon	pact of a paying com all effect s and 'on-rai trong in t red to risk	delay is ma npensation hould be po il comp' So\ the SoW m (=).	ade to rein to end-user isitive (+).Th N where fra ore highly	The trace is a likely nis impact anchisees 'specified
Value for	+	++	++	=	+	+	+	+
funders, taxpayers and users	This option incentivised passengers refunds, an passengers which they may be larg cost of the operators a This impact franchisees 'specified f	could imp d to make Although increase perception experience ger than in t compensat ind their en t may be st are more ranchises' weaker pos	rove the the clair the oper in passe n of the i ed a delay the currer tion. Simil d-users, a ronger in exposed where fra itive gradi	passenger r ms process ator would enger comp ndustry and but receive nt system wi arly this opt arly this opt s there is cu the 'dynam to risk (++), nchisees are ng (+).	efund pro easier ar incur adr ensation in particued adequa here the p tion would irrently no nic railway and less e less exp	beess if pass and more ac ministrative paid out i ular of the r te compens bassenger op d improve th ot a clear acr ' and 'on-ra strong in th osed to risk	enger oper ccessible to costs in pa s likely to route or operation. This perator bea he system fir coss-the-ind hil comp' So he SoW most s (=). All oth	ators are delayed aying out improve erator on incentive rs the full or freight ustry W where ore highly her SoWs
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
	+	++	++	+	+	+	+	+
	This option value for m it would in lower if aut There would	would be oney, inves crease trans comatic refu ld need to	likely to h tment inc saction cc Inds are a be a sepa	have positive centives, and osts. The inc vailable acro rate mecha	e impacts d allocatio rease in tr oss the en nism to co	on several n of costs (i ransaction c tire rail netv ompensate	criteria, in n all SoW). costs would vork. freight open	particular However, be much rators for

end-user compensation. This would be more complex than the passenger compensation mechanism as the end-users in freight have individual contracts rather than the two broad categories of compensation in passenger compensation. It would likely be necessary to include a component for freight as otherwise there might be an imbalance in the treatment of different operators.

The magnitude of the impact of this option is unclear, although it is likely to be stronger in the 'dynamic railway' and 'on-rail comp' SoW where franchisees are more exposed to risk (++). It is also not clear that train operators would desire the added complexity that this might bring to the compensation payment (See RDG Phase 2b Feature 8.12).

Impact on stakeholders

This option aims to redistribute the risk of delay costs relating to end-user compensation, by requiring those who created the delay to reimburse an amount of end-user compensation. The table below shows how this would impact passenger operators (freight has been excluded from this chart due to lack of data).





Source: CEPA analysis

The data used in modelling includes:

- passenger compensation paid out by franchised passenger operators in 2014-15 (with an estimate for open access operators based on relative passenger traffic);¹¹⁵
- attribution of delay into, the operator caused its own delay, or it was Network Rail, another passenger operator, or a freight operator.¹¹⁶

We have not modelled any aspect of freight end-user compensation, as the system of freight enduser compensation is more complex than in passenger services where the system is more

¹¹⁵ Department for Transport, via gov.uk (2015) "Compensation paid by Train Operating Companies: Passenger's Charter & Delay/Repay 2009-10 – 2014-15."

¹¹⁶ Network Rail's performance statistics for the year to 22nd August 2015 as published on their website, available <u>here</u>. Network Rail also publishes annual data through the National Rail Trends (NRT) Portal, but that data does not include a 'freight' attribution category.

transparent.

The Figure above provides an indication of the net flows that might be expected from implementing this option looking at passenger compensation alone (freight compensation would be included but was not included in the modelling due to lack of data). Passenger operators cause 20-30 percent of their own delays and approximately 10 percent of other passenger operators' delays. As passenger operators will also be able to recover from Network Rail for the 60 percent of delays that they cause, passenger operators are likely to receive more than they pay out under this option. The total passenger compensation paid is likely to rise as compensation schemes become more accessible to passengers and passenger operators (assuming that delays stay the same).

Sixty percent of historic passenger delays are attributed to Network Rail, although a quarter of those were due to external factors (such as weather or fatalities). To the extent that Network Rail are at risk, this option could therefore provide Network Rail with a further incentive to reduce delays, thus aligning its incentives more closely with the passenger operators – this should benefit passenger operators and passengers. It is likely that freight will have a similarly high proportion of its delays attributable to Network Rail, and therefore Network Rail may face the same incentive here. The strength of any incentive is not clear as it depends on the extent to which Network Rail is 'at risk' for the compensation it pays out under this, therefore this incentive should be considered a side effect or secondary impact rather than a key objective.

Franchised commuter passenger	Franchised commuter operators are attributed with approximately 30 percent of their own delays, therefore under this option they would recover up to 70 percent of the passenger compensation paid out.					
operators	As the majority of delays are attributed to Network Rail, even taking into account the amount that commuter operators will have to pay the other passenger operators results in a net 'gain' for franchised commuter passenger operators through this option.					
Franchised regional passenger operators	Franchised regional operators face a similar split of delay attribution as franchised commuter passenger operators. However, in 2014-15, regional operators paid out significantly less compensation than commuter operators and inter-city operators. There is a possibility, therefore, that these operators could pay out more to other operators than they receive directly from them. However, as Network Rail accounts for almost 60 percent of delays, franchised regional operators could potentially receive a net gain through this compensation mechanism.					
Franchised inter-city passenger operators	Inter-city passenger operators cause a lower percentage of their own delays than other franchised operators, at 20 percent, and also have the highest absolute compensation paid to passengers. Therefore, they may stand to benefit the most in terms of how much of their compensation outlay is recovered through this option.					
Open access passenger operators	Open access operators cause only 10 percent of their own delays, with other passenger operators causing 20 percent. Data on the amount of compensation paid by open access passenger operators is not available for analysis, therefore the model assumes that these operators pay out an amount of compensation per train km proportional to the franchised passenger operators. Open access operators do operate passenger compensation schemes and therefore would be eligible to recover through this option. ¹¹⁷					

¹¹⁷ First Hull Trains operate a system similar to Delay Repay (advertised <u>here</u>), while Grand Central Rail provides compensation according to the National Rail Conditions of Carriage (advertised <u>here</u>).

Open access operators are at a higher risk of receiving less than they might be required to pay out to other operators, given their small size as compared to the operators they might be expected to compensate.

Freight Using the published 2014-15 passenger compensation and delay attribution data, operators freight would be responsible for paying approximately £1m of passenger compensation. This figure is likely to rise as compensation schemes become more accessible to passengers and passenger operators (assuming that delays stay the same). However, this assessment has not been able to take into account the reciprocal compensation to freight.

Network Rail Under this option, all train operators would recover an amount of end-user compensation from Network Rail for all trains which Network Rail has delayed. This applies to *all* disruptions. This is in contrast with the Schedule 8 compensation, which is a benchmarked regime and therefore train operators are only eligible for compensation if overall performance is worse than the benchmark (and Network Rail is eligible for a reverse 'bonus' if it outperforms its benchmark). Therefore, even when performing at an 'efficient' level under Schedule 8, Network Rail will be required to pay out under Schedule 8.

Network Rail (and other operators causing delays) would be incentivised, through higher payments to other parties, to minimise delays. This would represent higher costs to Network Rail. Network Rail is responsible for the vast majority of delay and would therefore incur the highest costs of any party.

Funders Network Rail would be liable for the greatest amount of compensation under this scheme, and this cost would likely be passed through to DfT. However, there is a potential opportunity for ORR to use this to create an additional performance incentive to Network Rail.

At present, the operators are 'at risk' for the compensation that must be paid out. Franchised passenger operators, for example, therefore build an appropriate risk premium into their franchise bids. Under this option, that risk premium might be lower as the operators would be less at risk, which in the case of franchised passenger operators would increase the franchise value.

Franchised passenger operators would stand to gain from this option in the short term, as they have factored passenger compensation into their franchise bids but would under this option become able to recover some of this compensation from Network Rail and other passenger operators. This means that the current franchises would effectively become more valuable than they were at the time of bidding.

This option is not proposed as a 'benchmarked' compensation scheme. That is, train operators would recover end-user compensation from Network Rail for all trains which Network Rail has delayed and caused the operator to incur end-user compensation liabilities. This applies to *all* disruptions, not just those above a certain level.

PassengersPassengers should benefit as delayed passenger operators should be more inclined to
publicise and make it easier for delayed passengers to claim compensation. A
secondary impact is that Network Rail and other operators may be more incentivised
to prevent delays to passenger services.

Train operators are currently 'at risk' for the compensation that they are obliged to pay to delayed end-users; for franchised operators this is limited as they estimate a

certain amount of compensation when determining their franchise bids.

End-user compensation in freight is part of the freight operators' private contract with its customers. Therefore, to include any component of freight compensation in any 'end-user compensation' recovery mechanism, freight operators would need to work together with ORR to determine an appropriate formula. The modelling determined that freight's share of passenger compensation attribution would be approximately £1m if they were required to pay it.

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- There was some consensus on considering this option, as something which might improve the passenger perception of the industry. The intention of the performance regime, passenger operators noted, could be clearer to both industry and passengers.
- It was noted by the freight sector, but applies to both freight and passenger operators, that increasing the costs of delaying a service increases the expected cost of running an additional service if it is assumed that an additional service might at some point delay other operators. However, this is part of the incentive impact that is intended with the redistribution of risk through this option: this option envisages that operators would be at risk for the delays that they cause to other services.
- It would be more difficult to implement this option for freight operators. Freight contracts with their customers are confidential, one-to-one, and likely to vary widely depending on the type of customer (some types of commodity may be more affected by delays to transportation than others). Furthermore, freight operators are able to choose to forgo a compensation clause in a contract to instead charge their customers lower prices to account for the risk of potential delay.
- Franchised passenger operators will have included in their franchise bids a risk premium to reflect the uncertain level of passenger compensation they will need to provide. This option aims to reduce that risk premium in new franchises but given the staggered franchising process there would be a time lag until it was accounted for in all contracts. There could be some transaction costs and transitional issues if ORR tried to implement this option without double-reimbursing franchised operators.
- Passenger operators asked whether this option fits within Schedule 8, or whether it might be better placed as part of a wider review of passenger compensation. They also questioned the extent to which this option would impact Network Rail's incentives to prevent delay, as it is small in magnitude compared to overall business and Schedule 8.

ANNEX G MORE FREQUENT ACS CALCULATION DETAILED ASSESSMENT

Option 19: More frequent ACS calculation									
Overall performance against the RDG Vision in each SoW									
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
+	++	+	+	+	+	+	+		

The Access Charge Supplement (ACS) allows Network Rail to recover the amount it is expected to pay out in passenger Schedule 4 compensation over the control period, should it undertake the estimated level of works efficiently. The ACS is set at the start of a price control period based on the estimated volume of engineering works (maintenance and renewals) that will be carried out during that control period and a Schedule 4 unit cost for each type of activity estimated from historical data.

This option was selected for detailed assessment as it addresses some concerns in the industry that Network Rail has tended to delay work relative to the anticipated programme, thus taking fewer possessions than scheduled, and resulting in over-recovery of costs relative to the ACS set at the periodic review.

We envisage that the most likely possibility for implementing this option involves recalculating the ACS annually based on changes in the volume of engineering works planned, as more accurate predictions become available. Our study of this option, as we have envisaged it, suggests that it brings benefits primarily in terms of better reflecting Network Rail's efficient Schedule 4 costs based on the works carried out, and avoiding the risk that Network Rail will benefit from deferring or cancelling maintenance and renewals works. The main downside of the option would appear to be the potential difficulty with concluding whether variations in activity and expenditure volumes in any particular year are efficient, relative to ORR's Final Determination (the ACS is calculated to reflect the efficient level of work that Network Rail must undertake to maintain and renew the network in the control period). However Network Rail has already started reporting in its Regulatory Accounts the amount of financial under/over performance related to variation in the volume of works. This suggests the information necessary for recalculating the ACS on a more frequent basis should be available.

ACS reform potentially reduces the problem of Network Rail's over-recovery of costs in an environment where it under-delivers on the volume of work. This option has a direct impact only on franchised passenger operators as they are the only train operators to pay the ACS at the moment. But in the present SoW any potentially useful effect in reducing operator costs by reducing their risk exposure is limited, because of protection against changes in charges in existing franchise arrangements.

The impact on specific (types of) operator will depend on the extent to which the planned activities set out at the periodic review subsequently change on different routes. Train operators using a route where the planned activity volumes are delivered according to the initial plan would not see their ACS change (at least in relation to operations on that particular route) while operators on a route where a significant portion of activities are being deferred would benefit from a reduction in their ACS during the control period.

To some extent, the benefit that the option brings in terms of better reflecting efficient Schedule 4 costs based on volume of works carried out by Network Rail is counterbalanced by the costs associated with added complexity and volatility of charges within the price control period in the current SoW. However, even if franchised operators are less impacted by this option in this SoW, the option would still result in a better allocation of costs between funders and Network Rail. The proposed option would bring additional benefits in the "Dynamic railway" SoW where franchised passenger operators are exposed to changes in access charges. In that case, a more frequent

recalculation of the ACS reflecting volume of work actually carried out would reduce the financial risk exposure of passenger operators and potentially have a positive impact on the value for money of franchises and passenger fares.

Key characteristics

Description of option

The proposed option involves improving the accuracy and/or frequency of the ACS calculation to better reflect possessions requirements given the volume of engineering works carried out. This option could involve re-setting the ACS annually based on more accurate forecasts of activity volumes. This would contribute to enhancing the accuracy of the ACS.

There are some concerns that NR has tended to over-recover its passenger Schedule 4 costs through the ACS when planned work is not carried out (as the compensation payments are then lower than the ACS payments), and there has previously been some suggestions that ACS should take account of these work cancellations.

One of the main objectives of the Schedule 4 regime is to incentivise NR to deliver possessions efficiently. This option could also make the Schedule 4 regime more focused on NR's under/out performance due to effective possessions management by removing variations relative to baseline in Schedule 4 costs due to changes in the volume of works delivered.

The total level of actual Schedule 4 compensation paid out by NR is a function of several factors including:

- the volume of upgrade and renewal activity that needs to be carried out this is largely set out at the start of a control period based on the output specifications and NR's business plan;
- compensation rates (based on estimated costs of disruption such as running replacement buses, revenue loss, etc.) this is set periodically by ORR; and
- possessions management (e.g. time required to carry out planned works and scheduling works such as to minimise disruption to timetabled services) this is controlled by NR who can outperform its baseline estimate by improving its performance.

This option refers primarily to adjusting the ACS based on changes in the volume of engineering works carried out, however a more frequent ACS recalculation could also involve resetting compensation rates if material changes in the cost of disruption are identified although this could potentially lead to a significant increase in the costs and effort required to implement this option.

The graph below illustrates the impact of this option by showing a hypothetical variation in actual passenger Schedule 4 costs (red line) relative to the initial ACS set at the periodic review (green line). In the first four years of the control period, actual Schedule 4 costs are below the estimated level while in the last year they are slightly above the estimated level. Under the proposed option the ACS would be adjusted during the control period. The difference between the adjusted ACS line and the initial ACS line represents variations due in the level of work carried out. The difference between the adjusted ACS and the actual Schedule 4 costs represents cost savings due to better possessions management.

Figure E.1: Schedule 4 and ACS costs



At the CP5 price determination, East Coast proposed that passenger operators should be able to claim back ACS payments for planned work not carried out, on the basis that in CP3 and CP4 NR over-recovered on Schedule 4 payments as they under-delivered on work against the planned schedule of works.

Description of counterfactual

Franchised train operators receive compensation payments each time their services are disrupted due to NR restricting access to the network infrastructure to undertake planned engineering works under Schedule 4 arrangements.

Baseline Schedule 4 costs are recovered through the ACS, which is fixed ex-ante for the entire duration of the price control period. The ACS can be viewed as the equivalent of the Schedule 8 benchmark as it funds Network Rail for a target level of possessions (similar to a free allowance under a benchmark). Therefore, if NR takes more/less possessions than expected, it will under/over recover its actual Schedule 4 costs. The system is designed to be financially neutral for passenger operators if NR delivers its baseline engineering plans efficiently. The ACS covers estimated Schedule 4 costs related to maintenance and renewals activities. Schedule 4 costs that are incurred for enhancements works are not taken into account in setting the access charge supplements and are included in the capital into the costs of those enhancements.

The estimated Scheduled 4 costs are calculated at the periodic review based on planned maintenance and renewals activity volumes and a unit cost per asset type. Activity volumes are estimated at route-level but unit costs per activity are estimated at a national level. For certain asset types (e.g. track renewals, signalling, etc.) activity volumes are defined as a physical measure (e.g. km of track replaced). Other activity volumes, such as maintenance work, are measured in terms of expenditure as a proxy for volume delivered. If the required possessions are above the estimated level, passenger operators receive more compensation in Schedule 4 payments than they pay out through the ACS and vice versa.

Relevant factors impacting the form and/or the effectiveness of the option

- Franchising regime under the current franchised regime, franchised passenger operators are
 protected against changes in ACS between price control periods therefore this limits the impact of
 the proposed option.
- **Data availability** calculating the ACS more frequently will increase data requirements compared to the current ACS regime, and the ability to improve the accuracy of the charge depends heavily on the data available and the costs of collecting any additional data.

Option 19: More	e frequent ACS calculation
Implementation	
Information requirements	The option requires yearly updating of maintenance and renewals activity volumes planned by NR against the plan set out at the periodic review.
	Most of the information required for this option, such as Schedule 4 unit costs for each type of activity, is calculated at the periodic review and would not need to be recalculated every year. The variables that would need updating on an annual basis are:
	• Physical activity volumes (e.g. km of track, signalling, switches and crossings replaced) planned over that given year for activity types where such measures are available;
	• Monetary activity volumes (e.g. expenditure on maintenance, civils renewals, etc.) for the given year where physical volumes are not available/appropriate.
	Starting with CP5, Network Rail's Regulatory Accounts Statements separate financial performance due to efficient planning of possessions from the impact of deferrals of planned activities. ¹¹⁸ This type of analysis could also form the basis for recalculating the ACS.
Drivers	The main driver behind changes in the ACS during the price control period under this option is the variation in the volume of works carried out by NR. As activity volumes are estimated at route-level the change in the ACS for a particular operator will depend on changes in the delivery of planned activities on the routes used by that operator.
Calculation principles	The calculation necessary for this option involves re-running the ACS calculation currently used at the periodic review but with updated activity volumes. The steps generally needed to calculate the ACS involve: ¹¹⁹
	1. Estimate Schedule 4 unit costs for different types of activities (e.g. track renewal, signalling renewal, track maintenance) from historical data (for CP5 this was done using 2011/12 data) by dividing the passenger Schedule 4 cost for that activity by the respective volume of work;
	2. Multiply the Schedule 4 unit costs for each activity as estimated above by the planned volume of work at route-level for the period covered by the ACS;
	3. Adjust the resulting costs by taking account of factors such as improvements needed to meet targets for the passenger Possessions Disruption Index (PDI).
	The main change in the yearly recalculation of the ACS would be <u>updating the</u> <u>planned volumes of work used in step 2</u> of the calculation process described above. In the least complicated and most practical form, this option would not involve recalculating Schedule 4 unit costs or resetting possession disruption targets.
	There could be different ways of implementing the proposed option however we envisage the recalculation will involve updating the ACS calculation based on forecasts of the planned activity volumes at the start of each year.
Practical	1. Using forecasts of activity volumes at the start of the year may still lead to some under/over performance if some activities are then deferred however these short

¹¹⁸ Network Rail, *Regulatory Financial Statements: Year ended 31 March 2015,* available <u>here.</u> ¹¹⁹ The methodology used by Network Rail to forecast Schedule 4 costs for the next price control period is described here

0	ntion 10. Moro	froquont	ACS col	culation
U	puon 19. more	nequent	ACS Cal	culation

considerations term forecast should be much more accurate then forecasts developed several years ahead at the periodic review stage.

2. While measuring progress in activity volumes for asset types where a physical measure is available (e.g. track, signalling) would be somewhat more straight-forward, measuring activity volumes using financial measures (e.g. expenditure) might be more difficult. This would require determining to what extent lower than expected expenditure, for example, results from deferred activity or lower unit costs.

Lead time The option could be introduced relatively quickly and could be available for implementation at the start of CP6.

Resources Most of the effort required for implementing this option falls on NR and ORR. NR will be required to monitor and update maintenance and renewals activity volumes at route-level on a yearly basis. The adjustments for activity deferrals that have been included in the 2014-15 Regulatory Accounts Statements suggest that most, or at least part, of the data collection work required for this option is already being done. Furthermore NR should have a good view of the type and volumes of engineering activities planned for the upcoming year therefore the proposed option should not require extensive additional data or work.

ORR will be required to verify the activity volumes provided by NR and to confirm the ACS recalculation.

Performance	against crite	eria							
Axioms	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
System safety	=	=	=	=	=	=	=	=	
	The proposed option should have no major impact on system safety, although if it leads to better planned engineering works (by removing the incentive to defer maintenance and renewal activities), it might lead to a better maintained network which would in turn improve safety. However this effect is considered marginal.								
Consistency	=	=	=	=	=	=	=	=	
with iaw	Article 35 of Directive 2012/34 establishes that "infrastructure charging schere encourage railway undertakings and the infrastructure manager to a disruption and improve the performance of the railway network th performance scheme. This scheme may include penalties for actions which dis operation of the network, compensation for undertakings which suff disruption and bonuses that reward better-than-planned performance." In addition, ORR has the power under the Railways Act 1993 to prepare, pub vary model clauses for track access agreements. Model clauses are standard that are attached to all track access agreements of similar type. In particula model clauses set out the charges and incentives. The option is consistent with current legislation.						emes shall minimise hrough a lisrupt the ffer from Iblish, and rd clauses Ilar, these		
Funding of NR	++	++	++	++	++	++	++	++	
efficient costs	The option would still allow funding of NR's cost for an efficient level of possessions. By adjusting the amount recovered through the ACS annually depending on the volume of works carried out, there is less scope for NR to under/over-recover its efficient Schedule 4 costs.								

Option 19: M	ore frequer	t ACS calcu	lation						
Allowance for	=	++	=	=	=	=	=	=	
market conditions	Under a franchise regime where franchised passenger operators are protected against changes in access charges, recalculating the ACS annually has little or no impact on the passenger operators' financial performance.								
	Under a S charges, tl operators.	oW where ne propose	e franchise ed option	ed passenge would redu	er operato ice the fi	ors are exp nancial risk	oosed to c faced by	hanges in passenger	
	Other oper	ators that	do not pay	the ACS are	not impa	cted by the	proposed c	ption.	
A single	=	=	=	=	=	=	=	=	
approach for the network as a whole	The proposed option would apply to all train operators that pay the ACS. There would still be a difference in regimes between train operators that pay ACS (i.e. franchised passenger operators), those that can opt to pay the ACS (open access operators) and those that do not pay the ACS (freight operators). Therefore the impact is considered neutral compared to the current regime								
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
Service costs	++	++	++	++	++	++	++	++	
	The option would still allow recovery of NR's cost for an efficient level of possessions, however it would represent an improvement on the current regime given that it should make the amount recovered more accurate (i.e. reduce the potential for over- recovery of Schedule 4 costs by NR as in the present).								
Efficient	=	=	=	=	=	=	=	=	
whole-system whole -life industry net costs	The proposed option will not necessarily impact industry net costs. It rather involves a transfer of funds between NR, train operators and funders. For example, on a whole industry basis, ACS savings for train operators are counterbalanced by lower NR income.								
	While the option removes the risk of over-recovery by NR, which currently places additional costs on franchised passenger operators, there is also a risk of under-recovery which could shift costs to NR.								
Efficient long	+	+	+	+	+	+	+	+	
run investment decisions	The propo removing f long-run in	sed optior NR's incent vestment c	n might ha ive to unde lecisions.	ave some p er-deliver w	oositive in ork, thus i	npact on th mproving ir	nis criterion ncentives fo	1 through r efficient	
Efficient	=	=	=	=	=	=	=	=	
performance management	The propo changes in of network should not still be en targets thr	sed option the estima upgrades affect the titled to a ough efficie	refers to a ated level o and rene incentives any financi ent possess	adjusting the of possessio wals carriec of NR to mini ial benefits sions manag	e ACS mo ns require l out by f nimise pos arising fi ement.	re frequent ed due to ch NR. If applie ssessions dis rom outper	ly to take a hanges in the ed appropr sruption as forming po	iccount of ne volume iately this NR would ossessions	

Option 19: M	ore frequent	t ACS calcu	lation							
Efficient use	=	=	=	=	=	=	=	=		
of network capacity	The proposed option might have a small beneficial effect on the efficient use of the network as updating the ACS to better reflect actual work done could encourage NR to better plan its possessions (as the only available way to outperform its Schedule 4 target). However this impact is uncertain and somewhat speculative therefore we have rated this assessment Amber.									
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
Predictability	-	-	-	-	-	-	-	-		
	Adjusting th one year to the level fix	ne ACS ann another) a ed ex-ante	ually coul nd would for the en	d result in l make the re tire price co	higher vol egime less ontrol perio	atility (as w predictable od.	ork is defer compared	rred from to having		
	over a price end of a cor work until t annual ACS	e control p ntrol perioc he end of t update cou	eriod. The d, which m he period ild make t	e effect of ere is currer hay be in par to minimise hese peaks	ntly a peal rt due to N e its Sched much sma	k in planned NR wanting lule 4 liabilit iller and mo	d works tov to save non ties – switch re frequent	vards the -essential ning to an		
Simplicity	-	-	-	-	-	-	-	-		
	The annual recalculation of the ACS would add another layer of complexity to the regime, for example due to the need to distinguish between Schedule 4 savings due to efficient planning versus savings due to lower activity volumes.									
Transparency	=	=	=	=	=	=	=	=		
	The propose as the ACS v	ed option d would be de	loes not ha	ave a major ed on the sa	impact on ame set of	the transpa principles a	arency of th as in the pre	e regime, sent.		
Low		=	-	-	-	-	-	-		
transaction costs	The propos changes in determine adjustment arising from	ed option ACS levels franchise p would pro the propo	would lea would re ayments. obably no sed optior	ad to an ind equire an ac Under a So ot be requi n.	crease in t djustment W with le red thus	transaction of the fina ss franchise reducing th	costs partion ncial mode protection ne transact	cularly as I used to , such an ion costs		
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
NR	+	+	+	+	+	+	+	+		
accountability	The propos over recove have limited	ed option o ery due to d impact in	could be p changes i improving	positive in li n the volun g possession	miting the ne of wor s manage	e scope for k carried ou ment by NR	Schedule 4 ut, although	under or n it might		
Non-arbitrary	++	++	++	++	++	++	++	++		
allocation of costs	The propose passenger S adjustment then this w different par route.	sed option Schedule 4 should be vould also issenger op	would e costs bas done base result in erators de	nsure that sed on the ed on variat a better al epending or	the level volume of ions in the location of the amou	l of ACS b f works carr e volume of of Schedule unt of works	etter reflec ried out. As works at ro 4 costs ar s carried ou	ts actual the ACS oute-level nong the t on each		

Option 19: More	frequent ACS calculation

+

++

The impact on franchised passenger operators under the current SoW would be limited however due to the protection from changes in access charges. A positive impact would exist in a SoW where such protection is reduced.

Optimal traffic	=	=	=	=	=	=	=	=	
growth	There is growth.	no clear imp	oact from	the propose	d option	on incentiv	es for opt	imal traffic	
Aligning	+	+	+	+	+	+	+	+	
incentives	The impa current ir out, and ACS pass incentive	ct of the pro acentive stru franchised p es through s for NR to a	pposed opt cture for n passenger their frar ctually car	tion on this on ninimising d operators a nchise agree ry out non-e	criterion is isruption of re mostly ements. H ssential w	s unclear. It due to the v protected lowever, it orks to imp	would no works actu from char might in rove the n	t affect the ally carried nges as the nprove the etwork.	
Value for	+	+	+	+	+	+	+	+	
money for funders, taxpayers and users	The proposed option does not materially affect incentives for service quality or industry costs. However the ACS recalculation would result in estimated Schedule 4 costs not being passed on to funders when works are not carried out thus potentially resulting in a lower overall funding requirement.								
	The recalculation could similarly result in an increase in the ACS and therefore the funding requirement if NR activity volumes are above the estimated level. While an upward revision to the ACS is possible, particularly during individual years, recent evidence suggests there is a downside bias regarding variations in NR planned activity volumes which makes an over-recovery of Schedule 4 more likely particularly when considered over an entire price control period.								
Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	

+

The proposed option brings benefits primarily in terms of ensuring that NR's efficient Schedule 4 costs reflect works carried out and NR does not benefit from deferring or cancelling maintenance and renewals works. The main downside of the option is the potential difficulty with measuring activity and expenditure volumes in order to update the ACS annually. NR has however already started reporting in its Regulatory Accounts the amount of financial under/over performance related to variation in the volume of works. This information can serve as a starting point for recalculating the ACS more frequently.

+ +

+

+

+

The impact of the proposed option under the current SoW is somewhat limited by the fact that franchised passenger operators have reduced exposure to changes in access charges. To some extent, the benefit that the option brings in terms of better reflecting efficient Schedule 4 costs based on volume of works carried out by NR is counterbalanced by the costs associated with added complexity and volatility of charges within the price control period in the current SoW. However, even if franchised operators are less impacted by this option in this SoW, the option would still result in an improved allocation of costs between funders and NR.

The proposed option would bring additional benefits in a SoW where franchised passenger operators are exposed to changes in access charges. In that case, a more frequent recalculation of the ACS reflecting volume of work actually carried out would

reduce the financial risk exposure of passenger operators and potentially have a positive impact on passenger fares.

Impact on stakeholders

In the current SoW this option has an impact only on franchised passenger operators as they are the only train operators to pay the ACS at the moment. Recalculating the ACS on an annual basis would reduce the scope for franchised passenger operators to over / underpay relative to Schedule 4 compensation received due to variations in the level of engineering works carried out by NR. Under the current SoW with franchise protection however, the recalculated ACS would be largely passed through to the franchising authority (DfT or Transport Scotland).

It is important to note that the impact on specific (types of) operator will depend on the extent to which the planned activities set out at the periodic review subsequently change on different routes. Train operators using a route where the planned activity volumes are delivered according to the initial plan would not see their ACS change (at least in relation to operations on that particular route) while operators on a route where a significant portion of activities are being deferred would benefit from a reduction in their ACS during the control period.

The proposed option would also reduce the scope for NR to under/over recover its Schedule 4 costs due to variations in the planned volume of works. In recent years NR over-recovery of its Schedule 4 costs has been linked to planned work not carried out. Based on this recent experience NR is likely to lose revenue through this change.

An indicative quantitative analysis has been conducted to inform the assessment in this section. The analysis uses existing industry information to understand the potential impacts of the proposed option under different scenarios.

We have considered the implications under three scenarios depending on the extent to which under/over recovery of passenger Schedule 4 costs is a result of variation in activity volumes:

- Lower a third of passenger Schedule 4 under/over recovery is due to variation in activity volumes;
- **Mid** based on 2014-15 estimates, around 57% of Schedule 4 under/over recovery is due to variation in activity volumes¹²⁰;
- **Upper** 75% of passenger Schedule 4 under/over recovery is due to variation in activity volumes.

The quantitative impacts on different types of stakeholders are discussed below. While the estimated impacts are relatively small especially compared to the total amount of charges paid by train operators (impacts observed in the analysis amount to a less than 1% reduction in total charges paid by operators), it has to be noted that these impacts could potentially become more significant if the volume of work not carried out is very large or if a particular operator is disproportionately affected by variations in the volumes of works. Furthermore in a SoW with less franchise protection, the effect on operators' profit margins can be significant, as shown below.

Franchised
commuterOur quantitative analysis shows that recalculating the ACS based on the assumed
variations in activity volumes would result on average in an annual ACS saving
between £0.4m and £0.8m with the mid estimate around £0.6m for this operator
type. This means a decrease in the ACS by between 5.5% and 13%.

In a SoW with less franchise protection, the potential cost savings for train operators resulting from this option could increase the profit margin of a franchised commuter passenger operator by between 5% - 12%.

¹²⁰ Network Rail, Regulatory Financial Statements: Year ended 31 March 2015, (Statement 5a)

Franchised regional passenger operators	The quantitative analysis shows that recalculating the ACS based on the assumed variations in activity volumes would result on average in an annual ACS saving between £0.3m and £0.7m with the mid estimate around £0.5m for this type of operator.
	If these savings were kept by train operators rather than passed through franchise agreements, this would have a positive impact on the profit margin of this type of operator of between 2.5% and 6%.
Franchised inter-city passenger	The quantitative analysis shows that recalculating the ACS based on the assumed variations in activity volumes would result on average in an annual ACS saving for this sector between £0.8m and £1.7m with the mid estimate around £1.3m.
operators	In terms of profit margins for this operator type, this would result in an increase between 12% - 28%.
Open access passenger operators	As open access operators have not currently opted in to pay the ACS and receive full compensation under the Schedule 4 regime, the impact of the proposed option is negligible.
	It is possible that the asymmetric downside risk (paying more in ACS than receiving in Schedule 4 compensation) observed historically has been a factor deterring open access operators from opting in to pay the ACS. Addressing this risk may make the ACS regime more attractive to open access operators.
Freight operators	As the ACS regime does not apply to freight operators, the impact of this option is negligible.
NR	The proposed option means that NR would not retain any potential over-recovery of passenger Schedule 4 costs associated with variations in activity volumes. Based on the quantitative analysis undertaken, NR's total income would diminish on average between £11m and £25m per year with the mid estimate around £19m.
Funders	Where franchise agreements mean that train operators' savings from the ACS recalculation are passed on to funders, the impact of this option is represented by a lower funding requirement and is equal to the NR revenue loss presented above.
	The impact on different funders will depend on the variation in activity volumes on different routes. For example, a higher variation in activity volumes in Scotland would result in a higher adjustment to the ACS charged to ScotRail and consequently a lower support requirement from Transport Scotland.
Passengers and freight	Any impact on end users is likely to be limited, particularly regarding freight users as freight operators are not affected by this option.
users	Under a SoW with less franchise protection, this option would reduce the risk faced by franchised passenger operators with respect to Schedule 4 costs and could consequently have a positive impact of passenger fares.

Summary of industry commentary

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

- Stakeholders have generally commented that the benefits of the option for train operators are unlikely to be significant in the current SoW but that under-delivery of planned activity volumes leading to over-recovery of Schedule 4 costs has been an issue in recent years. In particular:
 - Some stakeholders remarked that the difference between the Schedule 4 ACS paid

and the compensation received was material.

- Open access operators commented that they do not currently see the benefit of paying the ACS in return for full Schedule 4 compensation.
- Train operators also raised general concerns regarding the benefits of having the ACS and stated the need to establish greater clarity as to its purpose before moving on to assess any reform options.
- Passenger operators commented that:
 - o a more general objectives-driven review of the possessions regime is needed; and
 - the benefits of this option would be counterbalanced by increased complexity and volatility of charges.
- Network Rail considers that it is important that the ACS reflects the efficient level of compensation Network Rail expects to pay to train operators, as a result of undertaking engineering work to maintain and renew the network.
- Funders considered that whilst this option addresses a recognised issue and could bring some benefits, a yearly change in the ACS would result in yearly adjustments to franchise payments increasing the negotiating burden between funders and franchised operators.

ANNEX H REFORM SCHEDULE 4 DISCOUNT STRUCTURE FOR NOTICE PERIOD OF POSSESSION DETAILED ASSESSMENT

Option 22: Reform Schedule 4 discount structure for notice period of possession								
Overall performance against the RDG Vision in each SoW								
Current SoW	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers	
=	=	=	=	=	=	=	=	

This option involves reforming the discount structure applied to Schedule 4 compensation rates paid by Network Rail to passenger train operators for revenue loss due to disruption arising from planned track possessions. The current discounts vary according to how much notice of the possession is given and reflect the existing evidence that there is a lower impact on long-term passenger demand from disruption announced well in advance.

Reforming the structure of discounts aims to address some concerns in the industry that while discounts incentivise early booking of possessions, they can be counterproductive for efficient planning of works and possessions. The discount structure provides incentives to Network Rail to plan and book possessions early, in particular more than 26 weeks in advance, when the largest discount is given. The concern is that the workplan for some of those possessions may not be fully developed at that stage, potentially resulting in inefficient use and late cancellations of possessions. Currently more than 90% of possessions are booked in this timeframe.

The discount structure also makes it costly for Network Rail to re-plan possessions, reducing the incentive for Network Rail to make changes to possession plans after the initial notification, even in cases when it would otherwise be beneficial. This option was also considered as an area where the approach may not have kept up to date with the fact that passengers now have better and more immediate access to information about timetables and impact on services of engineering works.

There are numerous ways in which such an option could be implemented, including increases, as well as reductions, in the discounts, and changes in the notice dates at which the discounts apply. We have examined several options that we envisage cover a range of possibilities for implementing this option. The scenarios considered involve:

- Reducing or removing the discounts current notification timeframes are kept but early notification discounts are reduced or removed altogether; and
- Reforming notification thresholds and discount rates notification thresholds are adjusted as well as reducing the discounts applied for early notification.

The viability of the option critically rests on how much difference early notifications make in reducing the disruptive impact of possessions and whether reducing discounts will alter Network Rail's possessions planning processes. An analysis of the impact of planned disruption on passenger demand is currently being undertaken by the Passenger Demand Forecast Council (PDFC) and is expected to conclude in early 2016. A decision on the level of discount factors should incorporate this latest evidence.

Given the current timetabling process, and particularly the requirement to have the timetable, and therefore most possessions agreed by T-12¹²¹ (the Informed Traveller timetable), reducing discounts alone is likely to have only a marginal impact on Network Rail's possessions planning process. Any reduction in discounts, particularly for the earliest timeframe, will increase total Schedule 4 compensation (and also be reflected in a higher ACS). In the current SoW, higher Schedule 4 payments would provide additional revenue to passenger train operators with the higher ACS costs

 $^{^{\}rm 121}$ T-12 refers to 12 weeks before the timetable is operational.

being largely passed on to funders, at least for the duration of existing franchise agreements. In the "Dynamic railway" SoW, with less franchise protection, the higher ACS would be borne by train operators with the overall impact being neutral if Schedule 4 costs are in line with the baseline set at the periodic review. The measure which could potentially provide most benefits involves reconsidering the notification thresholds, for example, by setting a threshold more closely linked to passenger milestones (such as T-12 when advance tickets become available).¹²² These benefits are harder to quantify without a clear view of how Network Rail's possessions planning would change as a result; however introducing more flexibility by allowing Network Rail to benefit from a discount when possessions are notified in time for T-12 could bring benefits in terms of more efficient planning of possessions without increasing Schedule 4 costs significantly, if discount factors are not significantly reduced.

At the moment there are several ongoing reviews that could have an impact on the Schedule 4 discount structure including: RDG's Asset, Programme and Supply Chain Management (APSCM) subgroup's work on how possessions are planned and delivered, Network Rail's IAP¹²³ and the PDFC's work on the impact of planned disruption on passenger demand. Any changes to possessions planning and timetabling or new evidence on passenger behaviour arising from these initiatives should be taken into consideration.

Key characteristics

Description of option

This option involves reforming the discount structure applied to Schedule 4 revenue loss compensation rates depending on how much notice of the possession is given. This reform could consist of modifying the discount factors and/or the notification thresholds for which the discounts are applied.

The current discount structure is based on evidence that planned disruption has a lower impact on long-term passenger demand when notified in advance. The current discount structure incentivises NR to book possessions early (more than 26 weeks in advance) and may also lower the incentives for NR to make changes to possession plans after it has notified rail operators of its intention. This may be counterproductive for efficient planning of possessions as possessions may be booked early to benefit from the discount even when work is uncertain resulting in inefficient use or late cancellations of possessions, a concern expressed by train operators at the PR13 price review process.¹²⁴

In addition, if, for example, NR realises after the initial notification that it would be beneficial to lengthen the possession slightly or to rearrange the possession due to the late announcement of a large event that will create heavy passenger demand, the discount scheme at present could discourage such changes in possessions.

Although a reform of the current discount structure could involve increasing as well as reducing the discounts, our assessment focuses on the option of reducing or even removing the discounts altogether as this more clearly addresses the concerns that have been voiced by the industry in terms of the incentives provided to NR. We envisage that increasing the discounts (i.e. lowering Schedule 4 compensation rates) would only be considered if there is new evidence suggesting a lower impact of planned disruption on passenger demand (relative to unplanned disruption) than

¹²² It should be noted that some intercity operators may allow passengers to book tickets up to six months in advance although this is mainly for mid-week services which are less likely to be affected by possessions.

¹²³ The **Industry Access Programme** (IAP) is a Network Rail and industry project looking at how to optimise track access including possessions for maintenance, renewals and enhancement works

¹²⁴ ORR (2012) Consultation on Schedules 4 and 8 possessions and performance regimes p.14

has been the case until now.

The issue of early notification discounts was taken into consideration by ORR in its consultation on the Schedule 4 and Schedule 8 regimes going into the CP5 price determination. It concluded that discounts are still appropriate as they reflected the lower marginal revenue loss for train operators when possessions are notified early. ORR also stated that early possessions planning by NR is driven more by internal timelines (such as the need to produce the Engineering Access Statement) rather than Schedule 4 discounts.

Description of counterfactual

Schedule 4 compensation rates for loss of revenue are calculated as a discounted percentage of the Schedule 8 compensation rate that would be applied, to reflect the benefit of providing early notice of a possession and giving train operators and end-users alike more time to make alternative/ appropriate arrangements and thus reduce disruption.

Schedule 4 compensation payments also reimburse train operators for the cost of providing bus replacement services however this element of the compensation rate is not subject to notification discounts.

The revenue loss compensation is based on an estimate of the marginal revenue effect (MRE) per passenger journey multiplied by the number of passenger journeys. It is considered that when given enough notification of disruption, passengers are less likely to be put off from travelling by train in the future therefore having a lower negative impact on long-term demand than late notified or unplanned disruption. The amount of discount is determined by factors that vary according to the amount of notice given to passenger operators, and the type of service that is being disrupted.

The discount rates distinguish between Service Groups with a higher or lower "late time multiplier" (reflecting the value passengers place on their journey being on time) using the same 'late time multiplier' that is also used for Schedule 8 payments.¹²⁵ A revised range of notification discount factors was set at CP5 to reflect changes in the estimated late time multipliers for each service group.

There are three levels of notice for possessions, and up to four rates of discounts possible within each level. The current notification thresholds are:

- 26 weeks before operation, with the amount payable between 40% and 55%. This is by the 'New Working Timetable', which is the earliest notification to operators of the next timetable to come into operation, thus the earliest opportunity to inform passengers of upcoming disruption to services.
- 22 weeks in advance, with the amount payable between 63% and 70%. This is meant to allow consultations with train operators and information finalised in time for inclusion in the 'Informed Traveller Timetable' (issued 12 weeks in advance).
- Before 10pm the previous night, the amount payable is 85% for all Service groups. This is when the 'Application Timetable' is set. Any disruption announced after this will fall under the Schedule 8 regime and have no discount applied.

This notification discount structure is illustrated in the diagram below.

Figure F.1:Notification discount structure

¹²⁵ ORR (2013) Final determination of Network Rail's outputs and funding for 2014-19 (p.801)



From the start of CP5, train operators can also claim compensation when possessions are cancelled at late notice if costs incurred as a result of this late cancellation are greater than £5,000 (for Type 1 possessions¹²⁶). This provision can also have implications for the proposed option as it may, at least in part, address the concern of the industry that the current notification discount structure incentivises NR to book possessions too early resulting in late cancellations of possessions.

On average, close to or just over 90% of possessions are or have been booked in the earliest discount window (at least 26 weeks in advance) since 2010/11, with the share of possessions booked in the earliest period close to 100% in many periods.

Relevant factors impacting the form and/or the effectiveness of the option

The proposed option is relatively unconstrained by the factors affecting the effectiveness of the incentives regime. There are however some implications stemming from these factors that should be considered:

- **Franchising** (Factors Report Section 3.2) the proposed option could result in changes in total Schedule 4 costs and thus in the amount that needs to be recovered through the ACS. The franchising regime means that at present franchised passenger operators would be largely unaffected by changes in the ACS.
- Track Access Arrangements (Factors Report Section 3.3) and Industry complexity (Factors Report Section 4.2) the mixed use of the network and the number of stakeholders involved mean that agreeing on a suitable possessions timetable may be difficult and there may be disagreement around the amount of compensation that is required by each user.

Implementation	
Information requirements	Setting the notification thresholds and discount factors requires information on what the impact on passenger demand of possessions notified in different timeframes is.
	The calculation of both the MRE and of notification discount factors takes into account the late time (delay) multipliers contained in the Passenger Demand Forecast Handbook (PDFH), which reflect the value passengers place on

¹²⁶ Type 1 possessions are possessions of less than 60 hours in duration. They account for the majority of possessions.

Option 22: Refo	rm Schedule 4 discount structure for notice period of possession
	unscheduled delays relative to scheduled journey time.
	The Passenger Demand Forecast Council (PDFC) is currently considering the impact of planned disruption on passenger demand. The result of this study should be a key input in setting future discount factors particularly if it results in changes to the estimated late time multipliers for particular types of services.
Drivers	Since 2010/11, close to or over 90% of possessions have been booked on average in the first discount window (more than 26 weeks in advance). This means that the largest impacts are likely to occur from changes to discount factors applied to this notification threshold.
	For example, increasing each of the discount factors in the first timeframe by 1% (e.g. from 40% to 41%) leads to an estimated increase in total Schedule 4 revenue loss compensation of 1.7%, while increasing each of the discount factors in the second timeframe by 5% (e.g. from 63% to 68%) leads to a 0.5% increase in total Schedule 4 compensation.
Calculation	The notification discounts factors are calculated using the following formula:
principles	Notification discount factor = $\left(p X \frac{1}{late time multiplier}\right) + (1-p)$
	where p is the proportion of passengers that are aware of the notified disruption. ¹²⁷
	The notification discounts applied depend therefore on estimates of how passengers react to delays (late time multiplier). A change to these multipliers or any other evidence arising from the analysis currently conducted by the PDFC should help inform what notification discount factors should be applied for Schedule 4 compensation in the future.
	The notification discounts factors are expressed as percentage of MRE payable as part of Schedule 8 compensation. The MRE is the estimated revenue loss when passengers suffer unplanned disruption (i.e. this is the payable rate under Schedule 8) and is defined as the long-term revenue lost by the train operator per passenger for each minute of lateness. As it is assumed that planned disruption has a lower effect on passenger demand, the payable rate under Schedule 4 is discounted relative to Schedule 8.
Practical considerations	The possessions planning process is heavily influenced by timetabling requirements reflected in Network Rail's licence conditions. One such condition is that engineering works arrangements are completed and necessary changes made to the timetable at least 12 weeks before the timetable comes into effect (i.e. by the "Informed Traveller" timetable). This gives rise to the Informed Traveller Process with the following timelines:
	 Disruptive period possessions plan (DPPP) provided by Network Rail at least 28 weeks in advance;
	• Confirmed period possessions plan (CPPP) issued 26 weeks in advance; and

¹²⁷ This algorithm for the calculation of notification discount factors is based on the observation made by Steer Davies Gleave (SDG) in its 2007 report to ORR that not all passengers are aware of planned disruption even when this is notified well in advance. The formula reflects the fact that for aware passengers the impact of planned disruption is similar to a timetable change while for unaware passengers this is equivalent to unplanned disruption.

Option 22: Re	form Scl	nedule 4 disco	ount struc	ture for noti	ce period	of possessi	on	
	 Consultation with train operators potentially resulting in ame Engineering Access Statement in time for inclusion in the Inf timetable. 							nts to the Traveller
According to the First Capital Connect response to ORR's CP5 Draft Determininformation regarding engineering works needs to be ready from T-16 in or be made publicly available in industry systems from T-12. ¹²⁸ Other industry shave mentioned T-14 as the critical point for having the timetable ready by T While it is possible that some of the timelines described above, such as pr the CPPP 26 weeks in advance (i.e. by the maximum discount threshold influenced by the Schedule 4 discount structure, it is also clear that any changes to this process would be limited by the need to ensure the T-12 dea met. This suggests that altering the discount factors alone is likely to have limited impact on the way possessions are planned and booked.						rmination, n order to ry sources y T-12. providing holds), are any major deadline is ave only a		
	A more effective measure might be to reform the notification thresholds as well a the discount factors. Changes to notification thresholds could then result in changes to the timetabling and engineering planning process although this would still need to ensure that engineering works are agreed in advance of T-12.						result in this would	
Lead time	The posit posit As di optic there in lig	The proposed option could be introduced relatively quickly and ORR could be in a position to adjust the notification discount structure at the next periodic review. As discussed in more detail later on in this assessment some of the impacts of this option are influenced by the provisions of existing franchise agreements. As such there could be a need to consider the timing of the implementation of the option in light of the length of these franchise agreements.						
Resources required for implementati	The resources required for implementing the option depend on the type of quired for changes envisaged to the current discount structure. Changing the value of the plementation discounts factors would not impose large costs.					of of the		
	More significant changes to the notification discount structure, such as modifying the thresholds, may also lead to (or require) changes to the timetabling and engineering planning process.					odifying nd		
Performance	against c	riteria						
Axioms	Current	: Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
System safety	=	=	=	=	=	=	=	=
	The proposed option could have benefits for system safety if it encourages NR to carry out better planned engineering works rather than focus on planning to minimise disruption.						R to carry nise	
	As the proposed option is likely to raise the cost of carrying out possessions, it could also have the effect of reducing the incentive at the margins for NR to carry out engineering work. However, we expect any such impacts to be minimal, therefore given an Amber rating under this criterion.							

¹²⁸ First Capital Connect Response to the ORR's Consultation of the Periodic Review 2013: Draft determination of Network Rail's outputs and funding for 2014-19, available <u>here</u>

Option 22: Re	form Sched	ule 4 disco	unt struct	ure for notio	ce period	of possessio	on			
Consistency	=	=	=	=	=	=	=	=		
with law	The propos	ed option i	s consiste	nt with exist	ing legisla	ition.				
Funding of NR	=	=	=	=	=	=	=	=		
encient costs	Removing/ given level assuming a the ACS est Reducing d could pote	reducing d of posses ny reductio imated at t iscounts wo ntially incre	iscounts v sions. Thi on in disco che start o ould also i ease the a	vould result s should no punts is refle f the price of increase the imount that	in an ind ot affect ected in th ontrol per effective NR over	crease in So NR's recove ne baseline riod. average cos -recovers w	hedule 4 c ery of effic Schedule 4 st of posses hen it is no	osts for a ient costs costs and ssions and ot carrying		
	out the lev of under/ov (such as up the rating u	el of posse ver-recover odating the inder this c	ssions esti ry of Scheo ACS calcu riterion.	imated at th dule 4 costs ulation more	e price co should be frequen	ontrol reviev dealt with tly) therefor	w. However through oth re it has no	the issue er means t affected		
Allowance for	+	=	+	+	+	+	+	+		
market conditions	The propo operators a increases th	The proposed option could potentially increase revenues for passenger train operators and thus improve their viability when operators are protected against ACS increases through franchise agreements.								
	In the "Dynamic railway" SoW, with less franchise protection, the benefit derived from higher Schedule 4 compensation would be offset by an increase in the ACS.									
	This option compensat	n would no ion for pos	t affect fr	eight opera otified early	tors for e	example, as	they do no	ot receive		
A single	=	=	=	=	=	=	=	=		
approach for the network as a whole	Similar to o operators different.	current reg but the re	ime. The segime for	same regime freight an	e would a d open a	apply to all access oper	franchised ators woul	passenger d remain		
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
Service costs	=	=	=	=	=	=	=	=		
	Reducing/removing the discounts would result in (potentially large) increase in Schedule 4 costs. To the extent that the current early notification discounts reflect the lower costs imposed on train operators and passengers, removing these discounts would make Schedule 4 less reflective of the actual costs of disruption. These higher costs for NR would be offset through a higher ACS, therefore while the distribution of the recovery of costs might be affected, the actual recovery should not be subject to a significant change. The assessment under this criterion could become positive if it is assumed that the current discount factors do not accurately reflect the costs to train operators of early- notified possessions - i.e. early and late notified possessions lead to similar revenue losses for train operators.									

Option 22: Re	form Sched	ule 4 discou	unt struct	ure for notic	e period	of possessio	on	
Efficient	-	-	-	-	-	-	-	-
whole-system whole -life industry net costs	Current regime may offer incentives to NR to book possessions early even when scope of works might not be clearly defined. It also disincentivises ongoing changes to possessions timetable. This includes instances where a train operator might have made suggestions to improve the possessions plan, but doesn't because it is aware that Network Rail would not be willing to make changes to already booked possessions. Reforming discounts could encourage better planning of engineering works that may result in lower industry net costs. On the other hand, reducing the discount factors will result in higher Schedule 4 costs (with an increase of almost 60% if all discounts for early notification are removed). These costs amount to an extra funding requirement that funders need to cover and as such they would represent an increase in industry costs. This effect is likely to be higher than the impact of better possessions planning therefore results in an overall negative score.							
Efficient long	=	=	=	=	=	=	=	=
investment decisions	The proposed option does not have a clear impact on network investment incential although it would encourage more careful planning on NR's behalf when planning booking possessions – as there will be a greater financial cost of booking possession It also increases the marginal costs of taking possessions and carrying out we therefore it might have a small impact on NR's investment decisions. The impact is uncertain to merit a negative rating.						ncentives, nning and ssessions. out works pact is too	
Efficient	+	+	+	+	+	+	+	+
performance management	Reforming discounts could encourage better planning of engineering works, at least at the margins. Incentives to minimise disruption can still be provided through Possession Disruption Index (PDI) targets that go into the calculation of efficient Schedule 4 costs.							
Efficient use	+	+	+	+	+	+	+	+
of network capacity	While other timetabling considerations mean that a major change in the books possessions is unlikely, there could be a marginal effect such tha possessions are booked in a different manner than they would be under the discount structure. The proposed option might therefore have a slight positive impact on the effi of the network by removing or reducing the incentive for NR to focus on possessions early rather than when it is most efficient to do so.						e way NR at certain ne current ficient use n booking	
Judgement criteria	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
Predictability	=	=	=	=	=	=	=	=
	A positive impact on this criterion would occur if the discounts were reme altogether or their number (rather than level) were reduced, since the number uncertain variables in the calculation would decrease – however this impact is like be small.						removed umber of is likely to	

Option 22: Re	form Sched	lule 4 disco	unt struct	ure for noti	ce period	of possessio	on			
Simplicity	+	+	+	+	+	+	+	+		
	Removing the discounts would make the charging regime simpler as it would reduce the number of variables that go into the calculation of compensation rates. Also better aligning the discount structure with the timetabling process may make the process more intuitive and easy to understand. Reducing discounts rates or adjusting the thresholds would result in a similar level of complexity to the current regime									
Transparency	=	=	=	=	=	=	=	=		
	Removing calculation on this crite	the discou of compen erion overa	ints would isation rat II.	d reduce th es, however	e numbei this chan	r of variab ge would ha	les that go ave a minim	into the nal impact		
Low	=	=	=	=	=	=	=	=		
transaction costs	Removing trequire co transaction	the discoun mplex calo costs is mi	its would culations inimal.	result in a si of compen	mpler pos sation rat	sessions reg ces. Theref	gime but it v ore, the in	would still mpact on		
Outputs	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers		
NR	=	=	=	=	=	=	=	=		
accountability	NR would be still be accountable under the Schedule 4 regime for planning and delivering possessions efficiently. There are no clear impacts on NR accountability stemming from the proposed option.									
Non-arbitrary	-	-	-	-	-	-	-	-		
allocation of costs	The proposed option could potentially result in large increases in Schedule 4 costs (up to 60%). While NR would be funded for the higher compensation rates through a higher ACS and train operators are likely to receive a net benefit, the higher costs would have to be covered by funders. This could potentially result in a worse allocation of costs compared to the current situation. These higher costs could however be justified if they reflect a higher impact of potentially and they be be covered by the base potentially result in a worse allocation of costs could however be justified if they reflect a higher impact of potentially result in the potential optimized on the base potential optimized on the potential optimized optimized on the potential optimized optized optimized optimized optimized optimized optimized optimized o									
	than currer	ntly estimat	ed.	•	0			, ,		
Optimal traffic	=	=	=	=	=	=	=	=		
growth	The proposed option may have beneficial impacts in terms of encouraging efficient network capacity use if possessions are better planned. However, the impact of this on optimal traffic growth is uncertain and probably minimal.									
Aligning	+	+	+	+	+	+	+	+		
industry incentives	+ + + + + + + + + + + + + + + + + + +									

given that early notification, particularly before the time when passenger tickets are put on sale, would be less disruptive and have less impact on passenger demand so some incentive for booking possessions before that time would be desirable.

Value for money for funders, taxpayers and users

The proposed option would result in higher Schedule 4 compensation being paid as the effective price of the average possession would rise. Under the current SoW these higher costs would ultimately have to be borne by funders. Some positive effects could arise from incentivising better planning and making better use of possessions which may result in lower industry costs.

Given that possession planning is largely influenced by timetabling processes and licence conditions, major changes in the way possessions are planned are unlikely to take place. Therefore the higher funding requirement is unlikely to bring enough efficiency benefits to create better value for money from the perspective of funders and taxpayers.

In a SoW with less franchise protection, the higher Schedule 4 costs would be covered by train operators through a higher ACS. This means the impact on funding requirement should be largely unchanged. However if there is asymmetric risk of train operators paying out more through the ACS than receiving in Schedule 4 compensation than this could reflect in higher risk premiums in franchise bids.

Summary	Current	Dynamic railway	On-rail comp	Specified franchises	Protect freight	Beneficiary pays	Capacity allocation	Regional powers
	=	=	=	=	=	=	=	=

This option is primarily about rebalancing the focus of incentives from aiming to minimise disruption towards better planning of possessions. The viability of this option critically rests on two aspects:

1) how much difference early notification makes in reducing the disruptive impact of possessions; and

2) whether removing/reducing discounts will alter NR's possessions planning process.

The answer to the first point requires primarily detailed analysis of the impact of planned disruption on passenger demand and thus train operators' revenue. Such analysis is currently undertaken by PDFC and is expected to conclude in early 2016.

Given the current timetabling process, and particularly the requirement to have most possessions agreed by T-12 (the Informed Traveller timetable), reducing or removing discounts alone is unlikely to have a major impact on the NR possessions planning process, although some marginal impact is possible.

The largest impact however of reducing discounts is likely to be higher Schedule 4 compensation payments. Under the current SoW, higher Schedule 4 payments would provide additional revenue to passenger train operators, some/all of which may flow to funders but the costs (reflected in an increase in the ACS) would be passed on to funders, at least for the duration of existing franchise arrangements.

Furthermore, while reducing the incentive for NR to book possessions early may improve possession planning in some cases, there is also the risk of this providing incentives to NR to delay booking possessions as long as possible. This is particularly the case if financial incentives for notifying possessions early are removed altogether. This could result in lower overall efficiency compared to the current situation, considering that some degree of early notification, particularly before the time when

passenger tickets are put on sale, is desirable to both passengers and train operators and would have a lower impact on future passenger demand.

The measure which could potentially bring some benefits in the future involves reconsidering the notification thresholds, for example, by setting a threshold for befitting from a discount more closely linked to passenger behaviour (such as T-12 when advance tickets become available). These benefits are harder to quantify without a clear view of how NR's possession booking behaviour would change as a result; however introducing more flexibility by allowing NR to benefit from a discount when possessions are notified in time for T-12 could bring benefits in terms of more efficient planning of possessions and without increasing Schedule 4 costs significantly (if discount factors are not significantly reduced).

There could also be merit in ensuring there is enough flexibility in the regime such that a possession booked early could be amended at no extra cost if there is agreement between NR and the affected train operators. This could help to address current concerns that the notification discount structure makes NR less open to modifying possessions once an early notification has been made.

Impact on stakeholders

The main justification for the current regime of discounts is that the further in advance a train operator is made aware of a possession, the lower the impact of that possession on the train operator's revenues because train operators and end-users can manage disruptions more easily and passengers are less likely to be put off from travelling by rail in the future if they are given early notification.¹²⁹

An indicative quantitative analysis has been conducted to inform the assessment of the proposed option. The analysis uses existing industry information to understand the potential impacts of the proposed option under different scenarios. The analysis was conducted assuming no dynamic changes in the number and distribution of possessions across the different notification timeframes and thus captured solely the financial impact of changes in the compensation rates payable under the Schedule 4 regime. Comments on potential behavioural effects on the way NR books possessions are however included in the discussion.

We have considered the implications under three scenarios. These scenarios consider the financial impact on NR and different types of train operators of implementing alternative discount structures. The scenarios considered involve:

- Scenario 1: Reducing discounts current notification timeframes are kept but notification discounts are reduced by 15% for the earliest timeframe (e.g. payable rate increases from 40% to 55% of MRE) and by 10% for the middle timeframe (e.g. payable rate increases from 65% to 75% of MRE);
- Scenario 2: Removing discounts discounts for early notification of possessions are removed such that 85% of MRE is paid on all possession notifications;
- Scenario 3: Reform notification thresholds and discount rates notification thresholds are adjusted so that the largest discount is given for possessions booked more than 22 weeks in advance and a medium discount is given for possessions booked more than 12 weeks in advance. The notification discount factors are also reduced as per the first scenario analysed.

As notification discounts apply to the passenger Schedule 4 regime, the impact regards franchised and open access passenger operators. Given that the split of possession bookings is roughly similar

¹²⁹ ORR (2013) Final determination of Network Rail's outputs and funding for 2014-19, p.801

for all service groups (around 90% of possessions are booked more than 26 weeks in advance) the impact of changing the discounts factors on different types of operators is equally similar.

These scenarios are meant to illustrate potential impacts under a range of possible changes to the current notification discount structure. However there are other ways in which reforms could be implemented and therefore the assessment presented here does not represent an exhaustive list of potential impacts. For example, if notification thresholds are reformed but current discount rates are kept in place, the financial impact under our static assessment will be small. Similarly if discounts are increased the impacts presented here are going to be reversed (i.e. total Schedule 4 compensation would decrease). We would however envisage that higher discounts would only result from new evidence showing a lower impact of planned disruption on passenger demand rather than being applied only as an incentive for NR.¹³⁰

At a high-level the main potential impacts under the scenarios considered are:

• Total Schedule 4 revenue loss compensation paid to passenger train operators for possession disruption would go up as a result of reducing (or removing) the discounts which means that the amount recovered through the ACS would also increase. Under the current SoW, franchised passenger operators would be held harmless against any change in the ACS which means that the higher Schedule 4 costs would be effectively borne by funders, at least during the duration of existing franchise agreements. The increased Schedule 4 compensation would represent a net increase in revenue for passenger operators which may have to be shared with funders under profit sharing mechanisms depending on the specific provisions of existing franchise agreements.

If the franchise protection were removed, the higher compensation that franchised passenger operators would receive would be counter-balanced by an increase in ACS.

- Reducing the discounts could rebalance NR's possessions planning process towards ensuring possessions are booked at the right time rather than early to benefit from the discount. As mentioned above, timetabling requirements mean that NR would still need to follow largely the same possession planning process. Therefore this is also likely to be a marginal rather than a large scale effect especially if only changes to discount rates are implemented. Changes to the possessions planning process are more likely to take place if there are wider changes to the notification thresholds.
- Reducing discounts could also lower the incentive for NR to book possessions early and could result in NR providing late notifications of possessions that would result in higher costs for train operators and end-users but again this is more likely to be a marginal rather than full scale impact. Given the current timetabling process requirements, we believe a large shift in the way possessions are planned and notified is unlikely therefore the costs imposed on train operators and passengers are likely to be limited. This impact would also be mitigated by the fact that train operators would receive higher compensation for all possessions.

Franchised commuter passenger operator passenger operators
 Removing discounts (scenario 1) leads to an estimated increase in Schedule 4 compensation paid to a typical franchised commuter passenger operator of 27% (around £2.4m per year).
 Removing discounts (scenario 2) leads to a higher estimated increase in Schedule 4 compensation paid to a typical franchised commuter passenger operator of 59% (approx. £5m per year).

¹³⁰ This is consistent with the SDG recommendation in the 2007 report to ORR that incentives in the notification discount structure should be applied as a penalty for unwanted behaviour by NR (i.e. increase payment rates for late notification) rather than by offering discounts beyond the estimated cost of disruption for early notification.

Reforming thresholds and discounts (scenario 3) would increase estimated Schedule 4 compensation paid to a typical franchised commuter passenger operator by around 24% (approx. £2m per year).

There would also be similar increases in the ACS to offset these higher compensation payments. In the current SoW with franchise protection, these higher costs may be passed on to funders, therefore the higher compensation would result in a net benefit to passenger operators, at least during the existing franchise agreement. For a typical franchised commuter operator the impact on profit margins would be significant (an increase of between 29% in scenario 3 and up to 70% in scenario 2) although some or all of this may flow back to funders.

Some of this benefit might be offset by higher costs for train operators as a result of later notification of possessions. However we would not expect a large shift in the way possessions are booked to occur therefore any costs are likely to be much lower than the estimated benefits.

In a SoW with less franchise protection, the overall impact on franchised passenger operators would be largely neutral if NR possessions match the level used to set the ACS. If possessions are below the estimated level as has been the case in recent years, there is the risk for passenger operators to receive less in Schedule 4 compensation than they pay through the ACS, with this difference being greater when early notification discounts are reduced.

Franchised
regionalReducing discounts (scenario 1) leads to an estimated increase in Schedule 4
compensation paid to a typical franchised regional passenger operator of 25%
(around £1m per year).

operators **Removing discounts (scenario 2)** leads to an estimated increase in Schedule 4 compensation paid to a typical franchised regional passenger operators of 59% (around £2.4m per year).

Reforming thresholds and discounts (scenario 3) would increase estimated Schedule 4 compensation paid to a typical franchised regional passenger operator by around 20% (approx. £800k per year).

There would also be similar increases in the ACS to offset these higher compensation payments. In the current SoW with franchise protection, these higher costs may be passed on to funders, therefore the higher compensation would result in a net benefit to passenger operators, at least during the existing franchise agreement. For a typical franchised regional passenger operator profit margins would increase from around 7% (in scenario 3) up to 19% (in scenario 2) although some or all of this may flow back to funders.

Some of this benefit might be offset by higher costs for train operators as a result of late notification of possessions. However we would not expect a large shift in the way possessions are booked to occur therefore any costs are likely to be much lower than the estimated benefits.

In a SoW with less franchise protection however the overall impact on franchised passenger operators would be largely neutral if NR possessions match the level used to set the ACS. If possessions are below the estimated level as has been the case in recent years, there is the risk for passenger operators to receive less in Schedule compensation than they pay through the ACS, with the difference being greater when early notification discounts are reduced.

Franchised Reducing discounts (scenario 1) leads to an estimated increase in Schedule 4 compensation paid to a typical franchised inter-city passenger operator of 25% inter-city passenger (around £4m per year). operators Removing discounts (scenario 2) leads to an estimated increase in Schedule 4 compensation paid to a typical franchised inter-city passenger operator of 59% (around £9m per year). Reforming thresholds and discounts (scenario 3) would increase estimated Schedule 4 compensation paid to a typical franchised inter-city passenger operator by around 20% (approx. £3m per year). There would also be similar increases in the ACS to offset these higher compensation payments. In the current SoW with franchise protection, these higher costs may be passed on to funders, therefore the higher compensation would result in a net benefit to passenger operators, at least during the existing franchise agreement. The potential benefit in terms of profit margins for this type of train operator would be particularly significant. In scenarios 1 and 3, profit margins for a typical franchised inter-city passenger operator could increase by around 50% while in scenario 2 by over 100% although some or all of this may flow back to funders. Some of this benefit might be offset by higher costs for train operators as a result of late notification of possessions. However as we would not expect a large shift in the way possessions are booked to occur any costs are likely to be much lower than the estimated benefits. In a SoW with less franchise protection however the overall impact on franchised passenger operators would be largely neutral if NR possessions match the level used to set the ACS. If possessions are below the estimated level as has been the case in recent years, there is the risk for passenger operators to receive less in Schedule compensation than they pay through the ACS, with the difference being greater when early notification discounts are reduced. Reducing discounts (scenario 1) leads to an estimated increase in Schedule 4 **Open access** passenger compensation paid to a typical open access passenger operator of 25% (around £150k operators per year). Removing discounts (scenario 2) leads to an estimated increase in Schedule 4 compensation paid to a typical open access passenger operator of 57% (around £350k per year). Reforming thresholds and discounts (scenario 3) would increase estimated Schedule 4 compensation paid to a typical open access passenger operator by around 21% (approx. £130k per year). As open access operators have not opted for paying the ACS they do not receive compensation for Type 1 possessions thus these benefits result from higher compensation paid on Type 2 and 3 possessions. Open access operators would also receive the benefit of increased compensation payments without the additional costs of higher ACS. The positive impact in terms of profit margins for an open access operator would be around 3% in scenario 1 and 3 and around 8% in scenario 2. Freight As the notification discounts apply to passenger Schedule 4 compensation and freight operators do not pay an ACS, the direct impact of this option on freight operators is

> As freight Schedule 4 compensation rates for possessions notified before and after T-12 differ, there could be an impact on freight operators if the proposed option results

limited.

Option 22: Re	Option 22: Reform Schedule 4 discount structure for notice period of possession					
	in a significant change in the way NR plans and books possessions. As discussed above, NR's licence conditions regarding the Informed Traveller timetable requirements mean that possessions would still have to be largely agreed before T-12.					
NR	The proposed option would have the following estimated impacts on the total passenger Schedule 4 compensation paid by NR:					
	 Reducing discounts (Scenario 1): 26% increase – approx. £54m; 					
	 Removing discounts (Scenario 2): 59% increase – approx. £124m; 					
	 Reform thresholds and rates (Scenario 3): 21% increase – approx. £45m. 					
	NR would however be funded for these higher costs through increases in the ACS.					
	Higher Schedule 4 payments would also increase the potential for under/over recovery if the number of possessions required during a control period is higher/lower than estimated at the periodic review.					
	If actual possessions are lower that the estimated level used to set the ACS, as has generally been the case in recent years, then NR could potentially over-recover a larger amount of Schedule 4 costs. By effectively raising the average cost of taking possessions, the proposed option also increases the savings achieved by NR when not carrying out a possession. A potential side effect of this option would then be that at the margins NR would have higher incentives to postpone or not carry out engineering work.					
Funders	The proposed option would result in higher Schedule 4 costs and a higher ACS of the magnitude shown in the section above. In the current SoW these higher ACS costs would be passed on to funders, at least for the duration of existing franchise agreements. These extra costs would be captured in future franchise agreements.					
	In a SoW with less franchise protection these higher costs would be borne by franchised passenger operators offsetting the benefit they receive from higher Schedule 4 compensation. If there is a risk that franchised train operators pay out more in ACS than receive in Schedule 4 compensation than this would be likely to have a knock-on effect on the value of franchise bids resulting in lower value for money franchise agreements.					
Passengers and freight users	The direct impacts on passengers of this option are likely to be limited. Any changes to the way NR plans and books possessions could also affect passengers either positively (if NR plans possessions more efficiently resulting in less overall disruption) or negatively (if there is more late notification of possessions). Given that we would expect the vast majority of possessions to still be agreed before the Informed Traveller timetable is published (when most passengers would become aware of train schedules and would be able to buy tickets), any such impacts are likely to be small. Freight users are less likely to be impacted by this option as the impact on freight operators are also limited (see assessment further above).					
Summary of i	ndustry commentary					

Stakeholder comments, reflected where relevant throughout the detailed assessment of this option, included the following points.

• Stakeholders have commented that the most important milestone to be borne in mind when setting discounts is to have possessions reflected in the Informed Traveller timetable (T-12) when tickets are first put on sale. In practice this would mean having possessions agreed prior to that

deadline in order to allow possessions to be incorporated into the timetable uploaded at T-12.

- Passenger operators noted that it is important to bear in mind that the main reason for Schedule 4 discounts is related to the impact on passengers' willingness to travel and not to incentivise better planning (both possessions and work planning).
- Passenger operators also stressed the need to have clarity about the purpose of the current regime in order to assess whether a reform option is sensible or not. A more general objectives driven review of the whole performance incentive/compensation arrangements is needed.
- Network Rail noted that the Notification Discount Factors (both the level of discount and the timing thresholds) must be based on robust evidence of genuine lost future operator revenue at different notice periods. Therefore, further work to review this as part of PR18 appears appropriate. However, the inherent uncertainty for Network Rail in planning a long way ahead of a possession must also be recognised.
- Transport Scotland noted that the issue of possessions being booked early has been raised multiple times before. Any perverse incentives in the regime should be removed and aligned with the Informed traveller T-12 deadline.

ANNEX I ASSESSMENT CRITERIA

As explained in Section 2.3, the RDG Vision provides the basis of the assessment criteria used for the initial assessment presented in this report.¹³¹ Table I.1 below, contains the full descriptions of the criteria used. They are presented in this annex for ease of reference when reading the individual detailed assessments in Annexes B to H.

The descriptions provided are drawn from the RDG Vision but, as noted in Section 2.3, four descriptions have been clarified to assist with the process of conducting the initial assessments and to reflect feedback from the industry. In each case, text has been added (identified in the table as underlined text), with no deletions being made.

Criterion	Description
Axioms	
System safety	Charges must fund, and should not create incentives to compromise, the safety of the railway system
Consistency with law	The charges and incentives regime should comply with the relevant regulations and laws, including EU and domestic legislation (e.g. Railways Act, and Access and Management Regulations). This includes consistency with the non-discrimination principle and facilitation of effective competition. Further key elements include legal requirements for transparency, efficiency, minimum charges of direct cost incurred, the EU framework for additional charges, and specific impact tests considered by the ORR such as those on the environment.
Funding of Network Rail efficient costs	Total revenues (access charges plus government support) should allow Network Rail to recover the total efficient costs of providing and improving all services
Allowance for market conditions	Where the charges for a service exceed the costs directly incurred for the provision of that service, any mark-up should recognise pressures from competitive external markets and may only be applied if the market segment concerned can bear the cost. For the avoidance of doubt, and to avoid duplication, any legal requirement related to the allowance for market conditions is considered under this option.
A single approach for the network as a whole	The charges and incentives approach and methodology should apply to the whole network, but may be different for different customers with different characteristics. Different methodological decisions regarding the calculations of charges should not be allowed: methodology and policy decisions should be the same for the whole network. This does not mean that actual charges will be the same.

Table I.1: Assessment criteria

¹³¹ RDG (Dec 2014) "RDG vision for the charges and incentives regime in the long run" available here p13-16

Criterion	Description
Objectives	
Service cost recovery	Charges for any service provided by Network Rail should recover at least the efficient costs directly incurred to provide that service. The level at which services are defined will need to be considered
Efficient whole system whole life industry net costs	The charges and incentives regime should incentivise or enable changes in the pattern of service (including in respect of journey times) where the resulting benefits exceed the change in efficient costs directly incurred
Efficient long run investment decisions	The charges and incentives regime should incentivise or enable Network Rail to invest where the long run benefits of the investment exceed its efficient costs
Efficient performance management	The charges and incentives regime should incentivise or enable the efficient management of both planned and unplanned disruptive work
Efficient use of network capacity	The charges and incentives regime should not result in distortionary incentives for the allocation, and should encourage the best use of, available network capacity
Judgement criteria	
Predictability	The regime should avoid undue volatility in the structure and level of charges across multiple control periods, so that operators can predict the future level of charges for a given pattern of operations with a reasonable degree of confidence
Simplicity	All charges to all operators should be easily understood. The regime should be straightforward, transparent, and readily understandable at the point of use by all parts of the industry and broader stakeholders. ¹³² It must also be practicable to calculate and apply the charges at the required level of granularity.
Transparency	All charges to all operators should be derived from a clear set of principles. Any deviations from these principles should be clearly identified, and their impact clearly shown.
Low transaction costs	The charges and incentives regime should impose low transaction costs.
Outputs	
Network Rail accountability	A transparent regime will result in Network Rail being accountable to its customers, funders and users in relation to charges and incentives. However, full accountability depends on non-charging structure issues such as institutional and contractual mechanisms, which cannot be reflected in a set of

¹³² RDG (May 2015) "Review of Charges Phase 2b: Assessment of the current charges and incentives regime" available on the RDG website <u>here</u> p7
Criterion	Description
	objectives that relates to the structure of charges.
Non-arbitrary allocation of costs	If a clear distinction can be made between the base services bought by operators, and the incremental enhancements to those services bought by the DfT, Transport Scotland and other funders, then a charges regime which recovers at least the efficient costs directly incurred to provide any service can generate a non-arbitrary charge for those incremental enhancements. This can result in a non-arbitrary allocation of costs between operators and funders.
Optimal traffic growth	A regime that provides efficient industry costs, efficient long run investment decisions and efficient use of network capacity will incentivise the growth of traffic volumes where the net benefits of doing so are positive.
Aligning industry incentives	Improved efficiency from and greater co-operation (e.g. through alliances) between Network Rail, train operating companies and freight operating companies.
Value for money for funders, taxpayers and users	A regime that facilitates investment and improvements in the customer experience for both passengers and freight users, supporting the trade-offs between competing requirements, and taking into account public funds available.

Key: <u>Underlined</u> text supplements the descriptions provided in the published RDG Vision.

ANNEX J KEY FEATURES OF ALTERNATIVE SOWS

The Phase 2a slide deck,¹³³ contains a description of the characteristics of each SoW. The key features noted for each SoW are reproduced below for ease of reference when reading the assessments in the preceding annexes.¹³⁴

J.1. A more dynamic railway

- More on-rail competition between passenger operators, i.e. increased provision of passenger services by open access operators.
- Low franchise protection from changes in access charges, i.e. franchisees are on risk for changes to a wider range of Network Rail's access charges.
- Increased franchise flexibility as a result of less highly specified franchise agreements, i.e. franchisees have more freedom to adjust service provision, e.g. in reaction to changes in patterns of demand.
- Beneficiary pays approach to fixed costs, i.e. government no longer provides funding of infrastructure via a 'lump sum' direct network grant and instead directs funding to specific projects or to cover specific industry costs.
- Decisions on allocation of network capacity are no longer based largely around the rights reflected in the existing timetable. Instead, allocation may reflect other factors, such as the overall benefits of use.

J.2. On-rail competition via more flexible franchising

- More on-rail competition between franchised passenger operators or from more open access as a result of fewer services being franchised on certain parts of the network.
- Increased franchise flexibility as a result of less highly specified franchise agreements, i.e. franchisees have more freedom to adjust service provision, e.g. in reaction to changes in patterns of demand.

J.3. More highly specified franchises

 Greater franchise protection from changes in the charges and incentives regime, i.e. franchisees are protected from the financial effects of more elements of Network Rail's charges and incentives regime, e.g. the Possessions Regime, Performance Regime and Electric Current for Traction charge.

¹³³ RDG (May 2015) "Current and potential alternative states of the world" available <u>here</u>

¹³⁴ A detailed description of SoWs was also produced in the Phase 3 report on factors impacting the form and/or the effectiveness of charges and incentives.

 Reduced franchise flexibility as a result of more highly specified franchise agreements, such as a management contract, i.e. franchisees have very little freedom to adjust service provision, e.g. in reaction to changes in patterns of demand.

J.4. Freight protection / subsidy

- More financial protection or a direct subsidy for freight operators provided by governments.
- This could either be:
 - Protection from changes to Network Rail's access charges; and/or
 - Direct subsidy from government to freight operators to reflect the positive externalities / societal benefits of freight.

J.5. Beneficiary pays for network capability

- Governments no longer provide a lump sum Network Grant directly to Network Rail to fund a mix of new and existing network capability.
- Instead, funding is directed to specific projects, potentially via the users that benefit most from those schemes (e.g. franchised operators or regional funders). Alternatively, funding is provided directly to Network Rail but for specific elements of existing capability, e.g. governments explicitly fund historic financing costs, or the societal benefits of enhancements to the rail network.

J.6. Change in approach to allocation of network capacity

- Decisions on allocation of network capacity are no longer based largely around the rights reflected in the existing timetable. Instead, allocation may reflect other factors, such as the overall benefits (both railway revenues and societal benefits) generated by a particular use of a train path, e.g. intercity, commuter, freight, possession for maintenance. Or, capacity allocation may respond more quickly to changes in patterns of demand.
- In practice, a more analytical approach would be taken to allocating train paths, compared to the current SoW.
- A change in approach to allocating network capacity should be considered in two SoWs:
 - Current capacity / capability remains; and
 - A significant increase in capacity, resulting from the outputs of the 'Digital Railway' and/or a major enhancement project such as High Speed 2.

J.7. Regional decision making

- More responsibility for decision making (funding, policy, operational) at a regional level. For example, with the provision of local passenger service being procured and funded by regional bodies, e.g. Passenger Transport Executives.
- Governments no longer provide lump sum grants directly to Network Rail to fund a mix of new and existing capability. Instead, funding is directed to specific projects, potentially via the users that benefit most from those schemes (e.g. franchised operators or regional funders), or funding is provided directly to Network Rail but for specific elements of existing capability.